

"NAMES AND PORTFOLIOS OF THE MEMBERS OF THE UNION COUNCIL OF MINISTERS"

Lok Sabha Secretariat New Delhi LOK SABHA SECRETARIAT NEW DELHI

COUNCIL OF MINISTERS (1947-2015)

NAMES AND PORTFOLIOS OF THE MEMBERS OF THE UNION COUNCIL OF MINISTERS (From 15 August 1947 to 28 August 2015)



LOK SABHA SECRETARIAT NEW DELHI 2016

LARRDIS/REFERENCE 2016

First Edition : 1968
Second Edition : 1978
Third Edition : 1985
Reprint : 1987
Fourth Edition : 1990
Fifth Edition : 1997
Sixth Edition : 2004
Seventh Edition : 2011
Eighth Edition : 2016

Price: ₹ 500.00 \$ 7.50 £ 6.00

© 2016 By Lok Sabha Secretariat

Published under Rule 382 of the Rules of Procedure and Conduct of Business in Lok Sabha (Fifteenth Edition) and printed by Aegean Offset Printers, New Delhi.

PREFACE

This brochure provides information on Members of the Union Council of Ministers and their portfolios since 15 August 1947. The brochure was first published in May 1968 and the present updated edition is the eighth one in the series, which contains information for the period from 15 August 1947 to 28 August 2015.

Information on the number of Ministers in each category, tenures of all Prime Ministers and the changes effected in the Union Council of Ministers from time to time has been incorporated in the Introduction.

This brochure is divided into seven Parts. Part I deals with the Prime Ministers since Independence; Part II contains information on the Ministerial career of those who became Deputy Prime Ministers; Part III gives the names of Cabinet Ministers; Part IV enumerates Ministers of Cabinet rank but who were not members of the Cabinet; Part V contains information about Ministers of State with independent charge of their Ministries/Departments; Part VI indicates names of the Ministers of State attached to the Prime Minister and Cabinet Ministers; and Part VII lists the names of Deputy Ministers and Parliamentary Secretaries holding the rank of Deputy Minister.

A new Council of Ministers headed by Shri Narendra Modi as the Prime Minister took oath of office on 26 May 2014. Names and portfolios of the Members of this Council of Ministers upto 28 August 2015 have been included as an Annexure to the brochure.

A Name Index and a Subject Index have been appended to the brochure for ready reference.

The principal source of information contained in the brochure is the Cabinet Secretariat. Apart from this, releases issued by the Press Information Bureau (PIB), Press reports and relevant publication on the subject have also been consulted.

It is hoped that this brochure will be found useful by all interested readers.

New Delhi February, 2016 Phalguna 1937 (Saka) Anoop Mishra Secretary-General Lok Sabha

CONTENTS

		Pages
1.	PREFACE	(i)
2.	INTRODUCTION	(v)
3.	PART-I Prime Ministers	1
4.	PART-II Deputy Prime Ministers	21
5.	PART-III Cabinet Ministers	24
6.	PART-IV Ministers of Cabinet rank but not Members of the Cabinet	124
7.	PART-V Ministers of State with Independent Charge of their Ministries/ Departments	127
8.	PART-VI Ministers of State attached to Prime Minister and Cabinet Ministers	149
9.	PART-VII Deputy Ministers/Parliamentary Secretaries	223
10.	ANNEXURE Council of Ministers headed by Shri Narendra Modi (upto 28.08.2015)	237
11.	Name Index	242
12.	Subject Index	264

INTRODUCTION

The Constitution of India, under articles 74 and 75, provides for a Council of Ministers. Article 74(1) reads that 'there shall be a Council of Ministers with the Prime Minister as the head to aid and advise the President who, shall, in the exercise of his functions, act in accordance with such advice'. The Prime Minister is appointed by the President and the other Ministers are appointed by the President on the advice of the Prime Minister.

The Council of Ministers is drawn from both the Houses of Parliament. However, under article 75(3) of the Constitution, it remains collectively responsible to the House of the People, *i.e.* the Lok Sabha.

Normally, a person has to be a member of either House of Parliament to be a Minister. However, under the Constitution, there is no bar to the appointment of a person from outside the Parliament as a Minister. But such a person cannot continue as a Minister for more than six months unless he/she secures a seat in either House of Parliament (by election or nomination as the case may be). Article 75(5) states "A Minister who for any period of six consecutive months is not a member of either House of Parliament shall at the expiration of that period cease to be a Minister". Some of the instances of non-MPs having been inducted into the Union Council of Ministers are Dr. Triguna Sen in 1966, Shri Shanti Bhushan in 1977, Justice H.R. Khanna and Shri S.N. Kacker in 1979, Shri Arjun Singh in 1985, Shri Bir Bahadur Singh in 1988, Shri Dinesh Goswami and Prof. M.G.K. Menon in 1989, Dr. Raja Ramanna in 1990, Dr. Manmohan Singh and Shri Sharad Pawar in 1991, Shri Yogendra K. Alagh and Shri Balwant Singh Ramoowalia in 1996 and Shri Arun Jaitley in 1999. Incidentally, Shri P.V. Narasimha Rao and Shri H.D. Deve Gowda were not members of Parliament at the time of their appointment as Prime Minister of India on 21 June 1991 and 1 June 1996, respectively.

The Constitution does not classify the members of the Council of Ministers into different ranks, viz. that of the Cabinet rank, that of the Minister of State rank and that of the Deputy Minister. All this has been done following the British practice and in view of administrative convenience. Section 2 of the Salaries and Allowances of Ministers Act, 1952, has, however, accorded legislative sanction to such classification in the Indian context. The Act defines 'Minister' as a Member of the Council of Ministers by whatever name called and includes a Deputy Minister.

The Cabinet Ministers are heads of their Ministries and are accountable to the Prime Minister and through him to the President and the Lok Sabha. The Ministers of State are of two categories: some of them are given Independent Charge of Ministries/Portfolios while others are attached to the Prime Minister/Cabinet Ministers for assisting them in the discharge of their Ministerial functions. There are also Deputy Ministers/Parliamentary Secretaries whose primary function is to assist their senior Ministers/Prime Minister.

This brochure is a compilation, in a compact and consolidated form, of the names of various Ministers and their portfolios from time to time since India's Independence and covers the period upto 28 August 2015. The name of each member of the Council of Ministers finds a place under the rank last held by him/her except that of Shri Gulzari Lal Nanda who has been included among the Prime Ministers. It may be noted that from 15 August 1947 to 26 May 2014, (the date of dissolution of the Council of Ministers), there have been 14 Prime Ministers, including an acting Prime Minister; seven Deputy Prime Ministers, (two of whom, viz. Shri Morarji Desai and Shri Charan Singh subsequently became Prime Ministers); 302 Cabinet Ministers; eight Ministers of Cabinet rank but who were not members of the Cabinet (these Ministers were of the same rank as that of the Ministers of State); 61 Ministers of State with independent charge; 323 Ministers of State attached to the Prime Minister/Cabinet Ministers and 69 Deputy Ministers.

Pandit Jawaharlal Nehru, the first Prime Minister of India served in that capacity for the longest period of 16 years, 9 months and 13 days in an uninterrupted manner. Shri Atal Bihari Vajpayee, during his first tenure as Prime Minister from 16 May to 1 June 1996, had the shortest term so far of only 17 days. Among others, Shri Gulzari Lal Nanda served as acting Prime Minister for two brief terms of 14 days each: first from 27 May to 9 June 1964 and again from 11 January to 24 January 1966. Shrimati Indira Gandhi had the distinction of being the first and so far the only lady to assume the office of the Prime Minister of India. She remained Prime Minister first for a period of 11 years and two months — from 24 January 1966 to 24 March 1977 and later from 14 January 1980 to 31 October 1984 for 4 years, 9 months and 18 days. As regards others, Shri Morarji Desai served as Prime Minister for 2 years, 4 months and 5 days (from 24 March 1977 to 28 July 1979) and Shri Charan Singh held the office for 6 months and 18 days (from 28 July 1979 to 14 January 1980); Shri Rajiv Gandhi for 5 years, 1 month and 2 days (from 31 October 1984 to 2 December 1989); Shri Vishwanath Pratap Singh for 11 months and 8 days (from 2 December 1989 to 10 November 1990); Shri Chandra Shekhar for 7 months and 11 days (from 10 November 1990 to 21 June 1991); and Shri P.V. Narasimha Rao for 4 years, 10 months and 26 days (from 21 June 1991) to 16 May 1996). Shri Atal Bihari Vajpayee was succeeded by Shri H.D. Deve Gowda, as the 12th Prime Minister of India on 1 June 1996. Shri H.D. Deve Gowda remained in office for 10 months and 20 days (from 1 June 1996 to 21 April 1997) while his successor Shri I.K. Gujral stayed for 10 months and 28 days (from 21 April 1997 to 19 March 1998).

During his second tenure as Prime Minister, Shri Atal Bihari Vajpayee remained in office for 1 year, 6 months and 24 days from 19 March 1998 to 13 October 1999 and in his third term Shri Vajpayee held the office from 13 October 1999 to 22 May 2004.

Dr. Manmohan Singh in his first term as Prime Minister assumed the office on 22 May 2004 and held the office upto 22 May 2009. During his second term Dr. Manmohan Singh held the office from on 22 May 2009 to 26 May 2014.

Among the Cabinet Ministers, Shri Jagjivan Ram served as such for the longest period of 32 years except for a brief interregnum of one month and 26 days. On the other hand, Justice H.R. Khanna had the briefest ever tenure as a Union Minister which lasted only for five days from 30 July to 3 August 1979 when he resigned from the Council of Ministers headed by Shri Charan Singh.

Changes/reshuffles are effected in the Union Council of Ministers from time to time by the Prime Ministers. During the Prime Ministership of Shri Rajiv Gandhi, there were 36 occasions when changes, including six major reshuffles, were effected in the Council of Ministers. During the second tenure of Smt. Indira Gandhi as Prime Minister from January 1980 to October 1984, there were 28 changes in the Council of Ministers which included 8 major reshuffles. Her first tenure as Prime Minister from 24 January 1966 to 24 March 1977 had changes on as many as 39 occasions, including 8 major reshuffles. Pandit Jawaharlal Nehru's entire tenure had 63 occasions of changes, including 8 major reshuffles. Shri Lal Bahadur Shastri as Prime Minister made 10 changes, including one major reshuffle in his Council of Ministers. Shri Morarji Desai as Prime Minister effected 10 changes including 4 major reshuffles, and Shri Charan Singh during his tenure as Prime Minister made 6 changes, including 2 major reshuffles. One change each was made during the tenure of Shri V.P. Singh and Shri Chandra Shekhar. During the tenure of Shri P.V. Narasimha Rao, nine changes, including three major reshuffles, were effected in the Council of Ministers.

The first brief tenure of Shri Atal Bihari Vajpayee as Prime Minister saw one Ministerial change while during his second tenure Shri Vajpayee effected 7 changes, including 2 reshuffles. During the Prime Ministership of Shri Deve Gowda there were four changes, including one major reshuffle, while Shri I.K. Gujral, during his Prime Ministership, effected two changes in the Council of Ministers. The third tenure of Shri Vajpayee as Prime Minister witnessed 19 changes, including 14 reshuffles.

During the first tenure of Dr. Manmohan Singh as Prime Minister, there were 29 occasions when changes, including two major reshuffles, were effected in the Council of Ministers. The second tenure of Dr. Manmohan Singh as Prime Minister witnessed 36 changes, including 3 reshuffles.

The Council of Ministers in India functioned as a two-tier one for the first time during the Prime Ministership of Shri Morarji Desai from 14 August 1977 to 28 July 1979. Similarly, Shri Rajiv Gandhi promoted all Deputy Ministers as Ministers of State and made the Council of Ministers a two-tier one during his tenure. Incidentally, the Council of Ministers headed by Shri P.V. Narasimha Rao, after the reshuffle effected in September 1995, was also a two-tier one. The Council of Ministers headed by Shri Atal Bihari Vajpayee in his first tenure had eleven Cabinet Ministers only. The Council of Ministers during the Prime Ministership of Shri H.D. Deve Gowda and Shri I.K. Gujral and during Shri Vajpayee's second and third tenures was a two-tier one. The Council of Ministers headed by Dr. Manmohan Singh, during his first and the continuing second tenure is also a two-tier one.

NAMES AND PORTFOLIOS OF THE MEMBERS OF THE COUNCIL OF MINISTERS WITH CHANGES FROM TIME TO TIME

(From 15.8.1947 to 28.8.2015)

PART-I

PRIME MINISTERS

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Chandra Shekhar, Shri	Prime Minister (From 10.11.1990 to 21.6.1991*)
		Held charge of the Ministries/ Departments of Atomic Energy, Defence, Electronics, Home Affairs, Information and Broadcasting; Industry, Labour, Ocean Development; Personnel, Public Grievances and Pensions; Planning and Programme Implementation; Science and Technology; Space; Welfare and other- subjects not allocated to any Cabinet Minister or Minister of State with Independent Charge. (From 21.11.1990 to 21.6.1991*)
		Also held charge of Ministries of External Affairs and Health and Family Welfare (From 20.2.1991 to 21.6.1991*)
		Held charge of the Ministry of Water Resources and Ministry of Surface Transport (From 26.4.1991 to 21.6.1991*)
2.	Charan Singh, Shri	Minister of Home Affairs (From 26.3.1977 to 1.7.1978)
		Deputy Prime Minister and Minister of Finance (From 24.1.1979 to 16.7.1979)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Prime Minister and Minister-in-charge of all other Ministries/Departments not allocated to any other Minister (From 28.7.1979 to 14.1.1980*)
3.	Desai, Shri Morarji R.	Minister of Commerce and Consumer Industries and Heavy Industries (From 14.11.1956 to 1.1.1957)
		Minister of Commerce and Industry (From 1.1.1957 to 17.4.1957*) (Again from 17.4.1957 to 22.3.1958) (Again from 22.3.1958 to 28.3.1958)
		Minister of Finance (From 22.3.1958 to 10.4.1962*) (Again from 10.4.1962 to 31.8.1963 the date of his resignation)
		Deputy Prime Minister and Minister of Finance (From 13.3.1967 to 16.7.1969)
		Deputy Prime Minister (From 16.7.1969 to 19.7.1969)
		Prime Minister (From 24.3.1977 to 28.7.1979*)
		Also assumed charge of Ministries and Departments not allocated to any other Minister (From 24.3.1977 to 28.7.1979*)
		Also assumed temporary charge of Ministry of Shipping and Transport (From 24.3.1977 to 14.8.1977)
		Also assumed temporary charge of Ministry of Home Affairs (From 1.7.1978 to 24.1.1979)
		Also assumed temporary charge of Ministry of Health and Family Welfare (From 1.7.1978 to 24.1.1979)

 $^{{}^{*}\}text{The date of dissolution of the Council of Ministers.}$

-	 			
Š	NI	\cap		

Name

Portfolios alongwith the period thereof

4. Deve Gowda, Shri H.D.

Prime Minister and also incharge of Ministries/Departments of Personnel, Public Grievances and Pensions, Atomic Energy, Ocean Development, Space, Electronics, Petroleum and Natural Gas, Power, Jammu and Kashmir Affairs as well as all other Ministries/Departments not allocated to any other Cabinet Minister or Minister of State (Independent Charge) (From 1.6.1996 to 21.4.1997*)

Held charge of the Ministries of Home and Agriculture, Science and Technology, Chemicals and Fertilisers, Communications, Environment and Forests, Health and Family Welfare, Mines, Planning and Programme Implementation, Steel and Textiles (From 1.6.1996 to 29.6.1996)

Held charge of the Ministry of Food Processing Industries (From 1.6.1996 to 6.7.1996)

Held charge of the Ministry of Urban Affairs and Employment (From 29.6.1996 to 21.4.1997*)

Also held charge of Non-Conventional Energy Sources (From 1.6.1996 to 21.2.1997)

5. Gandhi, Shri Rajiv

Prime Minister and held charge of all other Ministries/Departments not allocated to any other Minister (From 31.10.1984 to 2.12.1989*)

Held charge of the Ministries/ Departments of Commerce and Supply, Environment & Forests, External Affairs, Industry and Company Affairs, Science and Technology, Atomic

^{*}The date of dissolution of the Council of Ministers.

Sl.No. Name

Portfolios alongwith the period thereof

Energy, Space, Electronics, Ocean Development, Personnel and Administrative Reforms, Youth Affairs and Sports, Culture and Tourism and Civil Aviation and other subjects not allocated to any other Cabinet Minister or Minister of State (Independent Charge)

(From 31.12.1984 to 14.1.1985)

Also held charge of the Ministry of Environment and Forests (From 31.12.1984 to 22.10.1986)

Held charge of the Ministry of Defence (From 25.9.1985 to 24.1.1987)

Held charge of the Ministry of Transport

(From 4.6.1986 to 24.6.1986)

Assumed the Office of the Minister of Finance

(From 24.1.1987 to 25.7.1987)

Also Minister of Programme Implementation

(From 4.5.1987 to 25.7.1987)

Assumed charge of the Ministry of Tourism

(From 15.7.1987 to 28.7.1987)

Also Minister of External Affairs (From 25.7.1987 to 25.6.1988) and

Minister of Water Resources (From 22.8.1987 to 10.11.1987)

Also held charge of the Ministry of Communications

(From 31.5.1989 to 4.7.1989)

Sl.No.	Name	Portfolios	alongwith	the	period	thereof

6. Gandhi, Smt. Indira Nehru Minister

Minister of Information and Broadcasting

(From 2.7.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)

Prime Minister and Minister of Atomic Energy

(From 24.1.1966 to 13.3.1967*)

Also held charge of the Ministry of Home Affairs

(From 9.11.1966 to 13.11.1966)

Prime Minister and Minister of Atomic Energy

(From 13.3.1967 to 18.3.1971*)

Also assumed the Office of Minister of Planning

(From 5.9.1967 to 18.3.1971*)

Also assumed the Office of Minister of External Affairs

(From 5.9.1967 to 14.2.1969)

Also assumed charge of Ministry of Finance

(From 16.7.1969 to 27.6.1970)

Also held charge of Ministry of Home Affairs

(From 27.6.1970 to 18.3.1971*)

Prime Minister and Minister of Atomic Energy

(From 18.3.1971 to 24.3.1977*)

Also assumed the Office of Minister of Home Affairs

(From 18.3.1971 to 5.2.1973)

Also held charge of Ministry of Information and Broadcasting (From 18.3.1971 to 8.11.1973)

^{*}The date of dissolution of the Council of Ministers.

Held charge of the Ministry of External

(From 19.7.1984 to 31.10.1984**) Held charge of the Ministry of Industry (From 3.8.1984 to 14.8.1984)

Sl.No.	Name	Portfolios alongwith the period thereof
		Also held charge of Ministry of Planning (From 18.3.1971 to 24.4.1971)
		Also assumed the Office of Minister of Space (From 2.6.1972 to 24.3.1977*)
		Also assumed the charge as the Minister of Electronics (From 17.6.1971 to 24.3.1977)
		Also held charge of Ministry of Steel and Mines (From 6.6.1973 to 23.7.1973)
		Also held charge of Ministry of Planning (From 2.1.1975 to 24.3.1977*)
		Also assumed the Office of Minister of Science and Technology (From 2.1.1975 to 24.3.1977*)
		Assumed temporary charge of the Ministry of Defence (From 1.12.1975 to 21.12.1975)
		Prime Minister and Minister of Atomic Energy (From 14.1.1980 to 31.10.1984**)
		Also held charge of the Departments of Electronics, Science and Technology and Space (From 19.10.1980 to 31.10.1984**)
		Also held charge of Ministry of Defence (From 14.1.1980 to 15.1.1982)

Affairs

 $^{\ ^{*}}$ The date of dissolution of the Council of Ministers. ** The date of her assassination.

Portfolios alongwith the period thereof

7.

Gujral, Shri Inder Kumar Minister of State in the Department Parliamentary Affairs Communications

(From 18.3.1967 to 14.2.1969)

Minister of State in the Ministry of Information and Broadcasting and in the Department of Communications (From 14.2.1969 to 18.3.1971*)

Minister of State in the Ministry of Works, Housing and Urban Development

(From 18.3.1971 to 2.5.1971)

Minister of State in the Ministry of Works and Housing (From 2.5.1971 to 22.7.1972)

Minister of State in the Ministry of Information and Broadcasting (From 22.7.1972 to 28.6.1975)

Minister of State in the Ministry of **Planning**

(From 28.6.1975 to 12.5.1976)

Minister of External Affairs (From 6.12.1989 to 10.11.1990*)

Minister of External Affairs (From 1.6.1996 to 21.4.1997*)

Held charge as Minister of Water Resources

(From 1.6.1996 to 29.6.1996)

Prime Minister and in charge of Ministries/Departments of Atomic Energy, Personnel, Public Grievances and Pensions, Electronics, External Affairs, Jammu and Kashmir Affairs, Ocean Development and Space (From 21.4.1997 to 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of Non-Conventional Energy Sources & Minister of Coal (10.1.1998 to 19.3.1998)
		Also held charge of the Ministries of Urban Affairs and Employment, Petroleum and Natural Gas and Power (From 21.4.1997 to 9.6.1997)
		Ministries of Food, Civil Supplies and Consumer Affairs and Public Distribution (From 21.4.1997 to 23.4.1997)
		Ministries of Finance and Labour (From 21.4.1997 to 1.5.1997)
		Ministries of Health and Family Welfare and Planning and Programme Implementation (From 9.6.1997 to 19.3.1998*)
8.	Manmohan Singh, Dr.	Minister of Finance (From 21.6.1991 to 16.5.1996*)
	Prime Minister also incharge of Ministries/Departments not specifically allocated to the charge of any Minister (From 22.5.2004 to 23.5.2004)	
	Prime Minister and incharge of Ministries of Personnel, Public Grievances and Pensions; Ministry of Planning; Department of Atomic Energy; and Department of Space (From 23.5.2004 to 22.5.2009*)	
		Held charge of the Ministry of Coal and Mines (From 24.7.2004 to 27.11.2004)
		Held charge of the Ministry of Coal (From 2.3.2005 to 29.1.2006)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Held charge of the Ministry of Youth Affairs and Sports (From 25.05.2005 ¹ to 18.11.2005)
		Held charge of the Ministry of Overseas Indian Affairs (From 10.08.2005 to 18.11.2005)
		Held charge of the Ministry of Urban Development (From 1.11.2005 to 18.11.2005)
		Held charge of the Ministry of External Affairs (From 7.11.2005 to 24.10.2006)
		Held charge of the Ministry of Power (From 18.11.2005 to 29.1.2006)
		Held charge of the Ministry of Labour and Employment (From 24.8.2006 to 24.10.2006)
		Held charge of the Ministry of Coal (From 29.11.2006 to 22.5.2009*)
		Held charge of the Ministry of Environment and Forests (From 15.5.2007 to 22.5.2009*)
		Held charge of the Ministry of Information and Broadcasting (From 11.11.2008 to 22.5.2009*)
		Held charge of the Ministry of Finance (From 30.11.2008 to 24.1.2009)
		Prime Minister and also in-charge of the Ministries/Departments not specifically allocated to the charge of any Minister (From 22.5.2009 to 23.5.2009)

 $^{^1\}mbox{The}$ date of death of then Minister of Youth Affairs and Sports. *The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		(i) Prime Minister and incharge of Ministry of Personnel, Public Grievances & Pensions; (ii) Ministry of Planning; (iii) Department of Atomic Energy; and (iv) Department of Space. (from 23.5.2009 to 26.05.2014*)
		Held charge of the Ministry of Water Resources (from 31.5.2009 to 14.6.2009)
		Held charge of the Ministry of Culture (from 23.5.2009 to 19.1.2011)
		Held charge of the Ministry of Railways (from 19.5.2011 to 12.7.2011) (again from 15.6.2013 to 17.6.2013)
		Held charge of the Ministry of Finance (from 26.6.2012 to 31.7.2012)
		Held charge of the Ministry of Road Transport and Highways (from 15.6.2013 to 17.6.2013)
		Held charge of the Ministry of Housing and Urban Poverty Alleviation (from 15.6.2013 to 17.6.2013)
9.	Nanda, Shri Gulzari Lal	Minister of Planning (From 24.9.1951 to 13.5.1952*)
		Minister of Planning and River Valley Schemes (From 13.5.1952 to 6.6.1952)
		Minister of Planning and Irrigation and Power (From 6.6.1952 to 17.4.1957*)
		Minister of Labour and Employment and Planning (From 17.4.1957 to 10.4.1962*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Planning (From 10.4.1962 to 1.9.1963)
		Minister of Home Affairs (From 1.9.1963 to 27.5.1964*)
		Also held additional charge of the Ministry of Planning (From 1.9.1963 to 21.9.1963)
		Ministry of Labour and Employment (From 1.9.1963 to 24.1.1964)
		Prime Minister, Minister of External Affairs, Home Affairs and Atomic Energy (From 27.5.1964 to 9.6.1964)
		Minister of Home Affairs (From 9.6.1964 to 11.1.1966*)
		Prime Minister and Minister of Home Affairs and Atomic Energy (From 11.1.1966 to 24.1.1966*)
		Minister of Home Affairs (From 24.1.1966 to 9.11.1966 the date of his resignation)
		Minister of Railways (From 18.2.1970 to 18.3.1971*)
10.	Narasimha Rao, Shri P.V.	Minister of External Affairs (From 14.1.1980 to 19.7.1984)
		Minister of Home Affairs (From 19.7.1984 to 31.10.1984*) (Again from 31.10.1984 to 4.11.1984)
		Minister of Home Affairs and Planning (From 4.11.1984 to 31.12.1984*)
		Minister of Defence and Planning (From 31.12.1984 to 14.1.1985)

^{*}Date of resignation of the Council of Ministers of 15th Lok Sabha. *The date of dissolution of the Council of Ministers.

Sl.No.	Name

Portfolios alongwith the period thereof

Minister of Defence (From 14.1.1985 to 25.9.1985)

Minister of Human Resource Development

(From 25.9.1985 to 25.6.1988)

Also held charge of the Ministry of Home Affairs

(From 14.3.1986 to 12.5.1986)

Minister of Health and Family Welfare (From 24.6.1986 to 14.2.1988)

Minister of External Affairs (From 25.6.1988 to 2.12.1989)

Prime Minister and also incharge of the Ministries/Departments of Personnel, Public Grievances and Pensions, Science and Technology, Ocean Development, Electronics, Atomic Energy, Space and the additional charge of the Ministries and other subjects not allocated to any other Cabinet Minister or Minister of State (Independent Charge)

(From 21.6.1991 to 16.5.1996*)

Held charge of the Ministry of Defence (From 21.6.1991 to 26.6.1991)

Held charge of the Ministry of External Affairs

(From 31.3.1992 to 18.1.1993)

Held charge of the Ministry of Labour (From 30.7.1991 to 10.7.1992)

Held charge of the Ministry of Non-Conventional Energy Sources (From 2.7.1992 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Held charge of Ministry of Commerce (From 9.7.1992 to 18.1.1993)
		Held charge of Ministry of Law, Justice and Company Affairs (From 9.10.1992 to 16.5.1996*)
		Held charge of the Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 21.6.1991 to 18.1.1993)
		Assumed charge of the Ministry of Defence (From 5.3.1993 to 16.5.1996*)
		Held charge of the Ministry of Chemicals and Fertilizers (From 21.6.1991 to 17.2.1994)
		Held charge of the Ministry of Food (From 23.2.1994 to 10.2.1995)
		Assumed charge of the Department of Jammu and Kashmir Affairs (From 1.11.1994 to 16.5.1996*)
		Held charge of the Ministry of Health and Family Welfare (From 23.12.1994 to 11.6.1995)
		Also held charge of the Ministry of Human Resource Development (From 26.12.1994 to 10.2.1995)
		Held charge of the Ministry of Urban Affairs and Employment (From 10.9.1995 to 15.9.1995)
		Held charge of the Ministry of Railways (From 17.8.1995 to 16.5.1996*)
		Held charge of the Ministry of Human Resource Development (From 17.1.1996 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Held charge of the Ministry of Urban Affairs and Employment (From 21.2.1996 to 16.5.1996*)
		Held charge of the Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 20.2.1996 to 16.5.1996*)
		Held charge of the Ministry of Commerce (From 3.4.1996 to 16.5.1996*)
		Held charge of the Ministry of Rural Development (From 21.6.1991 to 11.6.1995)
11.	Nehru, Shri Jawaharlal	Prime Minister and Minister of External Affairs, Commonwealth Relations and Scientific Research (From 15.8.1947 to 26.1.1950*)
		Prime Minister and Minister of External Affairs and Scientific Research (From 26.1.1950 to 6.5.1950)
		Also held charge of the Ministries of Commerce and Industry and Supply (From 19.4.1950 to 6.5.1950*)
		Prime Minister and Minister of External Affairs and also incharge of Department of Scientific Research (From 6.5.1950 to 13.5.1952*)
		Also held charge of the Ministry of Industry and Supply (From 6.5.1950 to 13.5.1950)
		Also held charge of the Ministry of Commerce (From 6.5.1950 to 13.5.1950)
		Also held charge of the Ministry of Home Affairs (From 12.12.1950 to 26.12.1950)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Prime Minister and Minister of External Affairs (From 13.5.1952 to 17.4.1957*)
		Also held charge of the Ministry of Defence (From 27.1.1953 to 10.1.1955)
		Also held charge of the Ministry of Finance (From 24.7.1956 to 30.8.1956)
		Also held charge of the Ministry of Defence (From 30.1.1957 to 17.4.1957*)
		Prime Minister and Minister of External Affairs and Minister of Atomic Energy (From 17.4.1957 to 10.4.1962*)
		Also held charge of Ministry of Finance (From 14.2.1958 to 22.3.1958)
		Prime Minister and Minister of External Affairs and Minister of Atomic Energy (From 10.4.1962 to 27.5.1964* the date of his death)
		Also held charge of the Ministry of Defence (From 1.11.1962 to 21.11.1962)
12.	Shastri, Shri Lal Bahadur	Minister of Railways and Transport (From 13.5.1952 to 7.12.1956 the date of his resignation)
		Minister of Transport and Communications (From 17.4.1957 to 28.3.1958)
		Minister of Commerce and Industry (From 28.3.1958 to 5.4.1961)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also held charge of the Ministry of Home Affairs (From 26.2.1961 to 5.4.1961)
		Minister of Home Affairs (From 5.4.1961 to 10.4.1962) (Again from 10.4.1962 to 1.9.1963)
		Minister without portfolio (From 24.1.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964)
		Prime Minister and Minister of Atomic Energy (From 9.6.1964 to 11.1.1966* the date of his death)
		Also held charge of the Ministry of External Affairs (From 9.6.1964 to 19.7.1964)
13.	Singh, Shri Vishwanath Pratap	Deputy Minister in the Ministry of Commerce (From 10.10.1974 to 23.12.1976)
		Minister of State in the Ministry of Commerce (From 23.12.1976 to 24.3.1977*)
		Minister of Commerce (From 29.1.1983 to 7.9.1984)
		Also held charge of the Department of Supply (From 14.2.1983 to 7.9.1984)
		Also held charge of the Ministry of Industry (From 14.8.1984 to 7.9.1984)
		Minister of Finance (From 31.12.1984 to 14.1.1985)
		Minister of Finance, Commerce and Supply (From 14.1.1985 to 30.3.1985)

^{*}The date of dissolution of the Council of Ministers.

	,	
Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Finance and Commerce (From 30.3.1985 to 25.9.1985)
		Minister of Finance (From 25.9.1985 to 24.1.1987)
		Minister of Defence (From 24.1.1987 to 12.4.1987)
		Prime Minister (From 2.12.1989 to 10.11.1990*)
		Held charge of the Ministries/ Departments of Atomic Energy, Defence, Electronics, Environment and Forests, Ocean Development, Personnel, Public Grievances and Pensions, Science and Technology, Space and other subjects not allocated to any other Cabinet Minister or Minister of State with Independent Charge (From 5.12.1989 to 10.11.1990*)
		Also held charge of the Ministry of Agriculture (From 1.8.1990 to 10.11.1990*)
		Held charge of the Ministry of Environment and Forests (From 2.12.1989 to 23.4.1990)
14.	Vajpayee, Shri Atal Bihari	Minister of External Affairs (From 26.3.1977 to 28.7.1979*)
		Prime Minister and also incharge of Ministries/Departments of Chemicals and Fertilizers, Civil Supplies, Consumer Affairs and Public Distribution, Coal, Commerce,

Communications, Environment and Forests, Food, Food Processing Industries, Human Resource Development, Labour, Mines,

^{*}The date of dissolution of the Council of Ministers.

Sl.No. Name

Portfolios alongwith the period thereof

Non-Conventional Energy Sources, Personnel, Public Grievances and Pensions, Petroleum and Natural Gas, Planning and Programme Implementation, Power, Railways, Rural areas and Employment, Science and Technology, Steel, Surface Transport, Textiles, Water Resources, Atomic Energy, Electronics, Jammu Kashmir Affairs, Ocean Development, Space and other subjects not allocated to any other Cabinet Minister

(From 16.5.1996 to 1.6.1996*)

Also held charge of Ministry of External Affairs

(From 16.5.1996 to 21.5.1996)

Prime Minister and Minister-in-charge of Ministries of Agriculture, Personnel, Public Grievances and Pensions, Planning and Programme Implementation, Water Resources and Departments of Atomic Energy and Space

(From 19.3.1998 to 13.10.1999*)

Minister of Non-Conventional Energy Sources

(From 19.3.1998 to 3.2.1999)

Held charge of Ministry of External Affairs

(From 19.3.1998 to 5.12.1998)

Ministry of Food Processing Industries (From 19.3.1998 to 3.2.1999)

Department of Ocean Development and Department of Electronics (From 19.3.1998 to 3.2.1999)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Department of Jammu and Kashmir Affairs (From 19.3.1998 to 22.5.1998)
		Ministry of Information and Broadcasting (From 11.10.1998 to 5.12.1998)
		Ministry of Communications (From 11.10.1998 to 5.12.1998) (Again from 8.6.1999 to 13.10.1999*)
		Ministry of Railways (From 5.8.1999 to 6.8.1999)
		Ministry of Health and Family Welfare (From 14.8.1999 to 16.8.1999)
		Prime Minister and held charge of Ministry of Planning and Departments of Atomic Energy and Space (From 13.10.1999 to 22.5.2004*)
		Held charge of Ministry of Personnel, Public Grievances and Pensions (From 13.10.1999 to 29.1.2003)
		Ministry of Agriculture (From 13.10.1999 to 22.11.1999) (Again from 3.3.2000 to 6.3.2000)
		Ministry of Information Technology (From 13.10.1999 to 22.11.1999)
		Ministry of Labour (From 13.10.1999 to 22.11.1999)
		Department of Ocean Development (From 13.10.1999 to 22.11.1999)
		Ministry of Mines and Minerals (From 4.3.2000 to 6.3.2000)
		Ministry of Power (From 23.8.2000 to 30.9.2000)

^{*}The date of dissolution of the Council of Ministers.

<u> </u>
ace Transport 1000 to 7.11.2000)
nce 101 to 18.3.2001)
vays 101 to 20.3.2001)
and Mines 102 to 1.7.2002)
stics and Programme 999 to 18.11.2001) 02 to 22.5.2004)
er 102 to 26.8.2002)
onment and Forests .003 to 9.1.2004)
-Conventional Energy 003 to 9.1.2004)
Processing Industries 104 to 17.1.2004)
)

*PART II
DEPUTY PRIME MINISTERS

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Advani, Shri L.K.	Minister of Information and Broadcasting (From 26.3.1977 to 28.7.1979*)
		Minister of Home Affairs (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 29.6.2002)
		Deputy Prime Minister and Minister of Home Affairs (From 29.6.2002 to 1.7.2002)
		Deputy Prime Minister and Minister of Home Affairs and Minister of Coal and Mines (From 1.7.2002 to 26.8.2002)
		Deputy Prime Minister and Minister of Home Affairs (From 26.8.2002 to 29.1.2003)
		Deputy Prime Minister, Minister of Home Affairs and Minister of Personnel, Public Grievances and Pensions (From 29.1.2003 to 22.5.2004*)
2.	Chavan, Shri Y.B.	Minister of Defence
		(From 21.11.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.11.1966*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Home Affairs (From 13.11.1966 to 13.3.1967*) (Again from 13.3.1967 to 27.6.1970)
		Minister of Finance (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 10.10.1974)
		Minister of External Affairs (From 10.10.1974 to 24.3.1977*)
		Deputy Prime Minister and Minister of Home Affairs (From 28.7.1979 to 14.1.1980*)
3.	Devi Lal, Shri	Deputy Prime Minister and Minister of Agriculture (From 2.12.1989 to 1.8.1990)
		Deputy Prime Minister (From 10.11.1990 to 21.11.1990)
		Deputy Prime Minister and Minister of Agriculture and Tourism (From 21.11.1990 to 21.6.1991*)
4.	Jagjivan Ram, Shri	Minister of Labour (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1952*)
		Minister of Communications (From 13.5.1952 to 7.12.1956)
		Minister of Railways and Transport (From 7.12.1956 to 17.4.1957*)
		Minister of Railways (From 17.4.1957 to 10.4.1962*)
		Minister of Transport and Communications (From 10.4.1962 to 31.8.1963 the date of his resignation)
		Minister of Labour, Employment and Rehabilitation (From 24.1.1966 to 13.3.1967*)

 $^{{}^{*}\}text{The date of dissolution of the Council of Ministers.}$

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Food and Agriculture (From 13.3.1967 to 27.6.1970)
		Also Minister of Labour, Employment and Rehabilitation (From 15.11.1969 to 18.2.1970)
		Minister of Defence (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 10.10.1974)
		Minister of Agriculture and Irrigation (From 10.10.1974 to 2.2.1977)
		Minister of Defence (From 28.3.1977 to 24.1.1979)
		Held charge of the Ministry of Shipping and Transport (From 7.6.1977 to 17.6.1977)
		Deputy Prime Minister and Minister of Defence (From 24.1.1979 to 28.7.1979*)
5.	Patel, Sardar Vallabhbhai	Deputy Prime Minister and Minister of Home Affairs, States and Information and Broadcasting (From 15.8.1947 to 26.1.1950*)
		Deputy Prime Minister and Minister of Home Affairs (From 26.1.1950 to 6.5.1950 and again from 6.5.1950 to 15.12.1950)

^{*}The date of dissolution of the Council of Ministers.

PART IIICABINET MINISTERS

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Abdullah, Dr. Farooq	Minister of New and Renewable Energy (from 28.5.2009 to 26.05.2014*)
2.	Ahmed, Shri Fakhruddin Ali	Minister of Irrigation and Power (From 29.1.1966 to 13.11.1966)
		Minister of Education (From 13.11.1966 to 13.3.1967*)
		Minister of Industrial Development and Company Affairs (From 13.3.1967 to 14.2.1969)
		Minister of Industrial Development, Internal Trade and Company Affairs (From 14.2.1969 to 27.6.1970)
		Minister of Food and Agriculture (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Minister of Agriculture (From 2.5.1971 to 3.7.1974)
3.	Aiyar, Shri Mani Shankar	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Petroleum and Natural Gas (From 23.5.2004 to 29.1.2006)
		Minister of Panchayati Raj (From 23.5.2004 to 22.5.2009*)
		Also Minister of Youth Affairs and Sports (From 29.1.2006 to 6.4.2008)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of Development of North- Eastern Region (From 24.10.2006 to 22.5.2009*)
4.	Ajit Singh, Shri	Minister of Industry (From 6.12.1989 to 10.11.1990*)
		Minister of Food (From 10.2.1995 to 16.5.1996*)
		Minister of Agriculture (From 22.7.2001 to 24.5.2003, the date of his resignation)
		Minister of Civil Aviation (from 18.12.2011 to 26.05.2014*)
5.	Alagiri, Shri M.K.	Minister of Chemicals and Fertilizers (from 28.5.2009 to 20.3.2013)
6.	Ambedkar, Dr. B.R.	Minister of Law (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 8.5.1950*) (Again from 8.5.1950 to 11.10.1951)
7.	Amrit Kaur, Rajkumari	Minister of Health (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 23.6.1950 to 13.5.1952*) (Again from 13.5.1952 to 16.4.1957*)
8.	Ananth Kumar, Shri	Minister of Civil Aviation (From 19.3.1998 to 13.10.1999*)
		Also Minister of Tourism (From 30.1.1999 to 13.10.1999*)
		Minister of Culture, Youth Affairs and Sports (From 13.10.1999 to 2.2.2000)
		Minister of Tourism with additional charge of the Department of Culture (From 2.2.2000 to 27.5.2000)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Tourism and Culture (From 27.5.2000 to 1.9.2001)
		Minister of Urban Development and Poverty Alleviation (From 1.9.2001 to 12.7.2003 date of his resignation)
		Held charge of the Ministry of Rural Development (From 6.4.2003 to 24.5.2003)
9.	Ansari, Shri Ziaur Rehman	Deputy Minister in the Ministry of Industrial Development (From 6.2.1973 to 10.10.1974)
		Deputy Minister in the Ministry of Industry and Civil Supplies (From 10.10.1974 to 3.1.1976)
		Deputy Minister in the Ministry of Petroleum (From 3.1.1976 to 24.3.1977*)
		Minister of State in the Ministry of Commerce (From 3.3.1980 to 19.10.1980)
		Minister of State in the Ministry of Irrigation (From 19.10.1980 to 29.1.1983)
		Minister of State in the Ministry of Shipping and Transport (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State (Independent Charge) of the Ministry of Shipping and Transport (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Environment and Forests (From 25.9.1985 to 14.2.1988)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 14.2.1988 to 25.6.1988)
		Minister of Environment and Forests (From 25.6.1988 to 2.12.1989*)
10.	Antony, Shri A.K.	Minister of Civil Supplies, Consumer Affairs and Public Distribution (From 18.1.1993 to 8.2.1995)
		Minister of Defence (From 24.10.2006 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 23.5.2009)
		Minister of Defence (From 23.5.2009 to 26.05.2014*)
11.	Antulay, Shri A.R.	Minister of Health and Family Welfare (From 11.6.1995 to 16.5.1996*)
		Also Minister of Water Resources (From 7.2.1996 to 16.5.1996*)
		Minister of Minority Affairs (From 29.1.2006 to 22.5.2009*)
12.	Arjun Singh, Shri	Minister of Commerce (From 15.11.1985 to 20.1.1986)
		Minister of Communications (From 22.10.1986 to 14.2.1988)
		Minister of Human Resource Development (From 21.6 1991 to 24.12.1994)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Human Resource Development (From 23.5.2004 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
13.	Arunachalam, Shri M.	Minister of State in the Ministry of Industry—Department of Industrial Development (From 25.9.1985 to 2.12.1989*)
		Minister of State in the Ministry of Urban Development (From 21.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Small Scale Industries and Agro Industries (From 18.1.1993 to 3.4.1996)
		Minister of Urban Affairs and Employment (From 1.6.1996 to 29.6.1996)
		Minister of Labour (From 29.6.1996 to 21.4.1997*) (Again from 1.5.1997 to 9.6.1997)
		Minister of Chemicals and Fertilizers (From 9.6.1997 to 19.3.1998*)
14.	Ayyangar, Shri N. Gopalaswami	Minister without Portfolio (From 26.9.1947 to 17.8.1948)
		Minister of Transport and Railways (From 17.8.1948 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 26.12.1950*)
		Also held charge of Ministry of States (From 12.12.1950 to 26.12.1950) (Again from 26.12.1950 to 13.5.1952)
		Minister of Defence (From 13.5.1952 to 10.2.1953, the date of his death)
15.	Azad, Maulana Abul Kalam	Minister of Education (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1952*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Education and Natural Resources and Scientific Research (From 13.5.1952 to 17.4.1957*)
		Minister of Education and Scientific Research (From 17.4.1957 to 22.2.1958, the date of his death)
16.	Azad, Shri Ghulam Nabi	Minister of State in the Ministry of Parliamentary Affairs (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Parliamentary Affairs and in the Ministry of Tourism (From 25.9.1985 to 12.5.1986)
		Minister of State in the Ministry of Home Affairs (Department of States) (From 12.5.1986 to 22.10.1986)
		Minister of State in the Ministry of Food and Civil Supplies (From 22.10.1986 to 7.9.1987)
		Minister of Parliamentary Affairs (From 21.6.1991 to 18.1.1993)
		Also Minister of Civil Aviation and Tourism (From 9.1.1993 to 16.5.1996*)
		Also Minister of Parliamentary Affairs (From 18.1.1996 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Parliamentary Affairs and Minister of Urban Development (From 23.5.2004 to 1.11.2005, the date of his resignation)
		Minister without portfolio (from 22.5.2009 to 28.5.2009)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Health and Family Welfare (from 28.5.2009 to 26.05.2014*)
		Minister of Water Resources (from 1.2.2014 to 26.05.2014*)
17.	Baalu, Shri T.R.	Minister of State in the Ministry of Petroleum and Natural Gas (From 6.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
		Minister of State in the Ministry of Non-Conventional Energy Sources (From 10.1.1998 to 19.3.1998*)
		Minister of Environment and Forests (From 13.10.1999 to 21.12.2003, the date of his resignation)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Road Transport and Highways (From 23.5.2004 to 3.9.2004)
		Also Minister of Shipping (From 25.5.2004 to 3.9.2004)
		Minister of Shipping, Road Transport and Highways (From 3.9.2004 to 22.5.2009*)
18.	Badal, Shri Prakash Singh	Minister of Communications (From 26.3.1977 to 27.3.1977)
		Minister of Agriculture and Irrigation (From 28.3.1977 to 17.6.1977)
19.	Bahuguna, Shri H.N.	Minister of State in the Ministry of Communications (From 2.5.1971 to 8.11.1973)
		Minister of Chemicals and Fertilizers (From 28.3.1977 to 29.3.1977)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Petroleum, Chemicals and Fertilizers (From 29.3.1977 to 15.7.1979)
		Minister of Finance (From 28.7.1979 to 19.10.1979)
20.	Baishya, Shri Birendra Prasad	Minister of Steel and Mines (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
21.	Baldev Singh, Sardar	Minister of Defence (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1952*)
22.	Banerjee, Km. Mamata	Minister of State in the Ministry of Human Resource Development— Department of Youth Affairs and Sports (From 21.6.1991 to 17.1.1993)
		Also Minister of State in the Department of Women and Child Development (From 26.6.1991 to 17.1.1993)
		Minister of Railways (From 13.10.1999 to 16.3.2001, the date of her resignation)
		Minister without portfolio (From 8.9.2003 to 9.1.2004)
		Minister of Coal and Mines (From 9.1.2004 to 22.5.2004*)
		Minister without portfolio (From 22.5.2009 to 23.5.2009)
		Minister of Railways (From 23.5.2009 to 19.5.2011)
23.	Bansal, Shri Pawan Kumar	Minister of State in the Ministry of Finance (From 29.1.2006 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Parliamentary Affairs (From 6.4.2008 to 22.5.2009*)
		Minister of Parliamentary Affairs (From 28.5.2009 to 28.10.2012)
		Minister of Water Resources (From 14.6.2009 to 19.1.2011) Again from 12.7.2011 to 28.10.2012)
		Minister of Science and Technology and Minister of Earth Sciences (From 19.1.2011 to 12.7.2011)
		Minister of Railways (From 28.10.2012 to 11.5.2013)
24.	Bansi Lal, Shri	Minister without portfolio (From 1.12.1975 to 21.12.1975)
		Minister of Defence (From 21.12.1975 to 24.3.1977*)
		Minister of Railways (From 31.12.1984 to 25.9.1985)
		Minister of Transport (From 25.9.1985 to 4.6.1986)
25.	Barnala, Shri Surjit Singh	Minister of Agriculture and Irrigation (From 18.6.1977 to 28.7.1979*)
		Minister of Chemicals and Fertilizers (From 19.3.1998 to 13.10.1999*)
		Held additional charge of the Ministry of Food and Consumer Affairs (From 19.3.1998 to 13.10.1999*)
26.	Barooah, Shri Dev Kant	Minister of Petroleum and Chemicals (From 5.2.1973 to 10.10.1974*)
27.	Bhabha, Dr. C.H.	Minister of Commerce (From 15.8.1947 to 6.4.1948)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
28.	Bhagat, Shri B.R.	Deputy Minister in the Ministry of Finance (From 4.1.1956 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962) (Again from 16.4.1962 to 21.9.1963)
		Minister of State in the Ministry of Planning (From 21.9.1963 to 27.5.1964*)
		Also assumed the Office of Minister of State in the Ministry of Finance in the Department of Coordination (From 1.11.1963 to 27.5.1964*)
		Minister of State for Planning (From 27.5.1964 to 9.6.1964) (Again from 9.6.1964 to 13.6.1964)
		Minister of State for Planning and Minister of State in the Ministry of Finance in the Department of Coordination (From 13.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of Finance (From 24.1.1966 to 13.3.1967*)
		Minister of State in the Ministry of Defence (From 13.3.1967 to 14.11.1967)
		Minister of State in the Ministry of External Affairs (From 14.11.1967 to 14.2.1969)
		Minister of Foreign Trade and Supply (From 14.2.1969 to 4.11.1969)
		Minister of Foreign Trade (From 4.11.1969 to 27.6.1970)
		Minister of Steel and Heavy Engineering (From 27.6.1970 to 18.3.1971*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of External Affairs (From 25.9.1985 to 12.5.1986)
29.	Bhagat, Shri H.K.L.	Minister of State in the Ministry of Works and Housing (From 1.12.1975 to 24.3.1977*)
		Minister of State in the Ministry of Works and Housing and in the Department of Parliamentary Affairs (From 2.9.1982 to 14.2.1983)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting and Minister of State in the Department of Parliamentary Affairs
		(From 14.2.1983 to 31.10.1984*) (Again from 4.11.1984 to 1.12.1984*)
		Minister of Parliamentary Affairs (From 31.12.1984 to 2.12.1989*)
		Also held charge of the Ministry of Tourism (From 25.9.1985 to 12.5.1986)
		Also Minister of Food and Civil Supplies (From 12.5.1986 to 14.2.1988)
		Held additional charge of the Ministry of Information and Broadcasting (From 14.2.1988 to 2.12.1989*)
30.	Bhajan Lal, Shri	Minister of Environment and Forests (From 22.10.1986 to 14.2.1988)
		Minister of Agriculture (From 14.2.1988 to 2.12.1989*)
31.	Bhardwaj, Shri H.R.	Minister of State in the Ministry of Law and Justice (From 31.12.1984 to 2.12.1989*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Planning and Programme Implementation (From 21.6.1991 to 2.7.1992)
		Minister of State in the Ministry of Law, Justice and Company Affairs (From 2.7.1992 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Law and Justice (From 23.5.2004 to 22.5.2009*)
32.	Bhuria, Shri Kantilal	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Agriculture and in the Ministry of Consumer Affairs, Food and Public Distribution (From 23.5.2004 to 22.5.2009*)
		Minister of Tribal Affairs (From 28.5.2009 to 12.7.2011)
33.	Biswas, Shri C.C.	Minister of State—in terms of the Indo- Pakistan Agreement of 8.4.1950 (From 13.5.1950 to 13.5.1952*)
		Minister of Law and Minority Affairs— in terms of Indo-Pak Agreement of 8 April, 1950 (From 13.5.1952 to 17.4.1957*)
34.	Bommai, Shri S.R.	Minister of Human Resource Development (From 1.6.1996 to 21.4.1997*)
		Also Minister of Coal (From 1.6.1996 to 29.6.1996)
		Minister of Human Resource Development (From 21.4.1997 to 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
35.	Brahm Prakash, Shri	Minister of Agriculture and Irrigation (From 30.7.1979 to 14.1.1980*)
36.	Buta Singh, Shri	Deputy Minister in the Ministry of Railways (From 10.10.1974 to 23.12.1976)
		Deputy Minister in the Ministry of Commerce (From 23.12.1976 to 24.3.1977*)
		Minister of State in the Ministry of Shipping and Transport (From 8.6.1980 to 15.1.1982)
		Minister of State (Independent Charge) of the Ministry of Supply and Rehabilitation (From 15.1.1982 to 2.9.1982)
		Minister of State (Independent Charge) of the Ministry of Supply and Sports (From 2.9.1982 to 29.1.1983)
		Minister of Parliamentary Affairs, Sports and Works and Housing (From 29.1.1983 to 31.10.1984*) (Again from 31.10.1984 to 31.12.1984*)
		Minister of Agriculture and Rural Development (From 31.12.1984 to 25.9.1985)
		Minister of Agriculture (From 25.9.1985 to 12.5.1986)
		Minister of Home Affairs (From 12.5.1986 to 2.12.1989*)
		Minister of Civil Supplies, Consumer Affairs and Public Distribution (From 10.2.1995 to 20.2.1996*)
		Minister of Communications (From 19.3.1998 to 20.4.1998)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
37.	Chagla, Shri M.C.	Minister of Education (From 21.11.1963 to 27.5.1964) (Again from 27.5.1964 to 9.6.1964) (Again from 9.6.1964 to 11.1.1966) (Again from 11.1.1966 to 24.1.1966) (Again from 24.1.1966 to 13.11.1966)
		Minister of External Affairs (From 13.11.1966 to 13.3.1967) (Again from 13.3.1967 to 5.9.1967)
38.	Chattopadhyaya, Prof. D.P.	Minister of State in the Ministry of Health and Family Planning (From 2.5.1971 to 5.2.1973)
		Also Minister of State in the Ministry of Works and Housing (From 2.8.1972 to 5.2.1973)
		Minister of State in the Ministry of Commerce (From 5.2.1973 to 23.12.1976)
		Minister of Commerce (From 23.12.1976 to 24.3.1977*)
39.	Chaudhuri, Shri A.B.A. Ghani Khan	Minister of Energy (From 14.1.1980 to 2.9.1982)
		Also held charge of Ministry of Irrigation (From 16.1.1980 to 9.6.1980)
		Also held temporary charge of the Department of Coal (From 16.1.1980 to 15.1.1982)
		Minister of Railways (From 2.9.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of Programme Implementation (From 25.9.1985 to 4.5.1987)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
40.	Chaudhuri, Shri Sachindra	Minister of Finance (From 1.1.1966 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)
41.	Chavan, Shri S.B.	Minister of Education and Social Welfare (From 19.10.1980 to 8.8.1981)
		Minister of Planning (From 8.8.1981 to 19.7.1984)
		Minister without portfolio—to assist the Prime Minister in External Affairs (From 19.7.1984 to 2.8.1984)
		Minister of Defence (From 2.8.1984 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of Home Affairs (From 31.12.1984 to 11.3.1986)
		Minister of Finance (From 25.6.1988 to 2.12.1989*)
		Minister of Home Affairs (From 21.6.1991 to 16.5.1996*)
42.	Chetty, Shri R.K. Shanmukham	Minister of Finance (From 15.8.1947 to 17.8.1948)
43.	Chidambaram, Shri P.	Deputy Minister in the Ministry of Commerce (From 25.9.1985 to 27.9.1985)
		Deputy Minister in the Ministry of Commerce—Department of Textiles (From 27.9.1985 to 4.10.1985)
		Deputy Minister in the Ministry of Personnel, Administrative Reforms, Training, Public Grievances and Pensions (From 4.10.1985 to 20.1.1986)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Personnel, Public Grievances and Pensions (From 20.1.1986 to 24.6.1986)
		Minister of State in the Ministry of Personnel, Public Grievances and Pensions and Minister of State in the Ministry of Home Affairs (From 24.6.1986 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Commerce (From 21.6.1991 to 9.7.1992)
		Minister of State (Independent Charge) of the Ministry of Commerce (From 10.2.1995 to 3.4.1996)
		Minister of Finance (From 1.6.1996 to 21.4.1997*)
		Also Minister of Law, Justice and Company Affairs (From 1.6.1996 to 29.6.1996)
		Also Minister of Department of Company Affairs (From 29.6.1996 to 26.7.1996)
		Minister of Finance (From 1.5.1997 to 19.3.1998*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Finance (From 23.5.2004 to 30.11.2008)
		Minister of Home Affairs (From 30.11.2008 to 22.5.2009)
		Minister without portfolio (From 22.5.2009 to 23.5.2009)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Home Affairs (From 23.5.2009 to 31.7.2012)
		Minister of Finance (From 31.7.2012 to 26.05.2014*)
44.	Choudhury, Dr. Moinul Haque	Minister of Industrial Development (From 18.3.1971 to 22.7.1972)
		Also held temporary charge of Department of Supply (From 18.3.1971 to 2.5.1971)
45.	Chunder, Shri Pratap Chandra	Minister of Education and Social Welfare and Culture
		(From 26.3.1977 to 28.7.1979*)
46.	Dandavate, Prof. Madhu	Minister of Railways (From 26.3.1977 to 28.7.1979*)
		Minister of Finance (From 5.12.1989 to 10.11.1990*)
47.	Dasappa, Shri H.C.	Minister of Railways (From 21.9.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Irrigation and Power (From 9.6.1964 to 19.7.1964)
		Minister of Industry and Supply (From 19.7.1964 to 29.10.1964)
48.	Dasmunshi, Priyaranjan	Minister of State in the Ministry of Commerce (From 22.10.1986 to 2.12.1989*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Water Resources (From 23.5.2004 to 18.11.2005)
		Held additional charge of Ministry of Parliamentary Affairs (From 1.11.2005 to 6.4.2008)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Information and Broadcasting (From 18.11.2005 to 11.11.2008)
		Minister without portfolio (From 11.11.2008 to 22.5.2009*)
49.	Deb, Shri Santosh Mohan	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministries of Heavy Industries and Public Enterprises (From 23.5.2004 to 29.1.2006)
		Minister of State (Independent Charge) of the Ministry of Water Resources (From 18.11.2005 to 29.1.2006)
		Minister of Heavy Industries and Public Enterprises (From 29.1.2006 to 22.5.2009*)
50.	Deo, Shri V. Kishore Chandra	Minister of State in the Ministry of Steel, Mines and Coal (From 4.8.1979 to 14.1.1980*)
		Minister of Tribal Affairs and Minister of Panchayati Raj (From 12.7.2011 to 26.05.2014*)
51.	Deora, Shri Murli	Minister of Petroleum and Natural Gas (From 29.1.2006 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Petroleum and Natural Gas (From 28.5.2009 to 19.1.2011)
		Minister of Corporate Affairs (From 19.1.2011 to 12.7.2011)
52.	Desai, Shri Hitendra	Minister of Works and Housing (From 23.12.1976 to 24.3.1977)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Commerce and Civil Supplies (From 30.7.1979 to 14.1.1980*)
53.	Desai, Shri Khandubhai Kasanji	Minister of Labour (From 10.9.1954 to 16.4.1957*)
54.	Deshmukh, Dr. C.D.	Minister without portfolio (From 29.5.1950 to 1.6.1950)
		Minister of Finance (From 1.6.1950 to 13.5.1952*) (Again from 13.5.1952 to 1.8.1956)
55.	Deshmukh, Shri Vilasrao	Minister of Heavy Industries and Public Enterprises (From 28.5.2009 to 19.1.2011)
		Minister of Rural Development, Minister of Panchayati Raj (From 19.1.2011 to 12.7.2011)
		Minister of Science and Technology, Minister of Earth Sciences (From 12.7.2011 to 10.8.2012)
		Minister of Micro, Small and Medium Enterprises (From 26.6.2012 to 10.8.2012)
		Minister without portfolio (From 10.8.2012 to 14.8.2012**)
56.	Dhar, Shri D.P.	Minister of Planning (From 23.7.1972 to 31.12.1974)
57.	Dharia, Shri Mohan	Minister of State in the Ministry of Planning (From 2.5.1971 to 10.10.1974)
		Minister of State in the Ministry of Works and Housing (From 10.10.1974 to 2.3.1975)
		Minister of Commerce and Civil Supplies and Cooperation (From 26.3.1977 to 28.7.1979*)

^{*}The date of dissolution of the Council of Ministers. **Date of his death.

Sl.No.	Name	Portfolios alongwith the period thereof
58.	Dhillon, Dr. G.S.	Minister of Shipping and Transport (From 1.12.1975 to 24.3.1977*)
		Minister of Agriculture (From 12.5.1986 to 14.2.1988)
59.	Dhindsa, Shri Sukhdev Singh	Minister of Works and Estates (From 22.11.1999 to 26.11.1999)
		Minister of Urban Employment and Poverty Alleviation (From 26.11.1999 to 27.5.2000)
		Minister of Youth Affairs and Sports (From 2.2.2000 to 7.11.2000)
		Minister of Mines (From 27.5.2000 to 7.11.2000)
		Minister of Chemicals and Fertilisers (From 7.11.2000 to 22.5.2004*)
60.	Dikshit, Shri Uma Shankar	Minister of Works and Housing (From 2.5.1971 to 5.2.1973)
		Also held temporary charge of Ministry of Health and Family Planning (From 19.5.1971 to 5.2.1973)
		Minister of Home Affairs (From 5.2.1973 to 10.10.1974)
		Minister without portfolio (From 10.10.1974 to 10.2.1975)
		Minister of Shipping and Transport (From 10.2.1975 to 1.12.1975)
61.	Dinesh Singh, Shri	Deputy Minister in the Ministry of External Affairs (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of External Affairs (From 24.1.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Commerce (From 13.3.1967 to 14.2.1969)
		Minister of External Affairs (From 14.2.1969 to 27.6.1970)
		Minister of Industrial Development and Internal Trade (From 27.6.1970 to 18.3.1971*)
		Minister of Water Resources (From 14.2.1988 to 25.6.1988)
		Minister of Commerce and Industry (From 25.6.1988 to 2.12.1989*)
		Minister of External Affairs (From 18.1.1993 to 10.2.1995)
		Minister without portfolio (From 10.2.1995 to 30.11.1995 the date of his death)
62.	Dubey, Shri Bindeshwary	Minister of Law and Justice (From 14.2.1988 to 25.6.1988)
		Minister of Labour (From 25.6.1988 to 2.12.1989)
63.	Dutt, Shri Sunil	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Youth and Sports (From 23.5.2004 to 25.5.2005, the date of his death)
64.	Fazlur Rahman, Shri	Minister of State in the Ministry of Energy (From 14.8.1977 to 26.1.1979)
		Minister of State in the Ministry of Planning (From 26.1.1979 to 15.7.1979)
		Minister of Labour (From 30.7.1979 to 14.1.1980*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Held additional charge of Waqfs (From 7.12.1979 to 14.1.1980*)
65.	Fernandes, Shri George	Minister of Communications (From 28.3.1977 to 6.7.1977)
		Minister of Industry (From 6.7.1977 to 15.7.1979)
		Minister of Railways (From 6.12.1989 to 10.11.1990)
		Also Minister of Kashmir Affairs (From 13.3.1990 to 30.5.1990)
		Minister of Defence (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 16.3.2001, the date of his resignation)
		Minister of Defence (From 15.10.2001 to 22.5.2004*)
66.	Fernandes, Shri Oscar	Parliamentary Secretary to the Prime Minister (From 1.1.1985 to 25.9.1985)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation (From 23.5.2004 to 29.1.2006)
		Minister of State (Independent Charge) of the Ministry of Youth Affairs and Sports and of the Ministry of Overseas Indian Affairs (From 18.11.2005 to 29.1.2006)
		Minister of State (Independent Charge) without portfolio (From 29.1.2006 to 24.10.2006)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Labour and Employment (From 24.10.2006 to 3.3.2009, the date of his resignation)
		Minister of Road Transport and Highways (From 17.6.2013 to 26.05.2014*)
		Minister of Labour and Employment (From 15.12.2013 to 26.05.2014*)
67.	Fotedar, Shri M.L.	Minister of Steel and Mines (From 25.7.1987 to 6.2.1988)
		Minister of Steel and Mines and Minister of Labour (From 6.2.1988 to 14.2.1988)
		Minister of Steel and Mines (From 14.2.1988 to 2.12.1989*)
		Held temporary charge of the Ministry of Energy in the absence of Shri Vasant Sathe who proceeded to USA for Coronary treatment (From 3.9.1988 to 19.9.1988)
		Minister of Health and Family Welfare (From 21.6.1991 to 17.1.1993)
68.	Gadgil, Shri N.V.	Minister of Works, Mines and Power (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950) (Again from 6.5.1950 to 26.12.1950)
		Minister of Works, Production and Supply (From 26.12.1950 to 13.5.1952*)
69.	Gafoor, Shri Abdul	Minister of Works and Housing (From 31.12.1984 to 25.9.1985)
		Minister of Urban Development (From 25.9.1985 to 22.10.1986)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
70.	Geete, Shri Anant Gangaram	Minister of State in the Ministry of Finance and Company Affairs (From 1.7.2002 to 26.8.2002)
		Minister of Power (From 26.8.2002 to 22.5.2004*)
71.	Gill, Shri M.S.	Minister of State (Independent Charge) of the Ministry of Youth Affairs and Sports (From 6.4.2008 to 22.5.2009*)
		Minister of Youth Affairs and Sports (From 28.5.2009 to 19.1.2011)
		Minister of Statistics and Programme Implementation (From 19.1.2011 to 12.7.2011)
72.	Giri, Shri V.V.	Minister of Labour (From 13.5.1952 to 7.9.1954, the date of his resignation)
73.	Gokhale, Shri H.R.	Minister of Law and Justice (From 18.3.1971 to 5.2.1973)
		Also held charge of the Ministry of Petroleum and Chemicals (From 29.1.1972 to 5.2.1973)
		Minister of Law, Justice and Company Affairs (From 5.2.1973 to 24.3.1977*)
74.	Goswami, Shri Dinesh	Minister of Steel and Mines with additional charge of the Ministry of Law and Justice (From 6.12.1989 to 10.11.1990*)
75.	Goyal, Shri Vedprakash	Minister of Shipping (From 1.9.2001 to 29.1.2003, the date of his resignation)
76.	Gupta, Shri Inderjit	Minister of Home Affairs (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
77.	Gupta, Shri Prem Chand	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Company Affairs (From 23.5.2004 to 29.1.2006)
		Minister of Company Affairs (From 29.1.2006 to 9.5.2007**)
		Minister of Corporate Affairs (From 9.5.2007 to 22.5.2009*)
78.	Gurupadaswamy, Shri M.S.	Minister of State in the Department of Atomic Energy (From 18.3.1967 to 5.6.1967)
		Minister of State in the Ministry of Food, Agriculture, Community Development and Cooperation (From 5.6.1967 to 17.10.1969)
		Minister of Petroleum and Chemicals (From 6.12.1989 to 10.11.1990)
79.	Handique, Shri B.K.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Defence (From 23.5.2004 to 29.1.2006)
		Also Minister of State in the Ministry of Parliamentary Affairs (From 23.5.2004 to 6.4.2008)
		Minister of State in the Ministry of Chemicals and Fertilizers (From 29.1.2006 to 22.5.2009*)
		Also Minister of State in the Ministry of Mines (From 6.4.2008 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)

^{*} The date of dissolution of the Council of Ministers.

** w.e.f. 9.5.2007, the M/o Company Affairs had been renamed as M/o Corporate Affairs and Shri Prem Chand Gupta continued as its Minister.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Mines (From 28.5.2009 to 19.1.2011)
		Minister of Development of North- Eastern Region (From 28.5.2009 to 12.7.2011)
80.	Hanumanthaiya, Shri K.	Minister of Law and Social Welfare (From 26.6.1970 to 18.3.1971*)
		Minister of Railways (From 18.3.1971 to 22.7.1972)
81.	Hathi, Shri Jai Sukh Lal	Deputy Minister in the Ministry of Irrigation and Power (From 12.9.1952 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962*)
		Minister of Labour in the Ministry of Labour and Employment (From 16.4.1962 to 15.11.1962)
		Minister of Supply in the Ministry of Economic and Defence Coordination (From 15.11.1962 to 1.9.1963)
		Minister of State in the Ministry of Supply (From 1.9.1963 to 10.3.1964)
		Also held charge of the Department of Technical Development (From 1.9.1963 to 10.3.1964)
		Also assumed the Office of Minister of State in the Ministry of Labour and Employment (From 4.9.1963 to 24.1.1964)
		Minister of State in the Ministry of Home Affairs (From 10.3.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.11.1966*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also assumed the office of the Minister of Defence Supplies in the Ministry of Defence (From 29.10.1965 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.11.1966*)
		Minister of State in the Ministry of Defence (From 13.11.1966 to 13.3.1967*)
		Minister of Labour and Rehabilitation (From 13.3.1967 to 15.11.1969)
82.	Hegde, Shri Ramakrishna	Minister of Commerce (From 19.3.1998 to 13.10.1999*)
83.	Hussain, Shri Syed Shahnawaz	Minister of State in the Department of Food Processing Industries in the Ministry of Agriculture (From 13.10.1999 to 27.5.2000)
		Minister of State in the Ministry of Youth Affairs and Sports (From 27.5.2000 to 30.9.2000)
		Minister of State in the Ministry of Human Resource Development (From 30.9.2000 to 8.2.2001)
		Minister of State (Independent Charge) of the Ministry of Coal (From 8.2.2001 to 1.9.2001)
		Minister of Civil Aviation (From 1.9.2001 to 24.5.2003)
		Minister of Textiles (From 24.5.2003 to 22.5.2004)
84.	Ibrahim, Shri C.M.	Minister of Civil Aviation and Tourism (From 1.6.1996 to 29.6.1996)
		Also Minister of Information and Broadcasting (From 1.6.1996 to 21.4.1997*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Civil Aviation (From 29.6.1996 to 19.3.1998*)
		Held charge of the Ministry of Information and Broadcasting (From 21.4.1997 to 1.5.1997)
85.	Ibrahim, Shri Hafiz Mohammad	Minister of Irrigation and Power (From 2.4.1958 to 10.4.1962*) (Again from 10.4.1962 to 26.6.1963)
86.	Jaffer Sharief, Shri C.K.	Minister of State in the Ministry of Railways (From 14.1.1980 to 31.10.1984*)
		Minister of State (Independent Charge) of the Ministry of Irrigation (From 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Energy—Department of Coal (From 14.2.1988 to 2.12.1989*)
		Minister of Railways (From 21.6.1991 to 17.8.1995)
		Minister without portfolio (From 17.8.1995 to 22.11.1995)
87.	Jagmohan, Shri	Minister of Communications (From 5.12.1998 to 8.6.1999)
		Minister of Urban Development (From 8.6.1999 to 13.10.1999*) (Again from 13.10.1999 to 22.11.1999)
		Minister of Urban Development and Poverty Alleviation (From 22.11.1999 to 26.11.1999)
		Minister of Urban Development (From 26.11.1999 to 27.5.2000)
		Minister of Urban Development and Poverty Alleviation (From 27.5.2000 to 1.9.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Tourism and Minister of Statistics and Programme Implementation (From 1.9.2001 to 18.11.2001)
		Minister of Tourism and Culture (From 18.11.2001 to 22.5.2004*)
88.	Jain, Shri Ajit Prasad	Minister of State without portfolio (From 29.5.1950 to 1.6.1950)
		Minister of State for Rehabilitation (From 1.6.1950 to 13.5.1952*)
		**Minister of Rehabilitation (From 13.5.1952 to 11.8.1954) (Again from 11.8.1954 to 17.4.1957*)
		Also held charge of the Ministry of Food and Agriculture (From 25.11.1954 to 17.4.1957*)
		Minister of Food and Agriculture (From 17.4.1957 to 24.8.1959, the date of his resignation)
89.	Jairam Das Daulat Ram, Shri	Minister of Food and Agriculture (From 19.1.1948 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1950 the date of his resignation)
90.	Jaiswal, Shri Shriprakash	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Home Affairs (From 23.5.2004 to 22.5.2009*)
		Minister of State (Independent Charge) of the Ministry of Coal and Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation (From 28.5.2009 to 19.1.2011)

^{*}The date of dissolution of the Council of Ministers.

^{**}Minister of Cabinet rank but not member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Coal (From 19.1.2011 to 26.05.2014 [#])
91.	Jaitley, Shri Arun	Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 13.10.1999 to 30.9.2000)
		Held charge of the Department of Disinvestment as a Minister of State (Independent Charge) (From 10.12.1999 to 24.7.2000)
		Minister of State (Independent Charge) of the Ministry of Law, Justice and Company Affairs (From 24.7.2000 to 7.11.2000)
		Minister of Law, Justice and Company Affairs with additional charge of Shipping (From 7.11.2000 to 1.9.2001)
		Minister of Law, Justice and Company Affairs (From 1.9.2001 to 1.7.2002, the date of his resignation)
		Minister of Law and Justice and Minister of Commerce and Industry (From 29.1.2003 to 22.5.2004*)
92.	Jakhar, Shri Balram	Minister of Agriculture (From 21.6.1991 to 17.1.1996)
93.	Jalappa, Shri R.L.	Minister of State in the Ministry of Textiles (Independent charge) (From 29.6.1996 to 6.7.1996)
		Minister of Textiles (From 6.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 20.1.1998)

 $^{^{\#}\}mathrm{Date}$ of resignation of Council of Ministers of 15th Lok Sabha. *The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
94.	Jana Krishnamurthy, Shri K.	Minister of Law and Justice (From 1.7.2002 to 29.1.2003, the date of his resignation)
95.	Jaswant Singh, Shri	Minister of Finance (From 16.5.1996 to 1.6.1996*)
		Minister of External Affairs (From 5.12.1998 to 13.10.1999*)
		Held charge of the Department of Electronics (From 3.2.1999 to 13.10.1999*)
		Held charge of the Ministry of Surface Transport (From 5.8.1999 to 13.10.1999*)
		Minister of External Affairs (From 13.10.1999 to 1.7.2002)
		Also Minister of Defence (From 18.3.2001 to 15.10.2001)
		Minister of Finance and Company Affairs (From 1.7.2002 to 22.5.2004*)
96.	Jatiya, Dr. Satya Narayan	Minister of Labour (From 19.3.1998 to 13.10.1999*)
		Minister of Urban Affairs and Poverty Alleviation (From 13.10.1999 to 22.11.1999)
		Minister of Labour (From 22.11.1999 to 1.9.2001)
		Minister of Social Justice and Empowerment (From 1.9.2001 to 22.5.2004*)
97.	Jethmalani, Shri Ram	Minister of Law, Justice and Company Affairs (From 16.5.1996 to 1.6.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Urban Development (From 19.3.1998 to 8.6.1999)
		Minister of Law, Justice and Company Affairs (From 8.6.1999 to 13.10.1999*)
		Minister of Law, Justice and Company Affairs (From 13.10.1999 to 23.7.2000, the date of his resignation)
98.	Joshi, Dr. Murli Manohar	Minister of Home Affairs (From 16.5.1996 to 1.6.1996*)
		Minister of Human Resource Development (From 19.3.1998 to 13.10.1999*)
		Held charge of the Ministry of Science and Technology (From 19.3.1998 to 13.10.1999*)
		Held charge of the Department of Ocean Development (From 3.2.1999 to 13.10.1999*)
		Minister of Human Resource Development and Science and Technology (From 13.10.1999 to 22.5.2004*)
		Held charge of Ocean Development (From 22.11.1999 to 22.5.2004*)
99.	Joshi, Shri C.P.	Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Rural Development and Minister of Panchayati Raj (From 28.5.2009 to 19.1.2011)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Road Transport and
		Highways
		(From 19.1.2011 to 15.6.2013)
		Minister of Railways
		(from 22.9.2012 to 28.10.2012)
		(Again from 11.5.2013 to 15.6.2013)
100.	Joshi, Shri Manohar	Minister of Heavy Industries and Public Enterprises
		(From 13.10.1999 to 9.5.2002, the date of his resignation)
101.	Jual Oram, Shri	Minister of Tribal Affairs
		(From 13.10.1999 to 22.5.2004*)
102.	Kabir, Shri Humayun	Minister of State in the Ministry of
		Transport and Communications
		(From 17.4.1957 to 10.4.1958)
		Minister of Scientific Research and
		Cultural Affairs
		(From 10.4.1958 to 10.4.1962 ^{©©})
		(Again from 10.4.1962 to
		21.11.1963 ^{©©})
		Also held charge of the Ministry of Education
		(From 1.9.1963 to 21.11.1963)
		Minister of Petroleum and Chemicals
		(From 21.11.1963 to 27.5.1964 ^{©©})
		(Again from 27.5.1964 to 9.6.1964*)
		(Again from 9.6.1964 to 11.1.1966*)
		(Again from 11.1.1966 to 24.1.1966*)

^{*}The date of dissolution of the Council of Ministers. ${}^{\otimes \otimes}{\rm Minister}$ of Cabinet rank but not member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
103.	Kacker, Shri S.N.	Minister of Law, Justice and Company Affairs (From 3.8.1979 to 14.1.1980*)
104.	Kalvi, Shri Kalyan Singh	Minister of Energy (From 21.11.1990 to 21.6.1991*)
105.	Kamal Nath, Shri	Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 21.6.1991 to 15.9.1995)
		Minister of State (Independent Charge) of the Ministry of Textiles (From 15.9.1995 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Commerce and Industry (From 23.5.2004 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Road Transport and Highways (From 28.5.2009 to 19.1.2011)
		Minister of Urban Development (From 19.1.2011 to 26.05.2014*)
		Minister of Parliamentary Affairs (From 28.10.2012 to 26.05.2014*)
106.	Karan Singh, Dr.	Minister of Tourism and Civil Aviation (From 16.3.1967 to 18.3.1971*) (Again from 18.3.1971 to 9.11.1973)
		Minister of Health and Family Planning (From 9.11.1973 to 24.3.1977*)
		Minister of Education and Culture (From 30.7.1979 to 14.1.1980*)
107.	Karunakaran, Shri K.	Minister of Industry (From 11.6.1995 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
108.	Katju, Dr. Kailash Nath	Minister of Home Affairs and Law (From 5.11.1951 to 13.5.1952*)
		Minister of Home Affairs and States (From 13.5.1952 to 10.1.1955)
		Minister of Defence (From 10.1.1955 to 30.1.1957)
109.	Katoch, Smt. Chandresh Kumari	Minister of Culture (From 28.10.2012 to 26.05.2014*)
110.	Kaul, Smt. Sheila	Minister of State in the Ministry of Culture and Social Welfare (From 19.10.1980 to 8.8.1981)
		Also Minister of State in the Ministry of Education (From 24.11.1980 to 8.8.1981)
		Minister of State (Independent Charge) of the Ministry of Education and Culture and Social Welfare (From 8.8.1981 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of Urban Development (From 21.6.1991 to 3.5.1995)
		Minister of Urban Affairs and Employment (From 3.5.1995 to 10.9.1995)
111.	Kaushal, Shri Jagannath	Minister of Law, Justice and Company Affairs (From 15.1.1982 to 31.10.1984*)
		Minister of Law and Justice (From 4.11.1984 to 31.12.1984*)
112.	Kaushik, Shri Purushottam	Minister of Tourism and Civil Aviation (From 26.3.1977 to 15.7.1979)
		Minister of Information and Broadcasting (From 28.7.1979 to 14.1.1980*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
113.	Kesri, Shri Sitaram	Minister of State in the Department of Parliamentary Affairs (From 3.3.1980 to 15.1.1982)
		Minister of State in the Ministry of Shipping and Transport (From 15.1.1982 to 29.1.1983, the date of his resignation)
		Minister of State in the Ministry of Parliamentary Affairs (From 25.9.1985 to 22.10.1985)
		Minister of Welfare (From 21.6.1991 to 16.5.1996*)
114.	Khan, Shri Arif Mohammad	Deputy Minister in the Ministry of Information and Broadcasting (From 15.1.1982 to 29.1.1983)
		Minister of State in the Ministries of (i) Agriculture (From 29.1.1983 to 7.2.1984) (ii) Energy (From 7.2.1984 to 31.10.1984*) (Again from 12.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Industry and Company Affairs (From 31.12.1984 to 25.9.1985)
		Also Minister of State in the Ministry of Home Affairs (From 12.8.1985 to 25.9.1985)
		Minister of State in the Ministry of Energy—Department of Power (From 25.9.1985 to 26.2.1986)
		Minister of Energy with additional charge of the Ministry of Civil Aviation (From 6.12.1989 to 10.11.1990*)
115.	Khan, Shri K. Rahman	Minister without portfolio (From 22.5.2004 to 23.5.2004)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Chemicals and Fertilizers (23.5.2004 to 20.7.2004, the date of his resignation)
		Minister of Minority Affairs (from 28.10.2012 to 26.05.2014*)
116.	Khanduri, AVSM, Maj. Gen. (Retd.) Bhuwan Chandra	Minister of State in (Independent Charge) of the Ministry of Road Transport and Highways (From 7.11.2000 to 24.5.2003)
		Minister of Road Transport and Highways (From 24.5.2003 to 22.5.2004*)
		Held Charge of the Ministry of Urban Development and Poverty Alleviation (From 12.7.2003 to 8.9.2003)
117.	Khanna, Justice H.R.	Minister of Law, Justice and Company Affairs (From 30.7.1979 to 3.8.1979)
118.	Kharge, Shri Mallikarjun	Minister of Labour and Employment (from 28.5.2009 to 17.6.2013)
		Minister of Railways (from 17.6.2013 to 26.05.2014*)
		Minister of Social Justice and Empowerment (from 28.1.2014 to 26.05.2014*)
119.	Khurana, Shri Madan Lal	Minister of Parliamentary Affairs (From 19.3.1998 to 30.1.1999, the date of his resignation)
		Held charge of Ministry of Tourism (From 19.3.1998 to 30.1.1999, the date of his resignation)
120.	Khursheed, Shri Salman	Deputy Minister in the Ministry of Commerce (From 21.6.1991 to 18.1.1993)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of External Affairs (From 18.1.1993 to 16.5.1996*)
		Minister of State (Independent Charge) of the Ministry of Corporate Affairs, Minister of State (Independent Charge) of the Ministry of Minority Affairs (from 28.5.2009 to 19.1.2011)
		Minister of Water Resources (from 19.1.2011 to 12.7.2011)
		Minister of Minority Affairs (from 19.1.2011 to 28.10.2012)
		Minister of Law and Justice (from 12.7.2011 to 28.10.2012)
		Minister of External Affairs (from 28.10.2012 to 26.05.2014*)
121.	Kidwai, Shri Rafi Ahmed	Minister of Communications (From 15.8.1947 to 26.1.1950) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 2.8.1951)
		Minister of Food and Agriculture (From 13.5.1952 to 24.10.1954, the date of his death)
122.	Kidwai, Smt. Mohsina	Minister of State in the Ministry of Labour and Rehabilitation (From 11.9.1982 to 29.1.1983)
		Minister of State in the Ministry of Health and Family Welfare (From 29.1.1983 to 2.8.1984)
		Minister of State (Independent Charge) in the Ministry of Rural Development (From 2.8.1984 to 31.10.1984)
		Minister of Rural Development (From 4.11.1984 to 31.12.1984*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Health and Family Welfare (From 31.12.1984 to 24.6.1986)
		Minister of Transport (From 24.6.1986 to 22.10.1986)
		Minister of Urban Development (From 22.10.1986 to 2.12.1989*)
		Also held charge of the Ministry of Tourism (From 14.2.1988 to 25.6.1988)
123.	Kotadia, Shri Manubhai	Minister of State with Independent Charge of the Ministry of Water Resources (From 6.12.1989 to 5.11.1990)
		Minister of Water Resources (From 21.11.1990 to 26.4.1991)
		Held additional charge of the Ministry of Surface Transport (From 21.11.1990 to 26.4.1991)
124.	Krishna, Shri S.M.	Minister of State in the Ministry of Industry (From 29.1.1983 to 7.2.1984)
		Minister of State in the Ministry of Finance (From 7.2.1984 to 7.9.1984)
		Minister of State in the Ministry of Commerce (From 7.9.1984 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister without portfolio (from 22.5.2009)
		Minister of External Affairs (from 23.5.2009 to 27.10.2012)
125.	Krishnamachari, Shri T.T.	Minister of Commerce and Industry (From 13.5.1952 to 30.8.1956)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Finance (From 30.8.1956 to 17.4.1957) (Again from 17.4.1957 to 14.2.1958, the date of his resignation)
		Minister without portfolio (From 8.6.1962 to 14.11.1962)
		Minister of Economic and Defence Coordination (From 14.11.1962 to 31.8.1963)
		Minister of Finance (From 31.8.1963 to 31.12.1965)
126.	Kumar, Shri Ashwani	Minister of State in the Department of Industrial Policy and Promotion, Ministry of Commerce and Industries (From 29.1.2006 to 22.5.2009*)
		Minister of State in the Ministry of Planning (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Parliamentary Affairs (From 19.1.2011 to 12.7.2011)
		Minister of State in the Ministry of Science and Technology (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Earth Sciences (From 19.1.2011 to 28.10.2012)
		Minister of Law and Justice (From 28.10.2012 to 11.5.2013)
127.	Kumaramangalam, Shri P.R.	Minister of State in the Ministries of Parliamentary Affairs and Law, Justice and Company Affairs (From 21.6.1991 to 2.7.1992)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Science and Technology (Department of Electronics, Department of Ocean Development) and Parliamentary Affairs (From 2.7.1992 to 2.12.1993)
		Minister of Power (From 19.3.1998 to 13.10.1999*)
		Held charge of the Ministry of Parliamentary Affairs (From 30.1.1999 to 13.10.1999*)
		Held charge of the Ministry of Non- Conventional Energy Sources (From 3.2.1999 to 13.10.1999*)
		Held charge of the Ministry of Law, Justice and Company Affairs (From 9.4.1999 to 8.6.1999)
		Minister of Power (From 13.10.1999 to 23.8.2000, the date of his death)
		Held charge of the Ministry of Mines and Minerals (From 6.3.2000 to 27.5.2000)
128.	Kumaramangalam, Shri S. Mohan	Minister of Steel and Heavy Engineering (From 18.3.1971 to 2.5.1971)
		Minister of Steel and Mines (From 2.5.1971 to 31.5.1973, the date of his death)
129.	Kumari, Selja	Deputy Minister in the Ministry of Human Resource Development— Department of Education and Culture (From 2.7.1992 to 15.9.1995)
		Minister of State in the Ministry of Human Resource Development— Department of Education and Culture (From 15.9.1995 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Urban Employment and Poverty Alleviation (From 23.5.2004 to 1.6.2006)
		**Minister of State (Independent Charge) of the Ministry of Housing and Urban Poverty Alleviation (From 1.6.2006 to 22.5.2009*)
		Minister of Housing and Urban Poverty Alleviation (From 28.5.2009 to 28.10.2012)
		Minister of Tourism (from 28.5.2009 to 19.1.2011)
		Minister of Culture (from 19.1.2011 to 28.10.2012)
		Minister of Social Justice and Empowerment (from 28.10.2012 to 28.1.2014)
130.	Kyndiah, Shri P.R.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Development of North- Eastern Region (From 23.5.2004 to 24.10.2006)
		Minister of Tribal Affairs (From 23.5.2004 to 22.5.2009*)
131.	Mahajan, Shri Pramod	Minister of Defence and Parliamentary Affairs (From 16.5.1996 to 1.6.1996)
		Minister of Information and Broadcasting (From 5.12.1998 to 13.10.1999*)

^{*}The date of dissolution of the Council of Ministers.
**Date of notification of change of nomenclature of Ministry (1.6.2006).

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of Food Processing Industries (From 3.2.1999 to 13.10.1999*)
		Minister of Parliamentary Affairs and Water Resources (From 13.10.1999 to 22.11.1999)
		Minister of Parliamentary Affairs and Information Technology (From 22.11.1999 to 29.1.2003, the date of his resignation)
		Held charge of the Ministry of Communications** (From 1.9.2001 to 29.1.2003)
132.	Mahavir Prasad, Shri	Deputy Minister in the Ministry of Railways (From 14.2.1988 to 4.7.1989)
		Minister of State in the Ministry of Steel and Mines—Department of Mines (From 4.7.1989 to 2.12.1989*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Small Scale Industries and Minister of Agro and Rural Industries*** (From 23.5.2004 to 22.5.2009*)
133.	Maken, Shri Ajay	Minister of State in the Ministry of Urban Development (From 29.1.2006 to 22.5.2009*)
		Minister of State in the Ministry of Home Affairs (From 28.5.2009 to 19.1.2011)

 $[\]ensuremath{^{*}}$ The date of dissolution of the Council of Ministers.

^{**} w.e.f. 22.12.2001 both the Ministry of Communications and that of Information Technology had been merged together to form a single Ministry of Communications and Information Technology and Shri Mahajan continued as Minister of the same.

*** w.e.f. 9.5.2007, the desgination changed as Minister of Micro, Small and Medium

^{***} w.e.f. 9.5.2007, the desgination changed as Minister of Micro, Small and Medium Enterprises.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of Ministry of Youth Affairs and Sports (From 19.1.2011 to 28.10.2012)
		Minister of Housing and Poverty Alleviation (From 28.10.2012 to 15.6.2013)
134.	Malaviya, Shri Satya Prakash	Minister of Petroleum and Chemicals and Parliamentary Affairs (From 21.11.1990 to 21.6.1991)
135.	Malviya, Shri K.D.	Deputy Minister in the Ministry of Natural Resources and Scientific Research (From 12.8.1952 to 7.12.1954)
		^{©®} Minister in the Ministry of Natural Resources and Scientific Research (From 7.12.1954 to 8.1.1955)
		°°Minister of Natural Resources (From 8.1.1955 to 17.4.1957*)
		Minister of State in the Ministry of Steel, Mines and Fuel (From 17.4.1957 to 25.4.1957*)
		Minister of Mines and Oil (From 25.4.1957 to 10.4.1962*)
		Minister of Mines and Fuel (From 10.4.1962 to 26.6.1963)
		Minister of Steel and Mines (From 11.1.1974 to 10.10.1974)
		Minister of Petroleum and Chemicals (From 10.10.1974 to 21.12.1975)
		Minister of Petroleum (From 21.12.1975 to 24.3.1977*)
		Also held charge of the Ministry of Chemicals and Fertilizers (From 22.9.1976 to 24.3.1977*)

 $^{^{\}odot\odot}$ Minister of Cabinet rank but not member of the Cabinet. * The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
136.	Maran, Shri Dayanidhi	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Communications and Information Technology (From 23.5.2004 to 15.5.2007, the date of his resignation)
		Minister of Textiles (from 28.5.2009 to 12.7.2011)
137.	Maran, Shri Murasoli	Minister of Urban Development (From 6.12.1989 to 10.11.1990*)
		Minister of Industry (From 1.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
		Minister of Commerce and Industry (From 13.10.1999 to 9.11.2002)
		Minister without portfolio (From 9.11.2002 to 23.11.2003, the date of his death)
138.	Mathai, Dr. John	Minister of Transport (From 15.8.1947 to 22.9.1948)
		Minister of Finance (From 22.9.1948 to 26.1.1950*) (Again from 26.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 1.6.1950)
139.	Mehta, Shri Ashoka	Minister of Planning (From 24.1.1966 to 25.3.1966)
		Minister of Planning and Social Welfare (From 25.3.1966 to 13.3.1967*)
		Minister of Planning, Petroleum and Chemicals and Social Welfare (From 13.3.1967 to 5.9.1967)
		Minister of Petroleum and Chemicals and Social Welfare (From 5.9.1967 to 22.8.1968)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
140.	Mehtab, Shri Hare Krishna	Minister of Industry and Supply (From 13.5.1950 to 26.12.1950)
		Minister of Commerce and Industry (From 26.12.1950 to 13.5.1952)
141.	Meira Kumar, Smt.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Social Justice and Empowerment (From 23.5.2004 to 22.5.2009)
		Minister without portfolio (from 22.5.2009 to 28.5.2009)
		Minister of Water Resources (from 28.5.2009 to 31.5.2009, the date of her resignation)
142.	Menon, Shri P. Govinda	Minister of State in the Ministry of Food and Agriculture, Community Development and Cooperation (From 24.1.1966 to 13.3.1967*)
		Minister of Law (From 13.3.1967 to 22.8.1968)
		Minister of Law and Social Welfare (From 22.8.1968 to 23.5.1970)
		Also Minister of Railways (From 4.11.1969 to 18.2.1970)
143.	Menon, Shri V.K. Krishna	Minister without portfolio (From 14.2.1956 to 17.5.1956)
		Minister of Health (From 17.5.1956 to 17.4.1957*)
		Minister of Defence (From 17.4.1957 to 10.4.1962*) (Again from 10.4.1962 to 1.11.1962)
		Minister of Defence Production (From 1.11.1962 to 9.11.1962)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
144.	Mir Quasim, Shri Syed	Minister without portfolio (From 7.6.1976 to 9.8.1976)
		Minister of Civil Supplies and Cooperation (From 9.8.1976 to 24.3.1977*)
145.	Mirdha, Shri Nathuram	Minister of State in the Ministry of Agriculture and Irrigation (From 4.8.1979 to 25.10.1979)
		Minister of State in the Ministry of Finance (From 25.10.1979 to 14.1.1980*)
		Minister of Food and Civil Supplies (From 8.12.1989 to 10.11.1990)
146.	Mirdha, Shri Ram Niwas	Minister of State in the Ministry of Home Affairs (From 26.6.1970 to 18.3.1971*)
		Also Minister of State in the Department of Personnel in the Cabinet Secretariat (From 23.8.1970 to 18.3.1971*) (Again from 18.3.1971 to 10.10.1974)
		Minister of State, Defence Production in the Ministry of Defence (From 10.10.1974 to 21.12.1975)
		Minister of State in the Ministry of Supply and Rehabilitation (From 21.12.1975 to 24.3.1977*)
		Minister of State (Independent Charge) of the Ministry of Irrigation (From 29.1.1983 to 2.8.1984)
		Minister of State in the Ministry of External Affairs (From 2.8.1984 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Communications (From 31.12.1984 to 22.10.1986)
		Also Minister of State in the Ministry of Home Affairs (From 14.3.1986 to 14.6.1986)
		Minister of State (Independent Charge) of the Ministry of Textiles (From 22.10.1986 to 15.2.1988)
		Also Minister of State (Independent Charge) of the Ministry of Water Resources (From 10.11.1987 to 14.2.1988)
		Minister of Textiles (From 15.2.1988 to 2.12.1989*)
		Also Minister of Health and Family Welfare (From 24.1.1989 to 4.7.1989)
147.	Mishra, Shri Chaturanan	Minister of Agriculture (Excluding Department of Animal Husbandry and Dairying) (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of Agriculture (including Animal Husbandry and Dairying) (From 9.6.1997 to 19.3.1998*)
		Held charge of the Ministry of Food, Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 24.4.1997 to 9.6.1997)
148.	Mishra, Shri Jagannath	Minister of Rural Areas and Employment (From 11.6.1995 to 16.5.1996*)
		Also Minister of Agriculture (From 7.2.1996 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
149.	Mishra, Shri Janeshwar	Minister of State in Ministry of Petroleum and Chemicals and Fertilizers (From 14.8.1977 to 11.7.1978)
		Minister of State in the Ministry of Energy (From 26.1.1979 to 15.7.1979)
		Minister of State in the Ministry of Shipping and Transport (From 30.7.1979 to 14.1.1980*)
		Minister of State with Independent Charge of the Ministry of Communications (From 23.4.1990 to 5.11.1990)
		Minister of Railways (From 21.11.1990 to 21.6.1991*)
		Minister of Water Resources (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of Petroleum and Natural Gas (From 9.6.1997 to 19.3.1998*)
150.	Mishra, Shri L.N.	Deputy Minister in the Ministry of Planning and Labour and Employment (From 22.8.1960 to 10.4.1962*)
		Deputy Minister in the Ministry of Home Affairs (From 26.2.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Finance (From 24.1.1966 to 13.3.1967*)
		Minister of State in the Ministry of Labour, Employment and Rehabilitation (From 13.3.1967 to 14.11.1967)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Defence Production) in the Ministry of Defence (From 14.11.1967 to 27.6.1970)
		Minister of Foreign Trade (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 5.2.1973)
		Minister of Railways (From 5.2.1973 to 3.1.1975, the date of his death)
151.	Mishra, Shri Shyam Nandan	Deputy Minister in the Ministry of Planning (From 10.9.1954 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962)
		Minister of External Affairs (From 28.7.1979 to 14.1.1980*)
		Also held additional charge of the Ministry of Communications (From 7.12.1979 to 14.1.1980*)
152.	Mohite, Shri Subodh	Minister of Heavy Industries and Public Enterprises (From 24.5.2003 to 22.5.2004*)
153.	Moily, Shri M. Veerappa	Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Law and Justice (From 28.5.2009 to 12.7.2011)
		Minister of Corporate Affairs (From 12.7.2011 to 28.10.2012)
		Minister of Power (From 31.7.2012 to 28.10.2012)
		Minister of Petroleum and Natural Gas (From 28.10.2012 to 26.05.2014*)
		Minister of Environment and Forests (From 21.12.2013 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
154.	Mookerjee, Dr. Shyama Prasad	Minister of Industry and Supply (From 15.8.1947 to 26.1.1950*) (Again from 26.1.1950 to 19.4.1950)
155.	Mukherjee, Shri Pranab	Deputy Minister in the Ministry of Industrial Development (From 5.2.1973 to 11.1.1974)
		Deputy Minister in the Ministry of Shipping and Transport (From 11.1.1974 to 10.10.1974)
		Minister of State in the Ministry of Finance (From 10.10.1974 to 21.12.1975)
		Minister of State of the Department of Revenue and Banking (From 21.12.1975 to 24.3.1977*)
		Minister of Commerce (From 14.1.1980 to 15.1.1982)
		Also held charge of the Ministry of Steel and Mines and Civil Supplies (From 16.1.1980 to 15.1.1982)
		Minister of Finance (From 15.1.1982 to 31.10.1984*)
		Also held charge of the Ministry of Commerce (From 7.9.1984 to 31.10.1984*)
		Minister of Finance, Commerce and Supply (From 31.10.1984 to 31.12.1984*)
		Minister of Commerce (From 17.1.1993 to 9.7.1993) (Again from 31.8.1993 to 10.2.1995)
		Minister of External Affairs (From 10.2.1995 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Defence (From 23.5.2004 to 24.10.2006)
		Minister of External Affairs (From 24.10.2006 to 22.5.2009*)
		Held additional charge as Minister of Finance (From 24.1.2009 to 22.5.2009*)
		Minister without portfolio (from 22.5.2009 to 23.5.2009)
		Minister of Finance (from 23.5.2009 to 26.6.2012)
156.	Munda, Shri Kariya	Minister of State in the Ministry of Steel and Mines (From 14.8.1977 to 28.7.1979*)
		Minister of Welfare (From 16.5.1996 to 1.6.1996*)
		Minister of Agro and Rural Industries (From 1.9.2001 to 29.1.2003)
		Minister of Coal (From 29.1.2003 to 9.1.2004)
		Minister of Non-conventional Energy Sources (From 9.1.2004 to 22.5.2004*)
157.	Munshi, Shri K.M.	Minister of Food and Agriculture (From 13.5.1950 to 13.5.1952*)
158.	Muthiah, Shri R.	Minister of Surface Transport (From 19.3.1998 to 8.4.1998, the date of his resignation)
159.	Muthu, Sathiavani (Dr.) Smt.	Minister of Social Welfare (From 19.8.1979 to 23.12.1979)
160.	Naidu, Shri M. Venkaiah	Minister of Rural Development (From 30.9.2000 to 1.7.2002, the date of his Resignation)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
161.	Naidu, Shri Yerran K.	Minister of Rural Areas and Employment (From 1.6.1996 to 21.4.1997) (Again from 21.4.1997 to 19.3.1998*)
162.	Naik, Shri Ram	Minister of State in the Ministry of Railways (From 19.3.1998 to 6.8.1999)
		Minister of State in the Ministry of Parliamentary Affairs (From 20.3.1998 to 5.5.1999)
		Minister of State in the Ministry of Planning and Programme Implementation (From 20.4.1998 to 13.10.1999*)
		Minister of State (Independent Charge) of the Ministry of Railways (From 6.8.1999 to 13.10.1999*)
		Minister of State in the Ministry of Home Affairs (From 5.5.1999 to 13.10.1999*)
		Minister of Petroleum and Natural Gas (From 13.10.1999 to 22.5.2004*)
163.	Natwar Singh, Shri K.	Minister of State in the Ministry of Steel, Mines and Coal — Department of Steel (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Agriculture — Department of Fertilizers (From 25.9.1985 to 22.10.1986)
		Minister of State in the Ministry of External Affairs (From 22.10.1986 to 2.12.1989*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of External Affairs (23.5.2004 to 7.11.2005)
		Minister without portfolio (from 7.11.2005 to 7.12.2005, the date of his resignation)
164.	Nehru, Shri Arun K.	Minister of State in the Ministry of Energy—Department of Power (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Home Affairs—Department of Internal Security (From 25.9.1985 to 22.10.1986)
		Minister of Commerce with additional charge of the Ministry of Tourism (From 6.12.1989 to 10.11.1990*)
165.	Neogy, Dr. K.C.	Minister of Relief and Rehabilitation (From 6.9.1947 to 6.4.1948)
		Minister of Commerce (From 6.4.1948 to 26.1.1950*) (Again from 26.1.1950 to 19.4.1950, the date of his resignation)
166.	Nitish Kumar, Shri	Minister of State in the Ministry of Agriculture—Department of Agriculture and Cooperation (From 23.4.1990 to 10.11.1990)
		Minister of Railways (From 19.3.1998 to 5.8.1999, the date of his resignation)
		Held Charge of Ministry of Surface Transport (From 9.4.1999 to 5.8.1999)
		Minister of Surface Transport (From 13.10.1999 to 22.11.1999)
		Minister of Agriculture (From 22.11.1999 to 3.3.2000) (Again from 27.5.2000 to 22.7.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Railways (From 20.3.2001 to 22.5.2004*)
167.	Ola, Shri Sis Ram	Minister of State (Independent Charge) of the Ministry of Chemicals and Fertilisers (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State (Independent Charge of the Ministry of Water Resources (From 9.6.1997 to 19.3.1998*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Labour and Employment (From 23.5.2004 to 27.11.2004)
		Minister of Mines (From 27.11.2004 to 22.5.2009*)
		Minister of Labour and Employment (From 17.6.2013 to 15.12.2013**)
168.	Pai, Shri T.A.	Minister of Railways (From 23.7.1972 to 5.2.1973)
		Minister of Heavy Industries (From 5.2.1973 to 10.10.1974)
		Also held additional charge of Ministry of Steel and Mines (From 23.7.1973 to 11.1.1974)
		Minister of Industry and Civil Supplies (From 10.10.1974 to 9.8.1976)
		Also held charge of the Department of Science and Technology (From 10.10.1974 to 2.1.1975)
		Minister of Industry (From 9.8.1976 to 24.3.1977*)
		Minister of Railways (From 30.7.1979 to 14.1.1980*)

^{*}The date of dissolution of the Council of Ministers. **Date of his death.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also held charge as Minister of Industry (From 27.11.1979 to 14.1.1980*)
169.	Pajnor, Shri A. Bala	Minister of Petroleum, Chemicals and Fertilizers (From 19.8.1979 to 23.12.1979)
170.	Pallam Raju, Shri M.M.	Minister of State in the Ministry of Defence (From 29.1.2006 to 22.5.2009*)
		Minister of State in the Ministry of Defence (From 28.5.2009 to 28.10.2012)
		Minister of Human Resource Development (From 28.10.2012 to 26.05.2014*)
171.	Pandey, Shri Kedar	Minister of Irrigation (From 8.6.1980 to 12.11.1980) (Again from 15.1.1982 to 29.1.1983)
		Minister of Railways (From 12.11.1980 to 15.1.1982)
172.	Pandey, Shri Rajmangal	Minister of Human Resource Development (From 21.11.1990 to 21.6.1991*)
173.	Pant, Shri Govind Ballabh	Minister without portfolio (From 3.1.1955 to 10.1.1955)
		Minister of Home Affairs (From 10.1.1955 to 17.4.1957*)
		Also held charge of the Ministry of Heavy Industries (From 30.8.1956 to 14.11.1956)
		Minister of Home Affairs (From 17.4.1957 to 7.3.1961)
174.	Pant, Shri Krishna Chandra	Minister of State in the Ministry of Finance (From 13.3.1967 to 14.2.1969)

^{*}The date of dissolution of the Council of Ministers.

Portfolios alongwith the period thereof

Minister of State in the Ministry of Steel and Heavy Engineering (From 14.2.1969 to 27.6.1970)

Minister of State in the Ministry of Home Affairs and Department of Electronics and Scientific and Industrial Research (From 27.6.1970 to 18.3.1971*)

Minister of State in the Ministry of Home Affairs and Departments of Electronics, Atomic Energy and Science and Technology (From 18.3.1971 to 2.5.1971)

Minister of State in the Ministry of Home Affairs and Departments of Electronics and Atomic Energy (From 2.5.1971 to 9.11.1973)

Also assisted the Prime Minister in the Department of Space (From 26.8.1972 to 9.11.1973)

Minister of Irrigation and Power (From 9.11.1973 to 10.10.1974)

Minister of Energy and assisted the Prime Minister in the Departments of Atomic Energy, Space and Electronics (From 10.10.1974 to 24.3.1977)

Minister of Energy (From 30.7.1979 to 14.1.1980*)

Minister of Education (From 31.12.1984 to 25.9.1985)

Minister of Steel and Mines (From 25.9.1985 to 12.4.1987)

Minister of Defence (From 12.4.1987 to 2.12.1989*)

175. Paswan, Shri Ram Vilas

Minister of Welfare and Labour (From 6.12.1989 to 10.11.1990*)

^{*}The date of dissolution of the Council of Ministers.

	Name	Portfolios alongwith the period thereof
		Minister of Railways (From 1.6.1996 to 21.4.1997*)
		Also Minister of Parliamentary Affairs (From 1.6.1996 to 29.6.1996)
		Minister of Railways (From 21.4.1997 to 19.3.1998*)
		Minister of Communications (From 13.10.1999 to 1.9.2001)
		Minister of Coal and Mines (From 1.9.2001 to 29.4.2002, the date of his resignation)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Chemicals and Fertilizers (From 23.5.2004 to 22.5.2009*)
		Also Minister of Steel (From 23.5.2004 to 22.5.2009*)
176.	Patel, Shri Dinsha J.	Minister of State without portfolio (From 29.1.2006 to 31.1.2006)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 31.1.2006 to 22.5.2009*)
		Minister of State (Independent Charge) of the Ministry of Micro, Small and Medium Enterprises (From 28.5.2009 to 19.1.2011)
		Minister of State (Independent Charge) of the Ministry of Mines (From 19.1.2011 to 28.10.2012)
		Minister of Mines (From 28.10.2012 to 26.05.2014*)
177.	Patel, Shri H.M.	Minister of Finance and Revenue and Banking (From 26.3.1977 to 24.1.1979)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Home Affairs (From 29.1.1979 to 28.7.1979*)
178.	Patel, Shri Praful	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation (From 23.5.2004 to 22.5.2009*)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation (From 28.5.2009 to 19.1.2011)
		Minister of Heavy Industries and Public Enterprises (From 19.1.2011 to 26.05.2014*)
179.	Pathak, Shri G.S.	Minister of Law (From 24.1.1966 to 13.3.1967*)
180.	Patil, Shri Balasaheb Vikhe	Minister of State in the Ministry of Finance (From 13.10.1999 to 1.7.2002)
		Minister of Heavy Industries and Public Enterprises (From 1.7.2002 to 24.5.2003, the date of his resignation)
181.	Patil, Shri S.K.	Minister of Irrigation and Power (From 17.4.1957 to 29.3.1958)
		Minister of Transport and Communications (From 29.3.1958 to 24.8.1959)
		Also held additional charge of the Ministry of Irrigation and Power (From 29.3.1958 to 2.4.1958)
		Minister of Food and Agriculture (From 24.8.1959 to 10.4.1962*) (Again from 10.4.1962 to 1.9.1963)
		Minister of Railways (From 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
182.	Patil, Shri Shivraj V.	Minister of State in the Ministry of Defence (From 19.10.1980 to 15.1.1982)
		Minister of State in the Ministry of Commerce (Independent Charge) (From 15.1.1982 to 29.1.1983)
		Minister of State in the Departments of Science and Technology, Space, Atomic Energy, Electronics and Ocean Development (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 22.10.1986)
		Also Minister of State in the Ministry of Personnel and Training, Public Grievances and Pensions and Administrative Reforms (From 25.9.1985 to 4.10.1985)
		Minister of State in the Ministry of Defence—Department of Defence Production (From 22.10.1986 to 25.6.1988)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation and Tourism (From 25.6.1988 to 2.12.1989*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Home Affairs (From 23.5.2004 to 30.11.2008, the date of his resignation)
183.	Patil, Shri Veerendra	Minister of Petroleum and Chemicals (From 7.3.1980 to 19.10.1980)
		Minister of Shipping and Transport (From 19.10.1980 to 2.9.1982)
		Minister of Labour and Rehabilitation (From 2.9.1982 to 31.10.1984*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also held charge of the Ministry of Shipping and Transport (From 7.9.1984 to 31.10.1984*)
		Minister of Labour and Rehabilitation and Shipping and Transport (From 4.11.1984 to 31.12.1984*)
		Minister of Chemicals and Fertilizers (From 31.12.1984 to 25.9.1985)
		Also held charge of the Ministry of Industry and Company Affairs (From 14.1.1985 to 25.9.1985)
184.	Patnaik, Shri Biju	Minister of Steel and Mines (From 26.3.1977 to 15.7.1979)
		Minister of Steel, Mines and Coal (From 30.7.1979 to 14.1.1980*)
185.	Patnaik, Shri Janaki Ballabh	Deputy Minister in the Ministry of Defence (From 5.2.1973 to 23.12.1976)
		Minister of State in the Ministry of Defence (From 23.12.1976 to 24.3.1977*)
		Minister of Tourism and Civil Aviation (From 14.1.1980 to 7.6.1980)
		Also held charge of Ministry of Labour (From 16.1.1980 to 7.6.1980)
186.	Patnaik, Shri Naveen	Minister of Steel and Mines (From 19.3.1998 to 13.10.1999*)
		Minister of Mines and Minerals (From 13.10.1999 to 4.3.2000)
187.	Patwa, Shri Sunder Lal	Minister of Rural Development (From 13.10.1999 to 30.9.2000)
		Held charge of the Ministry of Agriculture (From 6.3.2000 to 27.5.2000)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Chemicals and Fertilisers (From 30.9.2000 to 7.11.2000)
		Minister of Mines (From 7.11.2000 to 1.9.2001)
188.	Pawar, Shri Sharad	Minister of Defence (From 26.6.1991 to 5.3.1993)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Agriculture and Minister of Consumer Affairs, Food and Public Distribution (From 23.5.2004 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 23.5.2009)
		Minister of Agriculture (From 23.5.2009 to 26.05.2014*)
		Minister of Consumer Affairs, Food and Public Distribution (From 23.5.2009 to 19.1.2011)
		Minister of Food Processing Industries (From 19.1.2011 to 26.05.2014*)
189.	Poonacha, Shri C.M.	Minister of Revenue and Expenditure in the Ministry of Finance (From 2.1.1966 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of Transport and Civil Aviation (From 24.1.1966 to 13.3.1967*)
		Minister of Railways (From 13.3.1967 to 14.2.1969)
		Minister of Steel and Heavy Engineering (From 14.2.1969 to 15.11.1969, the date of his resignation)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
190.	Prabhu, Shri Suresh Prabhakar	Minister of Industry (From 16.5.1996 to 1.6.1996*)
		Minister of Environment and Forests (From 19.3.1998 to 13.10.1999*)
		Minister of Chemicals and Fertilizers (From 13.10.1999 to 30.9.2000)
		Minister of Power (From 30.9.2000 to 24.8.2002, the date of his resignation)
		Held additional charge of the Ministry of Heavy Industries and Public Enterprises (From 9.5.2002 to 1.7.2002)
191.	Qureshi, Shri Mohammad Shafi	Deputy Minister in the Ministry of Commerce (From 29.1.1966 to 13.3.1967*) (Again from 18.3.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Steel and Heavy Engineering (From 14.2.1969 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Deputy Minister in the Ministry of Railways (From 2.5.1971 to 10.10.1974)
		Minister of State in the Ministry of Railways (From 10.10.1974 to 24.3.1977)
		Minister of Tourism and Civil Aviation (From 30.7.1979 to 14.1.1980*)
192.	Raguramaiah, Shri Kotha	Deputy Minister in the Ministry of Defence (From 21.5.1957 to 10.4.1962*)
		Minister of State in the Ministry of Defence (From 16.4.1962 to 14.11.1962)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Defence Production in the Ministry of Defence (From 14.11.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of State in the Ministry of Labour and Employment (From 9.6.1964 to 13.6.1964)
		Minister of Supply in the Ministry of Industry (From 13.6.1964 to 20.6.1964)
		Minister of Supply in the Ministry of Industry and Supply (From 20.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Also assumed the Office of Minister of Supply and Technical Development in the Ministry of Industry and Supply (From 30.10.1964 to 11.1.1966*)
		Minister of Supply and Technical Development (From 24.1.1966 to 25.3.1966)
		Also assumed charge of the Department of Social Welfare (From 24.1.1966 to 25.3.1966)
		Minister of Supply, Technical Development and Materials Planning (From 25.3.1966 to 13.3.1967*)
		Minister of State in the Ministry of Law (From 13.3.1967 to 18.3.1967)
		Minister of State in the Ministry of Petroleum and Chemicals and Planning and Social Welfare (From 18.3.1967 to 5.9.1967)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Petroleum and Chemicals and Social Welfare (From 5.9.1967 to 14.2.1969)
		Minister of Parliamentary Affairs and Shipping and Transport (From 14.2.1969 to 18.3.1971*) (Again from 18.3.1971 to 5.2.1973)
		Minister of Parliamentary Affairs (From 5.2.1973 to 10.10.1974)
		Minister of Works and Housing and Parliamentary Affairs (From 10.10.1974 to 23.12.1976)
		Also held charge of Ministry of Supply and Rehabilitation (From 1.2.1975 to 21.12.1975)
		Minister of Tourism and Civil Aviation and Parliamentary Affairs (From 23.12.1976 to 24.3.1977*)
193.	Raj Bahadur, Shri	Deputy Minister in the Ministry of Communications (From 29.5.1951 to 13.5.1952*) (Again from 4.6.1952 to 14.2.1956)
		©©Minister in the Ministry of Communications (From 14.2.1956 to 7.12.1956)
		Held charge of the Ministry of Communications (From 7.12.1956 to 20.1.1957)
		©©Minister of Communications (From 20.1.1957 to 17.4.1957*)
		Minister of State in the Ministry of Transport and Communications (From 17.4.1957 to 10.4.1962*)

^{*}The date of dissolution of the Council of Ministers. ${}^{\tiny{\text{\tiny CC}}}\!\text{Minister}$ of Cabinet Rank but not Member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Shipping in the Ministry of Transport and Communications (From 10.4.1962 to 1.9.1963)
		Minister of Shipping in the Ministry of Transport (From 1.9.1963 to 10.4.1964)
		Minister of Transport (From 10.4.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Also held charge of Ministry of Tourism and Civil Aviation (From 31.7.1965 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Information and Broadcasting (From 24.1.1966 to 13.3.1967*)
		Minister of Parliamentary Affairs and Shipping and Transport (From 18.3.1971 to 5.2.1973)
		Minister of Shipping and Transport (From 5.2.1973 to 8.11.1973)
		Minister of Communications (From 8.11.1973 to 11.1.1974)
		Minister of Tourism and Civil Aviation (From 9.11.1973 to 22.12.1976, the date of his resignation)
194.	Raj Narain, Shri	Minister of Health and Family Welfare (From 28.3.1977 to 1.7.1978)
195.	Raja, Shri A.	Minister of State in the Ministry of Rural Development (From 13.10.1999 to 30.9.2000)
		Minister of State in the Ministry of Health and Family Welfare (From 30.9.2000 to 21.12.2003, the date of his resignation)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Environment and Forests (From 23.5.2004 to 15.5.2007)
		Minister of Communications and Information Technology (From 15.5.2007 to 22.5.2009*)
		Minister of Communications and Information Technology (From 28.5.2009 to 15.11.2010)
196.	Rajagopalachari, Shri C.	Minister without portfolio (From 15.7.1950 to 26.12.1950)
		Minister of Home Affairs (From 26.12.1950 to 5.11.1951, the date of his resignation)
197.	Rajendra Prasad, Dr.	Minister of Food and Agriculture (From 15.8.1947 to 14.1.1948)
198.	Rajnath Singh, Shri	Minister of Surface Transport (From 22.11.1999 to 25.11.2000)
		Minister of Agriculture (From 24.5.2003 to 17.1.2004)
		Minister of Agriculture with additional charge of Ministry of Food Processing Industries (From 17.1.2004 to 22.5.2004*)
199.	Ram Kinkar, Shri	Minister of State in the Ministry of Works and Housing and Supply and Rehabilitation (From 14.8.1977 to 11.7.1978)
		Minister of State in the Ministry of Works and Housing and Supply and Rehabilitation (From 26.1.1979 to 15.7.1979)
		Minister of Works, Housing and Supply and Rehabilitation (From 28.7.1979 to 14.1.1980*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
200.	Ramachandran, Shri P.	Minister of Energy (From 26.3.1977 to 28.7.1979*)
201.	Ramamurthy, Shri K.	Minister of State (Independent Charge) of the Ministry of Labour (From 21.6.1991 to 30.7.1991)
		Minister of Petroleum and Natural Gas (From 19.3.1998 to 13.10.1999*)
202.	Ramdoss, Dr. (Shri) Anbumani	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Health and Family Welfare (From 23.5.2004 to 29.3.2009, the date of his resignation)
203.	Ramesh, Shri Jairam	Minister without portfolio (From 29.1.2006 to 31.1.2006)
		Minister of State in the Department of Commerce in the Ministry of Commerce and Industry (From 31.1.2006 to 25.2.2009)
		Minister of State in the Ministry of Power (From 6.4.2008 to 25.2.2009, the date of his resignation)
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 28.5.2009 to 12.7.2011)
		Minister of Rural Development (From 12.7.2011 to 26.05.2014*)
		Minister of Drinking Water and Sanitation (From 13.7.2011 to 28.10.2012)
204.	Ramoowalia, Shri Balwant Singh	Minister of Welfare (From 1.6.1996 to 21.4.1997*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of Labour (From 1.6.1996 to 29.6.1996)
		Minister of Welfare (From 21.4.1997 to 19.3.1998*)
		Also Minister of Food and Civil Supplies (From 10.1.1998 to 19.3.1998*)
205.	Rana, Shri Kashi Ram	Minister of Textiles (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 24.5.2003)
		Minister of Rural Development (From 24.5.2003 to 22.5.2004*)
206.	Rao, Dr. Kavuru Samba Siva	Minister of Textiles (From 17.6.2013 to 3.4.2014)
207.	Rao, Dr. V.K.R.V.	Minister of Transport and Shipping (From 13.3.1967 to 14.2.1969)
		Minister of Education and Youth Services (From 14.2.1969 to 18.3.1971*)
208.	Rao, Shri J. Vengala	Minister of Industry (From 22.10.1986 to 2.12.1989*)
209.	Rao, Shri K. Chandrasekhar	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Shipping (From 23.5.2004 to 25.5.2004)
		Minister without portfolio (From 25.5.2004 to 27.11.2004)
		Minister of Labour and Employment (From 27.11.2004 to 24.8.2006, the date of his resignation)
210.	Rawat, Shri Harish	Minister of State in the Ministry of Labour and Employment (From 28.5.2009 to 19.1.2011)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Agriculture, Minister of State in the Ministry of Food Processing Industry (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Parliamentary Affairs (From 12.7.2011 to 28.10.2012)
		Minister of Water Resources (From 28.10.2012 to 1.2.2014)
211.	Ray, Shri Rabi	Minister of Education and Social Welfare (From 25.1.1979 to 15.7.1979)
		Minister of Health and Family Welfare (From 28.7.1979 to 14.1.1980*)
212.	Ray, Shri S.S.	Minister of Education and Social Welfare (From 18.3.1971 to 20.3.1972)
		Held charge of the Ministry of Health and Family Planning (From 19.5.1971 to 20.3.1972)
		Held charge of the Department of Culture (From 18.3.1971 to 20.3.1972)
213.	Reddi, Shri B. Gopala	Minister of State in the Ministry of Economic Affairs (From 2.4.1958 to 10.5.1958)
		Minister of State in the Ministry of Revenue and Civil Expenditure (From 10.5.1958 to 7.4.1961)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in charge of the Ministry of Works, Housing and Supply (From 5.4.1961 to 10.4.1962*)
		Minister of Information and Broadcasting (From 10.4.1962 to 31.8.1963)
214.	Reddy, Dr. M. Chenna	Minister of Steel, Mines and Metals (From 16.3.1967 to 27.4.1968)
215.	Reddy, Shri S. Jaipal	Minister of Information and Broadcasting (From 1.5.1997 to 19.3.1998*)
		Also Minister of Food Processing Industries (From 25.12.1997 to 19.3.1998*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Information and Broadcasting (From 23.5.2004 to 18.11.2005)
		Minister of Culture (From 23.5.2004 to 29.1.2006)
		Minister of Urban Development (From 18.11.2005 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Urban Development (From 28.5.2009 to 19.1.2011)
		Minister of Petroleum and Natural Gas (From 19.1.2011 to 28.10.2012)
		Minister of Science and Technology (From 28.10.2012 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Earth Sciences (from 28.10.2012 to 26.05.2014*)
216.	Reddy, Shri K. Brahamananda	Minister of Communications (From 11.1.1974 to 10.10.1974)
		Minister of Home Affairs (From 10.10.1974 to 24.3.1977*)
		Also held temporary charge of the Ministry of Industry and Company Affairs (From 3.2.1977 to 24.3.1977*)
		Minister of Industry (From 30.7.1979 to 27.11.1979)
217.	Reddy, Shri K. Vijaya Bhaskara	Minister without portfolio (From 29.1.1983 to 2.2.1983)
		Minister of Shipping and Transport (From 2.2.1983 to 7.9.1984)
		Minister of Industry (From 7.9.1984 to 31.10.1984*)
		Minister of Industry and Company Affairs (From 4.11.1984 to 31.12.1984*)
		Minister of Law, Justice and Company Affairs (From 21.6.1991 to 9.10.1992)
218.	Reddy, Shri K.C.	Minister of Production (From 15.5.1952 to 17.4.1957*)
		Minister of Works, Housing and Supply (From 17.4.1957 to 5.4.1961)
		Minister of Commerce and Industry (From 5.4.1961 to 10.4.1962*) (Again from 10.4.1962 to 19.7.1963)
219.	Reddy, Dr. N. Sanjiva	Minister of Steel and Mines (From 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Transport, Aviation, Shipping and Tourism (From 24.1.1966 to 13.3.1967*)
220.	Routray, Shri Nilamani	Minister of Health and Family Welfare (From 6.12.1989 to 23.4.1990)
		Minister of Environment and Forests (From 23.4.1990 to 10.11.1990*)
221.	Roy, Shri Mukul	Minister without portfolio (From 22.5.2009 to 27.5.2009)
		Minister of State in the Ministry of Shipping (From 28.5.2009 to 21.1.2011)
		Minister of State in the Ministry of Shipping (From 28.5.2009 to 20.3.2012)
		Minister of State in the Ministry of Railways (From 19.5.2011 to 12.7.2011)
		Minister of Railways (From 20.3.2012 to 22.9.2012)
222.	Sahay, Shri Subodh Kant	Minister of State in the Ministry of Home Affairs (From 23.4.1990 to 5.11.1990)
		Minister of State in the Ministries of Home Affairs and Information and Broadcasting (From 21.11.1990 to 21.6.1991*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 23.5.2004 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Food Processing Industries (From 28.5.2009 to 19.1.2011)
		Minister of Tourism (From 19.1.2011 to 27.10.2012)
223.	Salve, Shri Narendra K.P.	Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 2.9.1982 to 14.2.1983)
		Minister of State in the Ministry of Steel and Mines (Independent Charge) (From 14.2.1983 to 31.10.1984*)
		Minister of State (Independent Charge) of the Ministry of Steel and Mines and also Minister of State in the Department of Parliamentary Affairs (From 4.11.1984 to 31.12.1984*)
		Minister of Power (From 18.1.1993 to 16.5.1996*)
224.	Sangma, Shri P.A.	Deputy Minister in the Ministry of Industry (From 31.10.1980 to 15.1.1982*)
		Deputy Minister in the Ministry of Commerce (From 15.1.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Commerce and Supply (From 1.1.1985 to 25.9.1985)
		Minister of State in the Ministry of Home Affairs—Department of States (From 25.9.1985 to 20.1.1986)
		Minister of State (Independent Charge) of the Ministry of Labour (From 20.9.1986 to 6.2.1988)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Coal (From 21.6.1991 to 18.1.1993)
		Also held charge of the Ministry of Labour as a Minister of State (Independent Charge) (From 10.7.1992 to 17.1.1993)
		Minister of State (Independent Charge) of the Ministry of Labour (From 17.1.1993 to 10.2.1995)
		Minister of Labour (From 10.2.1995 to 15.9.1995)
		Minister of Information and Broadcasting (From 15.9.1995 to 16.5.1996*)
225.	Sanjivayya, Shri D.	Minister of Labour and Employment (From 24.1.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Industry (From 24.1.1966 to 13.3.1967*)
		Minister of Labour and Rehabilitation (From 18.2.1970 to 18.3.1971*)
226.	Saran, Shri Daulat Ram	Minister of Urban Development (From 21.11.1990 to 21.6.1991*)
227.	Sartaj Singh, Shri	Minister of Health and Family Welfare (From 16.5.1996 to 1.6.1996*)
228.	Sathe, Shri Vasant	Minister of Information and Broadcasting (From 14.1.1980 to 2.9.1982)
		Also held temporary charge of the Ministry of Supply and Rehabilitation (From 16.1.1980 to 19.10.1980)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Chemicals and Fertilisers (From 2.9.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of Steel, Mines and Coal (From 31.12.1984 to 25.9.1985)
		Minister of Energy (From 25.9.1985 to 2.12.1989*)
		Also Minister of Steel and Mines (From 12.4.1987 to 25.7.1987)
		Held additional charge of the Ministry of Communications (From 14.2.1988 to 25.6.1988)
229.	Sayeed, Shri Mufti Mohammad	Minister of Tourism (From 12.5.1986 to 15.7.1987)
		Minister of Home Affairs (From 6.12.1989 to 10.11.1990*)
230.	Sayeed, Shri P.M.	Minister of State in the Ministry of Steel, Mines and Coal (From 4.8.1979 to 14.1.1980*)
		Minister of State in the Ministry of Home Affairs (From 19.1.1993 to 15.9.1995)
		Minister of State in the Ministry of Information and Broadcasting (From 15.9.1995 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Power (From 23.5.2004 to 18.12.2005, the date of his death)
231.	Scindia, Shri Madhav Rao	Minister of State in the Ministry of Railways (From 31.12.1984 to 25.9.1985)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Transport—Department of Railways (From 25.9.1985 to 22.10.1986)
		Minister of State (Independent Charge) of the Ministry of Railways (From 22.10.1986 to 2.12.1989*)
		Minister of Civil Aviation and Tourism (From 21.6.1991 to 9.1.1993)
		Minister of Human Resource Development (From 10.2.1995 to 17.1.1996)
232.	Sen, Dr. Triguna	Minister of Education (From 16.3.1967 to 14.2.1969)
		Minister of Petroleum and Chemicals and Mines and Metals (From 14.2.1969 to 18.3.1971*)
233.	Sen, Shri Asoke K.	Minister of Law as one of the Ministers of State (From 17.4.1957 to 12.5.1958)
		Minister of Law (From 12.5.1958 to 10.4.1962*) (Again from 10.4.1962 to 13.5.1964)
		Also held charge of the Department of Posts and Telegraphs (From 1.9.1963 to 27.5.1964)
		Minister of Law and Minister of Communications (From 13.5.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Law and Communications (From 9.6.1964 to 13.6.1964)
		Minister of Law and Social Security (From 13.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Law and Justice (From 31.12.1984 to 31.3.1987)
		Minister of Steel and Mines (From 21.11.1990 to 21.6.1991*)
234.	Sethi, Shri Arjun Charan	Minister of Water Resources (From 27.5.2000 to 22.5.2004*)
235.	Sethi, Shri Prakash Chand B.	Deputy Minister in the Ministry of Steel and Heavy Industries (From 8.6.1962 to 21.11.1963)
		Deputy Minister in the Ministry of Steel and Mines and Heavy Engineering (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Steel and Mines (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Iron and Steel (From 24.1.1966 to 13.3.1967*)
		Minister of State in the Ministry of Steel, Mines and Metals (From 13.3.1967 to 14.2.1969)
		(Consequent on the resignation of Dr. M. Chenna Reddy, looked after the work of the Department of Iron and Steel and Mines and Metals) (From 24.4.1968 to 14.2.1969)
		Minister of State in the Ministry of Finance (From 14.2.1969 to 27.6.1970)
		Minister of State (Defence Production) in the Ministry of Defence (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Petroleum and Chemicals (From 2.5.1971 to 29.1.1972)
		Minister of Chemicals and Fertilisers (From 25.12.1975 to 23.12.1976)
		Minister without portfolio (From 23.12.1976 to 24.3.1977*)
		Minister of Works and Housing (From 14.1.1980 to 19.10.1980)
		Also held charge of Ministry of Petroleum, Chemicals and Fertilisers (From 16.1.1980 to 19.10.1980)
		Minister of Petroleum, Chemicals and Fertilisers (From 19.10.1980 to 15.1.1982)
		Minister of Railways (From 15.1.1982 to 2.9.1982)
		Minister of Home Affairs (From 2.9.1982 to 19.7.1984)
		Minister of Planning (From 19.7.1984 to 31.10.1984*)
		Held additional charge of Ministry of Irrigation (From 2.8.1984 to 31.10.1984*)
236.	Shah, Shri K.K.	Minister of Information and Broadcasting (From 13.3.1967 to 14.2.1969)
		Minister of Health and Family Planning, Works, Housing and Urban Development (From 14.2.1969 to 18.3.1971*)
		Minister of Health and Family Planning (From 18.3.1971 to 19.5.1971)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
237.	Shah, Shri Manubhai	Minister of Industrial Development in the Ministry of Commerce and Industry (From 1.5.1956 to 13.6.1956)
		^{©©} Minister of Heavy Industries (From 13.6.1956 to 16.4.1957*)
		Minister of State in the Ministry of Commerce and Industry (From 17.4.1957 to 25.4.1957)
		©©Minister of State in the Ministry of Industry (From 25.4.1957 to 10.4.1962*)
		Minister of International Trade in the Ministry of Commerce and Industry (From 16.4.1962 to 19.7.1963)
		Minister of International Trade as a Minister of State (From 19.7.1963 to 27.5.1964) (Again from 27.5.1964 to 9.6.1964)
		Minister of Commerce (As one of the Ministers of State) (From 9.6.1964 to 11.1.1966) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Commerce (From 24.1.1966 to 13.3.1967*)
238.	Shakeel-Ur-Rehman	Minister of Health and Family Welfare (From 21.11.1990 to 20.2.1991**)
239.	Shankaranand, Shri B.	©©Deputy Minister in the Department of Parliamentary Affairs (From 2.5.1971 to 24.3.1977*)
		Minister of Education (From 14.1.1980 to 19.10.1980)
		Also held charge of the Ministry of Health and Family Welfare (From 16.1.1980 to 19.10.1980)

^{*} The date of dissolution of the Council of Ministers.

** Resigned w.e.f. 20.2.1991 following disqualification from membership of Lok Sabha under the Anti-Defection Law (The Tenth Schedule to the Constitution).

^{©©} Minister of Cabinet rank but not Member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Health and Family Welfare (From 19.10.1980 to 31.10.1984*) (Again From 4.11.1984 to 31.12.1984*)
		Minister of Irrigation and Power (From 31.12.1984 to 25.9.1985)
		Minister of Water Resources (From 25.9.1985 to 22.8.1987)
		Minister of Law and Justice and Minister of Water Resources (From 25.6.1988 to 4.7.1989)
		Minister of Law and Justice (From 4.7.1989 to 2.12.1989*)
		Minister of Petroleum and Natural Gas (From 21.6.1991 to 18.1.1993)
		Minister of Health and Family Welfare (From 18.1.1993 to 22.12.1994)
240.	Shanta Kumar, Shri	Minister of Consumer Affairs and Public Distribution (From 13.10.1999 to 17.7.2000)
		Minister of Consumer Affairs, Food and Public Distribution (From 17.7.2001 to 1.7.2002)
		Minister of Rural Development (From 1.7.2002 to 6.4.2003, the date of his resignation)
241.	Shanti Bhushan, Shri	Minister of Law, Justice and Company Affairs (From 26.3.1977 to 28.7.1979*)
242.	Sharma, Dr. Shanker Dayal	Minister of Communications (From 10.10.1974 to 24.3.1977*)
243.	Sharma, Shri Anand	Minister of State in the Ministry of External Affairs (From 29.1.2006 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of State in the Ministry of Information and Broadcasting (From 18.10.2008 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Commerce and Industry (From 28.5.2009 to 26.05.2014*)
		Minister of Textiles (From 12.7.2011 to 17.6.2013) (Again from 3.4.2014 to 26.05.2014*)
244.	Sharma, Shri Anant Prasad	Minister of State in the Ministry of Industry and Civil Supplies (From 10.10.1974 to 9.8.1976)
		Minister of State in the Ministry of Industry (From 9.8.1976 to 24.3.1977*)
		Minister of Shipping and Transport (From 14.1.1980 to 19.10.1980)
		Minister of Tourism and Civil Aviation (From 19.10.1980 to 2.9.1982)
		Minister of Communications (From 2.9.1982 to 14.2.1983)
245.	Shastri, Shri Bhola Paswan	Minister of Works and Housing (From 5.2.1973 to 10.10.1974)
246.	Shinde, Shri Sushil Kumar	Minister of Power (From 29.1.2006 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Power (From 28.5.2009 to 31.7.2012)
		Minister of Home Affairs (From 31.7.2012 to 26.05.2014*)
		Minister without portfolio (From 22.5.2009 to 28.5.20 Minister of Power (From 28.5.2009 to 31.7.20 Minister of Home Affairs

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
247.	Shiv Shankar, Shri P.	Minister of Law, Justice and Company Affairs (From 14.1.1980 to 15.1.1982)
		Minister of Petroleum, Chemicals and Fertilisers (From 15.1.1982 to 2.9.1982)
		Minister of Energy (From 2.9.1982 to 31.10.1984*) (Again from 31.10.1984 to 31.12.1984*)
		Minister of Commerce (From 20.1.1986 to 12.5.1986)
		Also Minister of Food and Civil Supplies (From 27.1.1986 to 12.5.1986)
		Minister of External Affairs and also Minister of Commerce (From 12.5.1986 to 22.10.1986)
		Minister of Commerce (From 22.10.1986 to 25.7.1987)
		Minister of Planning and Programme Implementation and also Minister of Law and Justice (From 25.7.1987 to 14.2.1988)
		Minister of Planning and Programme Implementation (From 14.2.1988 to 25.6.1988)
		Minister of Human Resource Development (From 25.6.1988 to 2.12.1989*)
248.	Shourie, Shri Arun	Minister of State assisting the PM in the Ministry of Statistics and Programme Implementation and in the Department of Administrative Reforms and Public Grievances of the Ministry of Personnel, Public Grievances and Pensions (From 22.11.1999 to 1.9.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State of the Ministry of Planning (From 22.11.1999 to 24.7.2000) (Again from 7.11.2000 to 1.9.2001)
		Minister of State (Independent Charge) of the Department of Disinvestment (From 24.7.2000 to 1.9.2001)
		Minister of Disinvestment (From 1.9.2001 to 22.5.2004)
		Minister of Department of Development of North Eastern Region (From 1.9.2001 to 29.1.2003)
		Minister of Commerce and Industry (From 9.11.2002 to 29.1.2003)
		Minister of Communications and information Technology and Minister of Disinvestment (From 29.1.2003 to 22.5.2004*)
249.	Shrimali, Dr. Kalu Lal	Deputy Minister in the Ministry of Education (From 14.5.1955 to 17.4.1957*)
		Minister of State in the Ministry of Education and Scientific Research (From 17.4.1957 to 10.4.1962*)
		Minister of State in the Ministry of Education (From 10.4.1958 to 10.4.1962*)
		Minister of Education (From 10.4.1962 to 31.8.1963)
250.	Shukla, Shri Vidya Charan	Deputy Minister in the Department of Parliamentary Affairs and Communications (From 24.1.1966 to 14.2.1966)
		Deputy Minister in the Ministry of Home Affairs (From 14.2.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Home Affairs (From 13.3.1967 to 27.6.1970)
		Minister of State in the Ministry of Finance (From 27.6.1970 to 2.5.1971)
		Minister of State in the Ministry of Defence (Defence Production) (From 2.5.1974 to 10.10.1974)
		Minister of State in the Ministry of Planning (From 10.10.1974 to 28.6.1975)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 28.6.1975 to 24.3.1977*)
		Minister of Civil Supplies (From 8.6.1980 to 19.3.1981)
		Minister of External Affairs (From 21.11.1990 to 20.2.1991**)
		Minister of Water Resources (From 21.6.1991 to 17.1.1996)
		Also Minister of Parliamentary Affairs (From 17.1.1993 to 17.1.1996)
251.	Sibal, Shri Kapil	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Science and Technology and Department of Ocean Development (From 23.5.2004 to 29.1.2006)
		Minister of Science and Technology and Minister of Ocean Development (From 29.1.2006 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009)

^{*} The date of dissolution of the Council of Ministers.

** Resigned w.e.f. 20.2.1991 following disqualification from membership of Lok Sabha under the Anti-Defection Law.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Human Resource Development (From 28.5.2009 to 28.10.2012)
		Minister of Science and Technology, Minister of Earth Sciences (From 10.11.2010 to 19.1.2011)
		Minister of Communication and Information Technology (From 15.11.2010 to 26.05.2014*)
		Minister of Law and Justice (From 11.5.2013 to 26.05.2014*)
252.	Sikander Bakht, Shri	Minister of Works and Housing and Supply and Rehabilitation (From 26.3.1977 to 28.7.1979*)
		Minister of Urban Affairs and Employment (From 16.5.1996 to 1.6.1996*)
		Minister of External Affairs (From 21.5.1996 to 1.6.1996*)
		Minister of Industry (From 19.3.1998 to 13.10.1999*)
253.	Singh, Dr. Raghuvansh Prasad	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Rural Development (From 23.5.2004 to 22.5.2009*)
254.	Singh, Dr. Ram Subhag	Minister of State in the Ministry of Food and Agriculture (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Social Security and Cottage Industries (As one of the Ministers of State) (From 9.6.1964 to 13.6.1964)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Railways (From 13.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)
		Minister of Parliamentary Affairs and Communications (From 13.3.1967 to 14.2.1969)
		Minister of Railways (From 14.2.1969 to 4.11.1969)
255.	Singh, Shri Bhishma Narain	Minister of Parliamentary Affairs (From 14.1.1980 to 17.10.1980)
		Also held temporary charge of the Ministry of Communications (From 16.1.1980 to 3.3.1980)
		Minister of Parliamentary Affairs and Works and Housing (From 17.10.1980 to 29.1.1983)
		Minister of Civil Supplies and Cooperation (From 29.1.1983 to 14.2.1983)
256.	Singh, Shri Bir Bahadur	Minister of Communications (From 25.6.1988 to 30.5.1989)
257.	Singh, Shri Rao Birendra	Minister of Agriculture (From 14.1.1980 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Also held temporary charge of Ministry of Rural Reconstruction/Rural Development (From 20.1.1980 to 29.1.1983)
		Minister of Food and Civil Supplies (From 31.12.1984 to 25.9.1985)
		Also held temporary charge of the Ministry of Civil Supplies and Cooperation (From 19.3.1981 to 2.9.1982)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also held charge of the Ministry of Irrigation (From 12.11.1980 to 15.1.1982)
		Ministry of Food and Civil Supplies (From 21.11.1990 to 21.6.1991*)
258.	Singh, Shri Virbhadra	Deputy Minister in the Ministry of Tourism and Civil Aviation (From 31.12.1976 to 24.3.1977*)
		Minister of State in the Ministry of Industry (From 11.9.1982 to 8.4.1983)
		Minister of Steel (From 28.5.2009 to 19.1.2011)
		Minister of Micro, Small and Medium Enterprises (From 19.1.2011 to 26.6.2012)
259.	Sinha, Shri Satya Narayan	Deputy Minister in the Department of Parliamentary Affairs (From 1.10.1948 to 26.2.1949)
		Minister of State in the Department of Parliamentary Affairs (From 26.2.1949 to 25.1.1950*) (Again from 31.1.1950 to 6.5.1950*) (Again from 13.5.1950 to 13.5.1952*) (Again from 13.5.1952 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962*)
		Minister of Parliamentary Affairs (From 10.4.1962 to 10.3.1964*)
		Also held charge of the Ministry of Information and Broadcasting (From 1.9.1963 to 10.3.1964*)
		Minister of Parliamentary Affairs and Minister of Information and Broadcasting (From 10.3.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Parliamentary Affairs and Civil Aviation (From 9.6.1964 to 13.6.1964)
		Also held charge of the Ministry of Information and Broadcasting (From 11.6.1964 to 2.7.1964)
		Minister of Communications and Parliamentary Affairs (From 13.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Parliamentary Affairs and Communications (From 24.1.1966 to 13.3.1967*)
		Minister without portfolio (From 13.3.1967 to 14.11.1967)
		Minister of Health, Family Planning and Urban Development (From 14.11.1967 to 14.2.1969)
		Minister of Information and Broadcasting and Communications (From 14.2.1969 to 8.3.1971)
260.	Sinha, Shri Shatrughan	Minister of Health and Family Welfare (From 1.7.2002 to 29.1.2003)
		Minister of Shipping (From 29.1.2003 to 22.5.2004*)
261.	Sinha, Shri Yashwant	Minister of Finance (From 21.11.1990 to 21.6.1991*)
		Minister of Finance (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 1.7.2002)
		Minister of External Affairs (From 1.7.2002 to 22.5.2004*)
262.	Solanki, Shri Madhavsinh	Minister of Planning and Programme Implementation and Deputy Chairman, Planning Commission (From 25.6.1988 to 2.12.1989*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of External Affairs (From 21.6.1991 to 31.3.1992)
263.	Soni, Smt. Ambika	Minister of Tourism and Minister of Culture (From 29.1.2006 to 1.6.2006*)
		Miniser of Tourism (From 1.6.2006 to 22.5.2009*)
		Also Minister of Culture (From 1.6.2006 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Information and Broadcasting (From 28.5.2009 to 27.10.2012)
264.	Soren, Shri Shibu	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Coal and Mines (From 23.5.2004 to 24.7.2004, the date of his resignation)
		Minister of Coal (From 27.11.2004 to 2.3.2005, the date of his resignation) (Again from 29.1.2006 to 29.11.2006, the date of his resignation)
265.	Soz, Prof. Saifuddin	Minister of Environment and Forests (From 21.2.1997 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
		Minister of Water Resources (From 29.1.2006 to 22.5.2009*)
266.	Sri Prakasa, Shri	Minister of Commerce (From 29.5.1950 to 26.12.1950)
		Minister of Natural Resources and Scientific Research (From 26.12.1950 to 12.3.1952, the date of his resignation)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
267.	Stephen, Shri C.M.	Minister of Communications (From 3.3.1980 to 2.9.1982)
		Minister of Shipping and Transport (From 2.9.1982 to 2.2.1983)
268.	Subbaryan, Dr. P.	Minister of Transport and Communications (From 2.9.1959 to 10.4.1962*)
269.	Subramaniam, Shri C.	Minister of Steel and Heavy Industries (From 10.4.1962 to 21.11.1963)
		Minister of Steel, Mines and Heavy Engineering (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Food and Agriculture (From 9.6.1964 to 11.1.1966*) (Again From 11.1.1966 to 24.1.1966*)
		Also exercised overall supervision over the work of the Ministry of Community Development and Cooperation (From 18.7.1964 to January, 1966)
		Minister of Food, Agriculture, Community Development and Cooperation (From 24.1.1966 to 13.3.1967*)
		Minister of Planning (From 24.4.1971 to 2.5.1971)
		Minister of Planning and Science and Technology (From 2.5.1971 to 22.7.1972)
		Minister of Industrial Development and Science and Technology (From 22.7.1972 to 10.10.1974)
		Also Minister of Agriculture (From 3.7.1974 to 10.10.1974)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Finance (From 10.10.1974 to 24.3.1977*)
		Minister of Defence (From 30.7.1979 to 14.1.1980*)
270.	Suraj Bhan, Shri	Minister of Agriculture (From 16.5.1996 to 1.6.1996*)
271.	Sushma Swaraj, Smt.	Minister of Information and Broadcasting (From 16.5.1996 to 1.6.1996*)
		Minister of Information and Broadcasting (From 19.3.1998 to 11.10.1998, the date of her resignation)
		Held charge of the Ministry of Communications (From 20.4.1998 to 11.10.1998)
		Minister of Information and Broadcasting (From 30.9.2000 to 29.1.2003)
		Minister of Health and Family Welfare and Minister of Parliamentary Affairs (From 29.1.2003 to 22.5.2004*)
272.	Swamy, Dr. Subramaniam	Minister of Commerce with additional charge of the Ministry of Law and Justice (From 21.11.1990 to 21.6.1991*)
273.	Swaran Singh, Shri	Minister of Works, Housing and Supply (From 13.5.1952 to 17.4.1957*)
		Also held charge of the Ministry of Commerce and Consumer Industries (From 30.8.1956 to 14.11.1956)
		Minister of Steel, Mines and Fuel (From 17.4.1957 to 10.4.1962*)
		Minister of Railways (From 10.4.1962 to 1.9.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
	Nume	
		Minister of Mines and Fuel (in addition) (From 26.6.1963 to 19.7.1963)
		Minister of Food and Agriculture (From 1.9.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Railways (in addition) (From 1.9.1963 to 21.9.1963)
		Minister of Industry (From 9.6.1964 to 20.6.1964)
		Minister of Industry and Supply (From 20.6.1964 to 19.7.1964)
		Minister of External Affairs (From 19.7.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.11.1966)
		Minister of Defence (From 13.11.1966 to 13.3.1967*) (Again from 13.3.1967 to 27.6.1970)
		Also Minister of Steel and Heavy Engineering (From 15.11.1969 to 27.6.1970)
		Minister of External Affairs (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 10.10.1974)
		Minister of Defence (From 10.10.1974 to 1.12.1975)
274.	Thakur, Dr. C.P.	Minister of Water Resources (From 22.11.1999 to 27.5.2000)
		Minister of Health and Family Welfare (From 27.5.2000 to 1.7.2002, the date of his resignation)
		Minister of Small Scale Industries and Minister of Development of North Eastern Region (From 29.1.2003 to 22.5.2004)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
275.	Thambi Durai, Shri M.	Minister of Law, Justice and Company Affairs (From 19.3.1998 to 8.4.1999, the date of his resignation)
		Also Minister of Surface Transport (From 8.4.1998 to 8.4.1999)
276.	Tiwari, Shri Narayan Dutt	Minister of Planning and Deputy Chairman, Planning Commission (From 8.6.1980 to 8.8.1981)
		Minister of Industry (From 8.8.1981 to 3.8.1984)
		Held charge of Ministry of Labour (From 19.10.1980 to 8.8.1981)
		Also held charge of Ministry of Steel and Mines (From 15.1.1982 to 14.2.1983)
		Also held charge of Department of Coal (From 2.9.1982 to 6.9.1982)
		Minister of Industry (From 25.9.1985 to 22.10.1986)
		Minister of Industry and also Minister of Petroleum and Natural Gas (From 24.6.1986 to 22.10.1986)
		Minister of External Affairs (From 22.10.1986 to 25.7.1987)
		Minister of Finance and Commerce (From 25.7.1987 to 25.6.1988)
277.	Tripathi, Shri Kamalapati	Minister of Shipping and Transport (From 8.11.1973 to 10.2.1975)
		Minister of Railways (From 10.2.1975 to 24.3.1977*) (Again from 14.1.1980 to 12.11.1980)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
278.	Trivedi, Shri Dinesh	Minister of State in the Ministry of Health and Family Welfare (From 28.5.2009 to 12.7.2011)
		Minister of Railways (From 12.7.2011 to 19.3.2012)
279.	Tyagi, Shri Mahavir	Minister of State in the Ministry of Finance (From 16.2.1951 to 13.5.1952*) (Again from 13.5.1952 to 21.8.1952)
		Minister of Revenue and Expenditure (As one of the Ministers of State) (From 21.8.1952 to 16.3.1953)
		Minister of Defence Organisation (As one of the Ministers of State) (From 16.3.1953 to 17.4.1957*)
		Minister of Rehabilitation (From 16.4.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 15.1.1966, the date of his resignation)
280.	Uma Bharati, Sushri	Minister of State in the Ministry of Human Resource Development (From 19.3.1998 to 13.10.1999*)
		Held charge of the Department of Youth Affairs, Sports, Women and Child Development as a Minister of State (Independent Charge) (From 1.3.1999 to 13.10.1999*)
		Minister of State (Independent Charge) of the Ministry of Tourism (From 13.10.1999 to 2.2.2000, the date of her resignation)
		Minister of Youth Affairs and Sports (From 7.11.2000 to 26.8.2002)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Coal and Mines (From 26.8.2002 to 29.1.2003, the date of her resignation)
281.	Unnikrishnan, Shri K.P.	Minister of Surface Transport with additional charge of the Ministry of Communications (From 6.12.1989 to 23.4.1990)
		Minister of Surface Transport (From 23.4.1990 to 10.11.1990*)
282.	Upendra, Shri P.	Minister of Information and Broadcasting and Parliamentary Affairs (From 6.12.1989 to 10.11.1990*)
283.	Vaghela, Shri Shankersinh	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Textiles (From 23.5.2004 to 22.5.2009*)
284.	Varma, Shri Beni Prasad	Minister of State in the Ministries of Communications and Parliamentary Affairs (From 1.6.1996 to 29.6.1996)
		Minister of State in the Ministry of Communications (Independent Charge) (From 29.6.1996 to 10.7.1996)
		Minister of Communications (From 10.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998)
		Minister of State (Independent Charge) of the Ministry of Steel (From 19.1.2011 to 12.7.2011)
		Minister of Steel (From 12.7.2011 to 26.05.2014*)
285.	Varma, Shri Ravindra	Minister of Parliamentary Affairs and Labour (From 26.3.1977 to 28.7.1979*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
286.	Vasan, Shri G.K.	Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation (From 29.1.2006 to 22.5.2009*)
		Also Minister of State (Independent Charge) of the Ministry of Labour and Employment (From 3.3.2009 to 22.5.2009*)
		Minister of Shipping (From 28.5.2009 to 26.05.2014*)
287.	Vayalar Ravi, Shri	Minister of Overseas Indian Affairs (From 29.1.2006 to 22.5.2009*)
		Held additional charge of Ministry of Parliamentary Affairs (From 6.4.2008 to 22.5.2009*)
		Minister without portfolio (From 22.5.2009 to 28.5.2009)
		Minister of Overseas Indian Affairs (From 28.5.2009 to 26.05.2014*)
		Minister of Civil Aviation (From 19.1.2011 to 18.12.2011)
		Minister of Science and Technology, Minister of Earth Sciences, Minister of Micro, Small and Medium Enterprises (From 10.8.2012 to 28.10.2012)
288.	Venkataraman, Shri R.	Minister of Finance (From 14.1.1980 to 15.1.1982)
		Also held charge of Ministry of Industry (From 16.1.1980 to 3.4.1980)
		Minister of Defence (From 15.1.1982 to 2.8.1984)
		Also held charge of the Ministry of Home Affairs (From 22.6.1982 to 2.9.1982)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
289.	Venkatraman, Shri T.G.	Minister of Surface Transport (From 1.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
		Held charge of the Ministry of Urban Affairs and Employment (From 14.11.1997 to 10.12.1997)
290.	Venkatswamy, Shri G.	Deputy Minister in the Ministry of Labour and Rehabilitation (From 5.2.1973 to 9.11.1973)
		Deputy Minister in the Ministry of Supply and Rehabilitation (From 9.11.1973 to 24.3.1977*)
		Minister of State in the Ministry of Rural Development (From 21.6.1991 to 2.7.1992)
		Minister of State in the Ministry of Rural Development—Department of Rural Development (From 2.7.1992 to 18.1.1993)
		Minister of State (Independent Charge) of the Ministry of Textiles (From 18.1.1993 to 10.2.1995)
		Minister of Textiles (From 10.2.1995 to 15.9.1995)
		Minister of Labour (From 15.9.1995 to 16.5.1996*)
		Held additional charge of the Ministry of Textiles (From 20.2.1996 to 16.5.1996*)
291.	Verma, Dr. Sahib Singh	Minister of Labour (From 1.7.2002 to 22.5.2004*)
292.	Verma, Shri Brij Lal	Minister of Industry (From 28.3.1977 to 6.7.1977)
		Minister of Communications (From 6.7.1977 to 28.7.1979*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
293.	Verma, Shri Vikram	Minister of Youth Affairs and Sports (From 26.8.2002 to 22.5.2004*)
294.	Vora, Shri Motilal	Minister of Health and Family Welfare (From 14.2.1988 to 24.1.1989)
		Also held temporary charge of the Ministry of Civil Aviation (From 14.2.1988 to 25.6.1988)
295.	Vyas, Dr. (Km.) Girija	Deputy Minister in the Ministry of Information and Broadcasting (From 21.6.1991 to 17.1.1993)
		Minister of Housing and Urban Poverty Alleviation (From 17.6.2013 to 26.05.2014*)
296.	Wasnik, Shri Mukul	Minister of State in the Ministry of Human Resource Development (Department of Youth Affairs and Sports) (From 18.1.1993 to 16.5.1996*)
		Also Minister of State in the Ministry of Parliamentary Affairs (From 19.1.1993 to 16.5.1996*)
		Minister of Social Justice and Empowerment (From 28.5.2009 to 27.10.2012)
297.	Yadav, Shri Devendra Prasad	Minister of Food and Minister of Civil Supplies, Consumer Affairs and Public Distribution (From 1.6.1996 to 21.4.1997*)
		Also Minister of Commerce (From 1.6.1996 to 29.6.1996)
298.	Yadav, Shri Lalu Prasad	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of Railways (From 23.5.2004 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
299.	Yadav, Shri Mulayam Singh	Minister of Defence (From 1.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
300.	Yadav, Shri Ram Lakhan Singh	Minister of Chemicals and Fertilisers (From 17.2.1994 to 16.5.1996*)
301.	Yadav, Shri Sharad	Minister of Food Processing Industries and Textiles (From 6.12.1989 to 10.11.1990*)
		Minister of Civil Aviation (From 13.10.1999 to 1.9.2001)
		Minister of Labour (From 1.9.2001 to 1.7.2002)
		Minister of Consumer Affairs, Food and Public Distribution (From 1.7.2002 to 22.5.2004*)
302.	Zail Singh, Giani	Minister of Home Affairs (From 14.1.1980 to 22.6.1982)

^{*}The date of dissolution of the Council of Ministers.

PART IVMINISTERS OF CABINET RANK BUT NOT MEMBERS OF THE CABINET

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Deshmukh, Dr. P.S.	Minister of Food and Agriculture (From 12.8.1952 to 21.8.1952)
		Minister of Agriculture (From 21.8.1952 to 17.4.1957*)
		Also held charge of the Ministry of Food and Agriculture (From 29.10.1954 to 25.11.1954)
		Minister of State in the Ministry of Food and Agriculture (From 17.4.1957 to 25.4.1957)
		Minister of State for Cooperation (From 25.4.1957 to 31.12.1958)
		Minister of Agriculture (From 31.12.1958 to 9.4.1962*)
2.	Dey, Shri Surendra Kumar	Minister of State for Community Development (From 20.9.1956 to 17.4.1957*)
		Minister of State for Community Development (From 17.4.1957 to 31.12.1958)
		Minister of State for Community Development and Cooperation (From 31.12.1958 to 10.4.1962*)
		Minister of State for Community Development, Panchayati Raj and Cooperation (From 10.4.1962 to 16.2.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State for Community Development and Cooperation (From 16.2.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State for Mines and Metals (From 24.1.1966 to 17.1.1967)
3.	Guha, Shri Arun Chandra	Deputy Minister in the Ministry of Finance (From 18.3.1953 to 7.12.1954)
		Minister of Revenue and Defence Expenditure (From 7.12.1954 to 17.4.1957*)
4.	Khanna, Shri Mehar Chand	Minister of Rehabilitation (From 7.12.1954 to 17.4.1957*) (Again from 17.4.1957 to 22.4.1957)
		Minister of Rehabilitation and Minority Affairs (From 22.4.1957 to 26.10.1961)
		Minister of Rehabilitation (From 26.10.1961 to 10.4.1962*)
		Minister of Works, Housing and Supply (From 10.4.1962 to 15.11.1962)
		Minister of Works, Housing and Rehabilitation (From 15.11.1962 to 16.4.1964)
		Minister of Works and Housing (From 16.4.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Also Minister of Rehabilitation (From 15.1.1966 to 24.1.1966*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of Works, Housing and Urban Development (From 24.1.1966 to 13.3.1967*)
5.	Pataskar, Shri H.V.	Minister in the Ministry of Law (From 7.12.1954 to 23.8.1955)
		Minister of Legal Affairs (From 23.8.1955 to 13.12.1956)
		Also Minister of Civil Aviation (Temporary Charge) (From 7.12.1956 to 13.12.1956)
		Minister of Legal Affairs and Civil Aviation (From 13.12.1956 to 16.4.1957)
6.	Shah, Shri M.C.	Deputy Minister in the Ministry of Finance (From 12.8.1952 to 7.12.1952)
		Minister of Revenue and Civil Expenditure (From 7.12.1954 to 17.4.1957*)
7.	Singh, Shri T.N.	Minister of Heavy Engineering in the Ministry of Industry and Supply (From 25.7.1964 to 5.2.1965)
		Also held charge of the Department of Industry (From 30.10.1964 to 5.2.1965)
		Minister of Industry in the Ministry of Industry and Supply (From 5.2.1965 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Iron and Steel (From 29.1.1966 to 13.3.1967*)
		Also Minister of Mines and Metals (From 17.1.1967 to 13.3.1969*)
8.	Syed, Mahmud, Dr.	Minister in the Ministry of External Affairs (From 7.12.1954 to 17.4.1957*)

^{*}The date of dissolution of the Council of Ministers.

PART V

MINISTERS OF STATE WITH INDEPENDENT CHARGE OF THEIR MINISTRIES/DEPARTMENTS

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Alagh, Shri Yogendra K.	Minister of State (Independent Charge) of the Ministry of Planning and Programme Implementation and the Ministry of Science and Technology (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State (Independent Charge) of the Ministry of Science and Technology and Ministry of Power (From 9.6.1997 to 19.3.1998*)
2.	Alam, Shri Rafique	Deputy Minister in the Ministry of Petroleum and Natural Gas (From 14.2.1988 to 25.6.1988)
		Minister of State in the Ministry of Textiles (From 25.6.1988 to 4.7.1989)
		Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 4.7.1989 to 2.12.1989*)
3.	Anjiah, Shri T.	Minister of State in the Ministry of Labour (From 8.6.1980 to 11.10.1980)
		Minister of State (Independent Charge) of the Ministry of Labour (From 31.12.1984 to 20.1.1986)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
4.	Azad, Shri Bhagwat Jha	Minister of State in the Ministry of Education (From 18.3.1967 to 14.2.1969)
		Minister of State in the Ministry of Labour, Employment and Rehabilitation (From 14.2.1969 to 18.3.1971*)
		Minister of State (Independent Charge) of the Ministry of Supply and Rehabilitation (From 19.10.1980 to 15.1.1982)
		Minister of State (Independent Charge) of the Ministry of Labour (From 15.1.1982 to 2.9.1982)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation and Civil Supplies (From 2.9.1982 to 14.2.1983)
		Minister of State (Independent Charge) of the Ministry of Food and Civil Supplies (From 14.2.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
5.	Bais, Shri Ramesh	Minister of State in the Ministry of Steel and Mines (From 19.3.1998 to 13.10.1999*)
		Minister of State in the Ministry of Chemicals and Fertilizers (From 13.10.1999 to 30.9.2000)
		Minister of State in the Ministry of Information and Broadcasting (From 30.9.2000 to 29.1.2003)
		Minister of State (Independent Charge) of the Ministry of Mines (From 29.1.2003 to 9.1.2004)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 9.1.2004 to 22.5.2004*)
6.	Bajpayee, Dr. Rajendra Kumari	Minister of State (Independent Charge) of the Ministry of Welfare (From 25.9.1985 to 2.12.1989*)
7.	Chandra Shekhar, Smt. M.	Deputy Minister in the Ministry of Health (From 12.8.1952 to 16.4.1957)
		Deputy Minister in the Ministry of Home Affairs (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Department of Social Security (From 15.6.1964 to 11.1.1966*)
		Deputy Minister in the Department of Social Welfare (From 24.1.1966 to 13.3.1967*)
		Minister of State (Independent Charge) of the Ministry of Women & Social Welfare (From 31.12.1984 to 25.9.1985)
8.	Chavan, Shri Prithviraj	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Prime Minister's Office (From 23.5.2004 to 22.5.2009*)
		Also Minister of State in the Ministry of Personnel, Public Grievances and Pensions (From 6.4.2008 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Science and Technology, Minister of State (Independent Charge) of the Ministry of Earth Sciences, Minister of State in the Prime Minister's Office, Minister of State in the Ministry of Personnel, Public Grievances and Pensions and Minister of State in the Ministry of Parliamentary Affairs (From 28.5.2009 to 10.11.2010)
9.	Chiranjeevi, Dr. K.	Minister of State (Independent Charge) of the Ministry of Tourism (From 28.10.2012 to 26.05.2014*)
10.	Chowdhury, Smt. Renuka	Minister of State (without portfolio) (From 3.6.1997 to 9.6.1997)
		Minister of State in the Ministry of Health and Family Welfare (From 9.6.1997 to 19.3.1998*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Tourism (From 23.5.2004 to 29.1.2006)
		Minister of State (Independent Charge) of the Ministry of Women & Child Development (From 29.1.2006 to 22.5.2009*)
11.	Dattatreya, Shri Bandaru	Minister of State in the Ministry of Urban Affairs (From 20.3.1998 to 13.10.1999*)
		Minister of State in the Ministry of Urban Development (From 13.10.1999 to 14.6.2000*)
		Minister of State in the Ministry of Urban Development and Poverty Alleviation (From 14.6.2000 to 1.7.2002)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Railways (From 1.7.2002 to 8.9.2003)
		Minister of State (Independent Charge) of the Ministry of Urban Development and Poverty Alleviation (From 8.9.2003 to 22.5.2004*)
12.	Dhawan, Shri Harmohan	Minister of State (Independent Charge) of the Ministry of Civil Aviation (From 21.11.1990 to 21.6.1991)
13.	Dhawan, Shri R.K.	Minister of State (Independent Charge) of the Department of Urban Development in the Ministry of Urban Affairs and Employment (From 15.9.1995 to 21.2.1996)
14.	Dutt, Shri Brahm	Minister of State in the Ministry of Commerce (From 12.5.1986 to 22.10.1986)
		Minister of State (Independent Charge) of the Ministry of Petroleum and Natural Gas (From 22.10.1986 to 2.12.1989*)
		Also Minister of State in the Ministry of Finance (From 24.1.1987 to 25.7.1987)
15.	Ezhilmalai, Shri Dalit	Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 20.3.1998 to 14.8.1999, the date of his resignation)
16.	Gadgil, Shri V.N.	Minister of State for Defence Production in the Ministry of Defence (From 25.12.1975 to 24.3.1977*)
		Minister of State in the Ministry of Communications (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 31.12.1984 to 22.10.1986)
		Also held Independent Charge of the Ministry of Communications as a Minister of State (From 26.9.1986 to 19.10.1986)
17.	Gamang, Shri G. Giridhar	Deputy Minister in the Ministry of Supply and Rehabilitation (From 15.1.1982 to 6.9.1982)
		Deputy Minister in the Ministry of Labour (From 6.9.1982 to 29.1.1983)
		Deputy Minister in the Ministry of Welfare (From 27.9.1985 to 15.2.1988)
		Minister of State in the Ministry of Tourism (From 15.2.1988 to 25.6.1988)
		Minister of State in the Ministry of Communications (From 25.6.1988 to 4.7.1989)
		Minister of State (Independent Charge) of the Ministry of Communications (From 4.7.1989 to 2.12.1989*)
		Minister of State (Independent Charge) of Ministry of Food Processing Industries (From 21.6.1991 to 17.1.1993)
		Minister of State (Independent Charge) of the Ministry of Planning and Programme Implementation (From 17.1.1993 to 5.9.1995)
		Minister of State (Independent Charge) of the Ministry of Mines (From 15.9.1995 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
18.	Gandhi, Smt. Maneka	Minister of State in the Ministry of Environment and Forests (From 6.12.1989 to 6.11.1990)
		Also held the additional charge of the Ministry of Programme Implementation (From 20.1.1990 to 23.4.1990)
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 21.11.1990 to 21.6.1991*)
		Minister of State (Independent Charge) of the Ministry of Welfare (From 19.3.1998 to 23.5.1998)
		Minister of State (Independent Charge) of the Ministry of Social Justice and Empowerment (From 23.5.1998 to 13.10.1999*) (Again from 13.10.1999 to 1.9.2001)
		Minister of State (Independent Charge) of the Ministry of Culture (From 1.9.2001 to 18.11.2001)
		Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation (From 18.11.2001 to 1.7.2002, the date of her resignation)
19.	Gautam, Shri Sangh Priya	Minister of State (Independent Charge) of the Ministry of Agro and Rural Industries (From 29.1.2003 to 22.5.2004*)
20.	Gehlot, Shri Ashok	Deputy Minister in the Ministry of Tourism (From 2.9.1982 to 14.2.1983)
		Deputy Minister in the Ministry of Tourism and Civil Aviation (From 14.2.1983 to 7.2.1984)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Sports (From 7.2.1984 to 31.10.1984*) (Again from 12.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Tourism and Civil Aviation (From 31.12.1984 to 24.8.1985)
		Minister of State (Independent Charge) of the Ministry of Textiles (From 21.6.1991 to 17.1.1993)
21.	Ghatowar, Shri Paban Singh	Deputy Minister in the Ministry of Labour (From 21.6.1991 to 18.1.1993)
		Deputy Minister in the Ministry of Health and Family Welfare (From 18.1.1993 to 15.9.1995)
		Minister of State in the Ministry of Health and Family Welfare— Department of Indian System of Medicines and Homoeopathy (From 15.9.1995 to 16.5.1996*)
		Minister of State (Independent Charge) of the Ministry of Development of North-Eastern Region (From 12.7.2011 to 26.05.2014*)
		Minister of State in the Ministry of Parliamentary Affairs (From 20.7.2011 to 26.05.2014*)
22.	Gogoi, Shri Tarun	Minister of State (Independent Charge) of the Ministry of Food (From 21.6.1991 to 18.1.1993)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industry (From 18.1.1993 to 13.9.1995)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
23.	Hasan, Shri Nurul Prof. S.	Minister of State in the Ministry of Education and Social Welfare and in the Department of Culture (From 4.10.1971 to 24.3.1972)
		Minister of State (Independent Charge) of the Ministry of Education, Social Welfare and Culture (From 24.3.1972 to 24.3.1977*)
24.	Jena, Shri Srikant	Minister of State in the Ministry of Industry (Department of Small Scale Industries and Agro and Rural Industries) (From 23.4.1990 to 10.11.1990)
		Minister of Parliamentary Affairs and Tourism (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
		Minister of State in the Ministry of Chemicals and Fertilizers (From 28.5.2009 to 20.3.2013)
		Minister of State (Independent Charge) of the Ministry of Chemicals and Fertilizers (From 20.3.2013 to 26.05.2014*)
		Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation (From 12.7.2011 to 26.05.2014*)
25.	Kamat, Shri Gurudas	Minister of State in the Ministry of Communications and Information Technology (From 28.5.2009 to 19.1.2011) (Again from 21.1.2011 to 12.7.2011)
		Minister of State in the Ministry of Home Affairs (From 19.1.2011 to 12.7.2011)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Drinking Water and Sanitation (From 12.7.2011 to 13.7.2011)
26.	Kannappan, Shri M.	Minister of State (Independent Charge) of the Ministry of Non-conventional Energy Sources (From 13.10.1999 to 30.12.2003, the date of his resignation)
27.	Khalap, Shri Ramakant D.	Minister of State (Independent Charge) of the Ministry of Law, Justice and Company Affairs (excluding Department of Company Affairs) (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)
28.	Khan, Shri Khurshid Alam	Minister of State in the Ministry of Commerce (From 19.10.1980 to 15.1.1982)
		Minister of State in the Ministry of Tourism and Civil Aviation (From 15.1.1982 to 2.9.1982)
		Minister of State (Independent Charge) of the Ministry of Tourism (From 2.9.1982 to 14.2.1983)
		Minister of State (Independent Charge) of the Ministry of Tourism and Civil Aviation (From 14.2.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of External Affairs (From 31.12.1984 to 25.9.1985)
		Minister of State (Independent Charge) of the Ministry of Commerce (From 25.9.1985 to 15.11.1985)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Textiles (From 15.11.1985 to 22.10.1986)
29.	Masood, Shri Rasheed	Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 23.4.1990 to 10.11.1990*)
30.	Muniyappa, Shri K.H.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Road Transport and Highways (From 23.5.2004 to 3.9.2004*)
		Minister of State in the Ministry of Shipping, Road Transport and Highways (From 3.9.2004 to 22.5.2009)
		Minister of State in the Ministry of Railways (From 28.5.2009 to 28.10.2012)
		Minister of State (Independent Charge) of the Ministry of Micro, Small and Medium Enterprises (From 28.10.2012 to 26.05.2014*)
31.	Murthy, Shri M. Rajasekara	Minister of State (Independent Charge) of the Ministry of Surface Transport (From 15.9.1995 to 16.5.1996*)
32.	Muttemwar, Shri Vilas Baburao	Minister of State in the Ministry of Rural Areas and Employment (Department of Rural Employment and Poverty Alleviation) and Parliamentary Affairs (From 15.9.1995 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Non-Conventional Energy Sources (From 23.5.2004 to 22.5.2009*)
33.	Natarajan, Smt. Jayanthi	Minister of State (without portfolio) (From 3.6.1997 to 9.6.1997)
		Minister of State in the Ministry of Civil Aviation and in Parliamentary Affairs (From 9.6.1997 to 19.3.1998*)
		Minister of State in the Ministry of Coal (From 10.1.1998 to 19.3.1998*)
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 12.7.2011 to 21.12.2013)
34.	Nishad, Shri Jainarayan Prasad	Minister of State in the Ministry of Environment and Forests (From 1.6.1996 to 29.6.1996)
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 29.6.1996 to 21.2.1997)
		Minister of State (Independent Charge) of the Ministry of Non-conventional Energy Sources (From 21.2.1997 to 21.4.1997*) (Again from 21.4.1997 to 10.1.1998)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
35.	Patel, Dr. (Shri) A.K.	Minister of State in the Ministry of Chemicals and Fertilisers (From 19.3.1998 to 13.10.1999*)
		Also Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 16.8.1999 to 13.10.1999*)
36.	Patil, Shri Babagounda	Minister of State (Independent Charge) of the Ministry of Rural Development (From 20.3.1998 to 13.10.1999*)
37.	Pilot, Shri Rajesh	Minister of State in the Ministry of Transport—Department of Surface Transport (From 25.9.1985 to 22.10.1986)
		Minister of State (Independent Charge) of the Ministry of Surface Transport (From 22.10.1986 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Communications (From 21.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Home Affairs—Department of Internal Security (From 18.1.1993 to 15.9.1995)
		Minister of State (Independent Charge) of the Ministry of Environment and Forests (From 15.9.1995 to 16.5.1996*)
38.	Pilot, Shri Sachin	Minister of State in the Ministry of Communications and Information Technology (From 28.5.2009 to 28.10.2012)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Corporate Affairs (From 28.10.2012 to 26.05.2014*)
39.	Prasad, Shri Ravi Shankar	Minister of State in the Ministry of Coal and Mines (From 1.9.2001 to 1.7.2002)
		Minister of State in the Ministry of Law and Justice and in the Ministry of Coal and Mines (From 1.7.2002 to 29.1.2003)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 29.1.2003 to 22.5.2004*)
40.	Rai, Shri Kalpnath	Deputy Minister in the Department of Parliamentary Affairs (From 15.1.1982 to 6.9.1982)
		Deputy Minister in the Ministry of Industry (From 6.9.1982 to 29.1.1983)
		Minister of State in the Department of Parliamentary Affairs (From 29.1.1983 to 31.10.1984*)
		Minister of State in the Ministry of Energy—Department of Power (From 25.6.1988 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Power and Non- conventional Energy Sources (From 21.6.1991 to 2.7.1992)
		Minister of State (Independent Charge) of the Ministry of Power (From 2.7.1992 to 18.1.1993)
		Minister of State (Independent Charge) of the Ministry of Food (From 18.1.1993 to 21.12.1994)

^{*}The date of dissolution of the Council of Ministers.

eriod thereof
Ministry of .10.1999*)
Ministry of vances and State 10.1999*)
dent Charge) mall Scale al Industries .9.2001)
ng the Prime of Personnel, ensions 2.11.1999)
Department Department 9.1.2003, the
Ministry of Pensions, and Public
dent Charge) ale Industries 1.2003)
y of Planning .1.2003)
dent Charge) erce .4.1997*) o 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of State (Independent Charge) of the Ministry of Textiles (From 20.1.1998 to 19.3.1998*)
43.	Ray, Shri Dilip Kumar	Minister of State (Independent Charge) of the Department of Animal Husbandry and Dairying (From 29.6.1996 to 6.7.1996)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 6.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 25.12.1997)
		Minister of State (Independent Charge) of the Ministry of Coal (From 20.3.1998 to 13.10.1999*)
		Minister of State in the Ministry of Parliamentary Affairs (From 22.5.1998 to 13.10.1999*) (Again from 13.10.1999 to 22.10.1999)
		Minister of State (Independent Charge) of the Ministry of Steel (From 13.10.1999 to 27.5.2000)
44.	Rudy, Shri Rajiv Pratap	Minister of State in the Ministry of Commerce and Industry (From 1.9.2002 to 24.5.2003)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation (From 24.5.2003 to 22.5.2004*)
45.	Scindia, Shri Jyotiraditya Madhavrao	Minister of State in the Ministry of Communications and Information Technology (From 6.4.2008 to 22.5.2009*)
		Minister of State in the Ministry of Commerce and Industry (From 28.5.2009 to 28.10.2012)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Power (From 28.10.2012 to 26.05.2014*)
46.	Shanmugam, Shri N.T.	Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 13.10.1999 to 27.5.2000)
		Minister of State (Independent Charge) of the Ministry of Coal (From 27.5.2000 to 7.2.2001, the date of his resignation)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 1.7.2002 to 15.1.2004, the date of his resignation)
47.	Sharma, Capt. Satish	Minister of State (Independent Charge) of the Ministry of Petroleum and Natural Gas (From 18.1.1993 to 16.5.1996*)
48.	Sharma, Shri Nawal Kishore	Minister of State in the Ministry of Finance (From 12.11.1984 to 31.12.1984*)
		Minister of State (Independent Charge) of the Ministry of Petroleum (From 31.12.1984 to 25.9.1985)
		Minister of State (Independent Charge) of the Ministry of Petroleum and Natural Gas (From 25.9.1985 to 20.1.1986)
49.	Singh, Deo Shri K.P.	Deputy Minister in the Ministry of Defence (From 15.1.1982 to 29.1.1983)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Defence (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Department of Personnel and Administrative Reforms and Culture (From 31.12.1984 to 15.3.1985)
		Minister of State in the Ministry of Personnel and Training, Administrative Reforms, Public Grievances and Pensions and Department of Culture (From 15.3.1985 to 25.9.1985)
		Minister of State (Independent Charge) of the Ministry of Food and Civil Supplies (From 25.9.1985 to 27.1.1986)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 18.1.1993 to 15.9.1995)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 15.9.1995 to 16.5.1996*)
50.	Singh, Dr. Sanjay	Minister of State (Independent Charge) of the Ministry of Communications (From 21.11.1990 to 21.6.1991*)
51.	Singh, Shri Chandra Shekhar	Minister of State in the Ministry of Energy—Department of Power (From 29.1.1983 to 14.8.1983)
		Minister of State (Independent Charge) of the Ministry of Supply and Textiles (From 30.3.1985 to 25.9.1985)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Petroleum and Natural Gas (From 20.1.1986 to 9.7.1986)
52.	Solanki, Shri Bharatsinh	Minister of State in the Ministry of Power (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Railways (From 19.1.2011 to 28.10.2012)
		Minister of State (Independent Charge) of the Ministry of Drinking Water and Sanitation (From 28.10.2012 to 26.05.2014*)
53.	Sukh Ram, Shri	Minister of State in the Ministry of Defence—Department of Defence Production and Defence Supplies (From 25.9.1985 to 22.10.1986)
		Minister of State in the Ministry of Planning (From 22.10.1986 to 14.2.1988)
		Minister of State (Independent Charge) of the Ministry of Food and Civil Supplies (From 14.2.1988 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Planning and Programme Implementation and also Minister of State in the Ministry of Non-Conventional Energy Sources (From 2.7.1992 to 18.1.1993)
		Minister of State (Independent Charge) of the Ministry of Communications (From 18.1.1993 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
54.	Tewari, Shri Manish	Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 28.10.2012 to 26.05.2014*)
55.	Thomas, Prof. K.V.	Minister of State in the Ministry of Agriculture and Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 28.5.2009 to 19.1.2011)
		Minister of State (Independent Charge) of the Ministry of Consumer Affairs, Food and Public Distribution (From 19.1.2011 to 26.05.2014*)
56.	Tirath, Shrimati Krishna	Minister of State (Independent Charge) of the Ministry of Women and Child Development (From 28.5.2009 to 26.05.2014*)
57.	Tripathy, Shri Braja Kishore	Minister of State (Independent Charge) of the Ministry of Steel (From 27.5.2000 to 22.5.2004*)
58.	Tytler, Shri Jagdish	Minister of State in the Ministry of Transport—Department of Civil Aviation (From 25.9.1985 to 22.10.1986)
		Minister of State (Independent Charge) of the Ministry of Civil Aviation (From 22.10.1986 to 28.7.1987)
		Minister of State (Independent Charge) of the Ministry of Tourism and Civil Aviation (From 25.7.1987 to 14.2.1988)
		Minister of State (Independent Charge) of the Ministry of Labour (From 14.2.1988 to 25.6.1988)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 25.6.1988 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Surface Transport (From 21.6.1991 to 15.9.1995)
		Minister of State (Independent Charge) of the Ministry of Coal (From 15.9.1995 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State (Independent Charge) of the Ministry of Non-Resident Indian Affairs (From 23.5.2004 to 3.9.2004)
		Minister of State (Independent Charge) of the Ministry of Overseas Indian Affairs (From 3.9.2004 to 10.8.2005)
59.	Veerendra Kumar, Shri M.P.	Minister of State in the Ministry of Finance (From 21.2.1997 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State (Independent Charge) of the Ministry of Labour (From 9.6.1997 to 19.3.1998*)
		Held charge of the office of the Minister of State in the Ministry of Urban Affairs and Employment and the Minister of State in the Ministry of Parliamentary Affairs (From 26.5.1997 to 2.7.1997)
60.	Venkateswarlu, Dr. U.	Minister of State in the Ministries of Agriculture and Parliamentary Affairs (From 1.6.1996 to 29.6.1996)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Urban Affairs and Employment and in the Ministry of Parliamentary Affairs (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State (Independent Charge) of the Ministry of Urban Affairs and Employment (From 9.6.1997 to 19.3.1998*)
61.	Yadav, Shri Balram Singh	Minister of State (Independent Charge) of the Ministry of Mines (From 21.6.1991 to 15.9.1995)
		Minister of State (Independent Charge) of the Ministry of Planning and Programme Implementation (From 15.9.1995 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

PART VI

MINISTERS OF STATE ATTACHED TO PRIME MINISTER AND
CABINET MINISTERS

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Abdullah, Shri Omar	Minister of State in the Ministry of Commerce and Industry (From 13.10.1999 to 22.7.2001)
		Minister of State in the Ministry of External Affairs (From 22.7.2001 to 23.12.2002, the date of his resignation)
2.	Adhikari, Shri Sisir	Minister of State in the Ministry of Rural Development (from 28.5.2009 to 22.9.2012)
3.	Adsul, Shri Anandrao Vithoba	Minister of State in the Ministry of Finance and Company Affairs (From 26.8.2002 to 22.5.2004*)
4.	Aggarwal, Shri Satish	Minister of State in the Ministry of Finance (From 14.8.1977 to 28.7.1979*)
5.	Ahamed, Shri E.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of External Affairs (From 23.5.2004 to 22.5.2009*)
		Minister of State in the Ministry of Railways (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of External Affairs (From 19.1.2011 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Human Resource Development (From 12.7.2011 to 28.10.2012)
6.	Ahluwalia, Shri S.S.	Minister of State in the Ministry of Urban Affairs and Employment— Department of Urban Employment and Poverty Alleviation (From 15.9.1995 to 8.3.1996*)
		Also Minister of State in the Ministry of Parliamentary Affairs (From 19.9.1995 to 16.5.1996*)
		Minister of State in the Ministry of Urban Affairs and Employment (From 8.3.1996 to 16.5.1996*)
7.	Ahmad, Dr. (Shri) Shakeel	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Communications and Information Technology (From 23.5.2004 to 6.4.2008)
		Minister of State in the Ministry of Home Affairs (From 6.4.2008 to 22.5.2009*)
8.	Ahmed, Dr. Abrar	Minister of State in the Ministries of Finance and Parliamentary Affairs (From 18.1.1993 to 2.4.1994)
9.	Ahmed, Shri Kamaluddin	Minister of State in the Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 21.6.1991 to 29.9.1994)
		Also Minister of State in the Ministry of Commerce (From 19.2.1993 to 29.9.1994)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
10.	Ahmed, Shri Sultan	Minister of State in the Ministry of Tourism (From 28.5.2009 to 22.9.2012)
11.	Alageshan, Shri O.V.	Deputy Minister in the Ministry of Railways and Transport (From 12.8.1952 to 16.4.1957)
		Minister of State in the Ministry of Irrigation and Power (From 8.5.1962 to 19.7.1963)
		Minister of Mines and Fuel (As one of the Ministers of State) (From 19.7.1963 to 21.11.1963)
		Minister of State in the Ministry of Petroleum and Chemicals (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Petroleum and Chemicals (As one of the Ministers of State) (From 24.1.1966 to 13.3.1967*)
12.	Alva, Smt. Margaret	Minister of State in the Ministry of Parliamentary Affairs (From 13.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Human Resource Development— Department of Women and Child Development and Youth Affairs and Sports (From 25.9.1985 to 2.12.1989*)
		Minister of State in the Ministry of Personnel, Public Grievances and Pensions (From 21.6.1991 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of State in the Ministry of Parliamentary Affairs (From 19.1.1993 to 16.5.1996*)
13.	Ambareesh, Shri M.H.	Minister of State in the Ministry of Information and Broadcasting (From 24.10.2006 to 22.5.2009*)
14.	Apang, Shri Omar	Minister of State in the Ministry of Tourism (From 20.3.1998 to 13.10.1999*)
15.	Arun Singh, Shri	Parliamentary Secretary to the Prime Minister (From 1.1.1985 to 25.9.1985)
		Minister of State in the Ministry of Defence-Defence Research and Development Organisation (From 25.9.1985 to 18.7.1987)
16.	Austin, Dr. Henry	Minister of State in the Ministry of Commerce and Civil Supplies (From 30.7.1979 to 14.1.1980*)
17.	Badal, Shri Sukhbir Singh	Minister of State in the Ministry of Industry (From 20.3.1998 to 13.10.1999*)
18.	Bagrodia, Shri Santosh	Minister of State in the Ministry of Coal (From 6.4.2008 to 22.5.2009*)
19.	Baitha, Shri D.L.	Deputy Minister in the Ministry of Food and Civil Supplies (From 14.2.1988 to 4.7.1989)
		Minister of State in the Ministry of Defence—Department of Defence Production and Supplies (From 4.7.1989 to 2.12.1989*)
20.	Balasubramanian, Shri S.R.	Minister of State in the Ministry of Personnel, Public Grievances and Pensions and Parliamentary Affairs (From 29.6.1996 to 21.4.1997*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		(Again from 1.5.1997 to 9.6.1997) (Again from 9.6.1997 to 19.3.1998*)
21.	Baleshwar Ram, Shri	Minister of State in the Ministry of Agriculture and Rural Reconstruction— Department of Rural Development (From 16.1.1982 to 29.1.1983)
22.	Bandyopadhayay, Shri Sudip	Minister of State in the Ministry of Health and Family Welfare (From 12.7.2011 to 22.9.2012)
23.	Bartakaki, Smt. Renuka Devi	Minister of State in the Ministry of Education, Social Welfare and Culture (From 16.8.1977 to 28.7.1979*)
24.	Barua, Shri Bedabrata	Deputy Minister in the Department of Company Affairs (From 2.5.1971 to 5.2.1973)
		Deputy Minister in the Ministry of Law, Justice and Company Affairs (From 5.2.1973 to 24.3.1977*)
		Minister of State in the Ministry of External Affairs (From 4.8.1979 to 14.1.1980*)
25.	Basavarajeshwari, Smt.	Minister of State in the Ministry of Human Resource Development— Department of Women and Child Welfare (From 19.1.1993 to 15.9.1995)
26.	Beg, Shri Arif	Minister of State in the Ministry of Commerce, Civil Supplies and Cooperation (From 14.8.1977 to 28.7.1979*)
27.	Behra, Shri Bhajman	Minister of State in the Ministry of Petroleum and Chemicals (From 23.4.1990 to 10.11.1990*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
28.	Bhakt Darshan, Shri	Deputy Minister in the Ministry of Education (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)
		Deputy Minister in the Ministry of Transport and Shipping (From 18.3.1967 to 18.2.1969)
		Minister of State in the Ministry of Education and Youth Services (From 18.2.1969 to 18.3.1971*)
29.	Bhatia, Shri R.L.	Minister of State in the Ministry of External Affairs (From 2.7.1992 to 16.5.1996*)
30.	Bhoi, Dr. Krupasindhu	Minister of State in the Ministry of Human Resource Development— Department of Education (From 15.9.1995 to 16.5.1996*)
31.	Bir Sen, Shri	Minister of State in the Ministry of Environment and Forests (From 31.12.1984 to 25.9.1985)
32.	Chakravarty, Smt. Bijoya	Minister of State in the Ministry of Water Resources (From 13.10.1999 to 22.5.2004*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
33.	Chanana, Dr. Charanjit	Minister of State in the Ministry of Industry (From 3.3.1980 to 15.1.1982)
		Minister of State in the Ministry of Industry and Steel and Mines (From 15.1.1982 to 2.9.1982)
34.	Chand Ram, Shri	Minister of State in the Ministry of Shipping and Transport (From 14.8.1977 to 28.7.1979*)
35.	Chandrakar, Shri Chandu Lal	Minister of State in the Ministry of Tourism and Civil Aviation (From 8.6.1980 to 15.1.1982)
		Minister of State in the Ministry of Agriculture and Rural Development (From 1.1.1985 to 27.1.1986)
36.	Chandrasekhar, Dr. S.	Minister of State in the Ministry of Health and Family Planning (From 13.3.1967 to 14.11.1967)
		Minister of State in the Ministry of Health and Family Planning and Urban Development (From 14.11.1967 to 14.2.1969)
		Minister of State in the Ministry of Health and Family Planning, Works, Housing and Urban Development (From 14.2.1969 to 26.6.1970)
37.	Chaturvedi, Shri Bhuvnesh	Minister of State in the Prime Minister's Office (From 18.1.1993 to 16.5.1996*)
		Also assisted Prime Minister as the Minister of State in the Department of Atomic Energy and Space (From 16.2.1993 to 16.5.1996*)
		Also Minister of State in the Ministry of Science and Technology (From 2.12.1993 to 16.5.1996*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
38.	Chaudhary, Shri Tusharbhai	Minister of State in the Ministry of Tribal Affairs (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Road Transport and Highways (From 19.1.2011 to 26.05.2014*)
39.	Chaudhury, Dalbir Singh	Deputy Minister in the Ministry of Petroleum and Chemicals (From 2.5.1971 to 9.11.1973)
		Deputy Minister in the Ministry of Heavy Industry (From 9.11.1973 to 10.10.1974)
		Deputy Minister in the Ministry of Works and Housing (From 10.10.1974 to 1.12.1975)
		Deputy Minister in the Ministry of Shipping and Transport (From 1.12.1975 to 24.3.1977*)
		Minister of State in the Ministry of Petroleum and Chemicals (From 8.6.1980 to 17.10.1980)
		Minister of State in the Ministry of Petroleum, Chemicals and Fertilisers (From 17.10.1980 to 2.9.1982) (Again from 2.9.1982 to 29.1.1983)
		Minister of State in the Ministry of Energy—Department of Petroleum (From 2.9.1982 to 29.1.1983)
		Minister of State in the Ministry of Energy—Department of Coal (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984)
40.	Chaudhury, Smt. Rashida Haque	Minister of State in the Ministry of Education, Culture and Social Welfare (From 30.7.1979 to 24.8.1979)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Education and Culture (From 24.8.1979 to 14.1.1980*)
41.	Chauhan, Shri Sriram	Minister of State in the Ministry of Parliamentary Affairs (From 13.10.1999 to 22.11.1999)
		Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 22.11.1999 to 17.7.2000)
		Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 17.7.2000 to 1.9.2001)
42.	Chavan, Shri D.R.	Deputy Minister in the Ministry of Defence (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Food and Agriculture (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Planning (From 24.1.1966 to 14.2.1966)
		Deputy Minister in the Ministry of Labour, Employment and Rehabilitation (From 14.2.1966 to 13.3.1967*)
		Deputy Minister in the Ministry of Law (From 18.3.1967 to 14.11.1967)
		Deputy Minister in the Ministry of Labour, Employment and Rehabilitation (From 14.11.1967 to 14.2.1969)
		Minister of State in the Ministry of Petroleum and Chemicals, Mines and Metals (From 14.2.1969 to 18.3.1971*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Petroleum, Chemicals and Non-ferrous Metals (From 18.3.1971 to 2.5.1971)
		Minister of Supply (as one of the Ministers of State) (From 2.5.1971 to 5.2.1973)
		Minister of State in the Ministry of Law, Justice and Company Affairs (From 5.2.1973 to 8.7.1973)
43.	Chikhalia, Smt. Bhavanaben Devrajbhai	Minister of State in the Ministry of Parliamentary Affairs and in the Ministry of Tourism and Culture (From 29.1.2003 to 22.5.2004*)
44.	Choudhary, Shri Nikhil Kumar	Minister of State in the Ministry of Agro and Rural Industries (From 1.7.2002 to 29.1.2003)
45.	Choudhury, Shri A.H. Khan	Minister of State in the Ministry of Health and Family Welfare (From 28.10.2012 to 26.05.2014*)
46.	Chowdhary, Smt. Santosh	Minister of State in the Ministry of Health and Family Welfare (From 17.6.2013 to 26.05.2014*)
47.	Chowdhury, Shri Adhir Ranjan	Minister of State in the Ministry of Railways (From 28.10.2012 to 26.05.2014*)
48.	Dalbir Singh, Shri	Minister of State in the Ministry of Urban Development (From 25.9.1985 to 2.12.1989*)
		Minister of State in the Ministry of Finance (From 21.6.1991 to 17.1.1993)
49.	Dar, Shri Mohammad Maqbool	Minister of State in the Ministry of Home Affairs (From 10.7.1996 to 21.4.1997*) (Again from 1.5.1997 to 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
50.	Das, Shri Akhilesh	Minister of State in the Ministry of Steel (From 29.1.2006 to 6.4.2008, the date of his resignation)
51.	Das, Shri Bhakta Charan	Deputy Minister in the Ministry of Human Resource Development— Department of Youth Affairs and Sports (From 23.4.1990 to 5.11.1990)
		Minister of State in the Ministry of Railways (From 21.11.1990 to 21.6.1991*)
52.	Dasmunsi, Smt. Deepa	Minister of State in the Ministry of Urban Development (From 28.10.2012 to 26.05.2014*)
53.	Datar, Shri B.N.	Deputy Minister in the Ministry of Home Affairs (From 12.8.1952 to 14.2.1956)
		**Minister in the Ministry of Home Affairs (From 14.2.1956 to 17.4.1957*)
		Minister of State in the Ministry of Home Affairs (From 17.4.1957 to 10.4.1962*) (Again from 16.4.1962 to 13.2.1963)
54.	Deora, Shri Milind	Minister of State in the Ministry of Communications and Information Technology (From 12.7.2011 to 26.05.2014*)
		Minister of State in the Ministry of Shipping (From 30.10.2012 to 26.05.2014*)
55.	Dhakne, Shri Baban Rao	Minister of State in the Ministry of Energy (From 21.11.1990 to 21.6.1991*)
56.	Dhananjaya Kumar, Shri V.	Minister of Civil Aviation and Tourism (From 16.5.1996 to 1.6.1996*)

^{*}The date of dissolution of the Council of Ministers.
**Minister of Cabinet rank but not member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Finance (From 13.10.1999 to 30.9.2000)
		Minister of State in the Ministry of Textiles (From 30.9.2000 to 1.7.2002, the date of his resignation)
57.	Dhanushkodi Athithan, Shri R.	Minister of State in the Ministry of Human Resource Development— Department of Youth Affairs and Sports (From 29.6.1996 to 21.4.1997*) (Again from 1.5.1997 to 9.6.1997) (Again from 9.6.1997 to 19.3.1998)
58.	Dharam Vir, Shri	Deputy Minister in the Ministry of Labour (From 15.1.1982 to 6.9.1982)
		Deputy Minister in the Ministry of Labour and Rehabilitation (From 6.9.1982 to 29.1.1983)
		Minister of State in the Ministry of Labour and Rehabilitation (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 22.12.1984)
59.	Digvijay Singh, Shri	Deputy Minister in the Department of Ecology (From 2.9.1982 to 31.10.1984*)
		Deputy Minister in the Department of Environment (From 4.11.1984 to 31.12.1984*)
		Deputy Minister in the Ministry of Finance (From 21.11.1990 to 21.6.1991*)
		Deputy Minister in the Ministry of External Affairs (From 28.11.1990 to 21.6.1991*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Railways (From 13.10.1999 to 22.7.2001) (Again from 1.8.2001 to 1.7.2002)
		Minister of State in the Ministry of Commerce and Industry (From 22.7.2001 to 1.9.2001)
		Minister of State in the Ministry of External Affairs (From 1.7.2002 to 22.5.2004*)
60.	Dikshit, Smt. Sheila	Minister of State in the Ministry of Parliamentary Affairs (From 12.5.1986 to 2.12.1989*)
		Also Minister of State in the Prime Minister's Office (From 25.6.1988 to 2.12.1989*)
61.	Diwakar, Shri R.R.	Minister of State in the Ministry of Information and Broadcasting (From 7.10.1948 to 26.1.1950*) (Again from 31.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1952*)
62.	Elangovan, Shri E.V.K.S.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 23.5.2004 to 25.5.2004)
		Minister of State in the Ministry of Commerce and Industry (From 25.5.2004 to 29.1.2006)
		Minister of State in the Ministry of Textiles (From 29.1.2006 to 22.5.2009*)
63.	Engti, Shri Biren Singh	Deputy Minister in the Department of Personnel and Training (From 12.5.1986 to 4.6.1986)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Personnel, Public Grievances and Pensions (From 4.6.1986 to 14.2.1988)
		Minister of State in the Ministry of Planning and Programme Implementation (From 14.2.1988 to 2.12.1989*)
64.	Ering, Shri Ninong	Minister of State in the Ministry of Minority Affairs (From 28.10.2012 to 26.05.2014*)
65.	Faleiro, Shri Eduardo	Minister of State in the Ministry of External Affairs (From 12.5.1986 to 14.2.1988)
		Minister of State in the Ministry of Finance — Department of Economic Affairs (From 14.2.1988 to 2.12.1989*)
		Minister of State in the Ministry of External Affairs (From 21.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Chemicals and Fertilizers (From 18.1.1993 to 16.5.1996*)
		Also Minister of State in the Ministry of Parliamentary Affairs (From 18.12.1993 to 19.9.1995)
		Also Minister of State in the Department of Electronics and Ocean Development (From 18.12.1993 to 16.5.1996*)
66.	Farook, Shri M.O.H.	Minister of State in the Ministry of Civil Aviation and Tourism (From 21.6.1991 to 2.7.1992)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Civil Aviation—Department of Civil Aviation (From 2.7.1992 to 17.1.1993)
67.	Fatmi, Shri Md. Ali Ashraf	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Human Resource Development (From 23.5.2004 to 22.5.2009*)
68.	Gadhvi, Shri B.K.	Minister of State in the Ministry of Finance—Department of Expenditure (From 12.5.1986 to 2.12.1989)
69.	Gaikwad Patil, Shri Jaysingrao	Minister of State in the Ministry of Human Resource Development (From 13.10.1999 to 27.5.2000)
		Minister of State in the Ministry of Mines (From 27.5.2000 to 1.9.2001, the date of his resignation)
70.	Gandhi, Shri Dilip Kumar Mansukhlal	Minister of State in the Ministry of Shipping (From 29.1.2003 to 15.3.2004)
71.	Gandhiselvan, Shri S.	Minister of State in the Ministry of Health and Family Welfare (From 28.5.2009 to 20.3.2013)
72.	Ganesh, Shri K.R.	Deputy Minister in the Ministry of Finance (From 26.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Finance (From 2.5.1971 to 10.10.1974)
		Minister of State in the Ministry of Petroleum and Chemicals (From 10.10.1974 to 1.12.1975)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
73.	Gangwar, Shri Santosh Kumar	Minister of State in the Ministry of Petroleum and Natural Gas (From 20.3.1998 to 13.10.1999*)
		Held charge of the Ministry of Parliamentary Affairs as a Minister of State (From 16.2.1999 to 13.10.1999*)
		Minister of State in the Ministry of Science and Technology (From 13.10.1999 to 22.11.1999)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 22.11.1999 to 24.5.2003)
		Minister of State in the Ministry of Labour and in the Ministry of Parliamentary Affairs (From 24.5.2003 to 8.9.2003)
		Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 8.9.2003 to 22.5.2004*)
74.	Gavit, Shri Manikrao	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Home Affairs (From 23.5.2004 to 6.4.2008, the date of his resignation)
		Minister of State in the Ministry of Social Justice and Empowerment (From 17.6.2013 to 26.05.2014*)
75.	George, Shri A.C.	Deputy Minister in the Ministry of Foreign Trade (From 2.5.1971 to 5.2.1973)
		Minister of State in the Ministry of Industry and Civil Supplies (From 10.10.1974 to 9.8.1976)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Civil Supplies and Cooperation (From 9.8.1976 to 24.3.1977*)
76.	Ghose, Dr. Sanker	Minister of State in the Ministry of Planning (From 21.4.1976 to 24.3.1977*)
77.	Ghosh, Shri Parimal	Minister of State in the Ministry of Railways (From 13.3.1967 to 17.10.1969)
		Minister of State in the Ministries of Health and Family Planning and Works, Housing and Urban Development (From 26.6.1970 to 18.3.1971)
78.	Gobardhan, Shri Bhagey	Minister of State in the Ministry of Planning and Programme Implementation (From 23.4.1990 to 10.11.1990*)
		Minister of State in the Ministry of Human Resource Development (From 21.11.1990 to 20.2.1991**)
79.	Goel, Shri Vijay	Minister of State in the Prime Minister's Office and Minister of State in the Ministry of Planning
		(From 1.9.2001 to 2.11.2001)
		Minister of State in the Prime Minister's Office (From 2.11.2001 to 1.7.2002)
		Minister of State in the Prime Minister's Office and in the Ministry of Statistics and Programme Implementation (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Parliamentary Affairs and in the Ministry of Labour (From 29.1.2003 to 24.5.2003)

^{*} The date of dissolution of the Council of Ministers.

** Resigned w.e.f. 20.2.1991 following disqualification from membership of Lok Sabha under Anti-Defection Law (Tenth Schedule to the Constitution).

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Youth Affairs and Sports (From 24.5.2003 to 22.5.2004*)
80.	Gopal, Shri K.	Minister of State in the Ministry of Parliamentary Affairs (From 4.8.1979 to 14.1.1980*)
81.	Goyal, Shri Krishna Kumar	Minister of State in the Ministry of Commerce, Civil Supplies and Cooperation (From 14.8.1977 to 28.7.1979*)
82.	Guha, Smt. Phulrenu	Minister of State in the Department of Social Welfare (From 13.3.1967 to 14.2.1969)
		Minister of State in the Ministry of Law and Department of Social Welfare (From 14.2.1969 to 26.6.1970)
83.	Gulshan, Shri Dhanna Singh	Minister of State in the Ministry of Education, Social Welfare and Culture (From 14.8.1977 to 28.7.1979*)
84.	Gupta, Prof. Chaman Lal	Minister of State in the Ministry of Civil Aviation (From 13.10.1999 to 1.9.2001)
		Minister of State (Independent Charge) of the Ministry of Food Processing Industries (From 1.9.2001 to 1.7.2002)
		Minister of State in the Ministry of Defence (From 1.7.2002 to 22.5.2004*)
85.	Hajarnavis, Shri R.M.	Deputy Minister in the Ministry of Law (From 4.1.1958 to 10.4.1962) (Again from 16.4.1962 to 8.5.1965)
		Deputy Minister in the Ministry of Mines and Fuel (From 3.5.1962 to 22.2.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Home Affairs (From 22.2.1963 to 10.3.1964)
		Minister of Supply (As one of the Ministers of State) (From 10.3.1964 to 27.5.1964*)
		Held Charge of the Department of Technical Development as a Minister of State (From 10.3.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 13.6.1964)
		Minister of Cultural Affairs in the Ministry of Education as a Minister of State (From 13.6.1964 to 29.10.1965)
		Minister of State in the Ministry of Law and also in the Department of Social Security (From 29.10.1965 to 11.1.1966*)
86.	Hussain, Shri Sarwar	Minister of State in the Ministry of Food and Civil Supplies (From 21.11.1990 to 20.2.1991**)
87.	Jacob, Shri M.M.	Minister of State in the Ministry of Parliamentary Affairs (From 22.10.1986 to 4.7.1989)
		Minister of State (Independent Charge) of the Ministry of Water Resources and Minister of State in the Ministry of Parliamentary Affairs (From 4.7.1989 to 2.12.1989*)
		Minister of State in the Ministries of Home and Parliamentary Affairs (From 21.6.1991 to 17.1.1993)

^{*} The date of dissolution of the Council of Ministers.

** Resigned w.e.f. 20.2.1991 following disqualification from membership of Lok Sabha under Anti-Defection Law (Tenth Schedule to the Constitution).

Sl.No.	Name	Portfolios alongwith the period thereof
88.	Jagadeesan, Smt. Subbulakshmi	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Social Justice and Empowerment (From 23.5.2004 to 22.5.2009*)
89.	Jagathrakshakan, Dr. S.	Minister of State in the Ministry of Information and Broadcasting (From 28.5.2009 to 28.10.2012)
		Minister of State in the Ministry of New and Renewable Energy (From 28.10.2012 to 2.11.2012)
		Minister of State in the Ministry of Commerce and Industry (From 2.11.2012 to 20.3.2013)
90.	Jagbir Singh, Shri	Minister of State in the Ministry of Information and Broadcasting (From 14.8.1977 to 11.7.1978)
		Minister of State in the Ministry of Information and Broadcasting (From 26.1.1979 to 15.7.1979)
		Minister of State in the Ministry of Defence (From 30.7.1979 to 14.1.1980*)
91.	Jaichandra Singh, R.K.	Minister of State in the Department of Youth Affairs and Sports (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Industry (Department of Chemicals and Petrochemicals) (From 25.9.1985 to 18.2.1988)
92.	Jain, Shri Pradeep	Minister of State in the Ministry of Rural Development (From 28.5.2009 to 26.05.2014)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
93.	Janarthanan Kadambur, Shri R.	Minister of State in the Ministry of Personnel, Public Grievances and Pensions (From 20.3.1998 to 8.4.1999)
		Also Minister of State in the Ministry of Finance (From 22.5.1998 to 8.4.1999)
94.	Jatua, Shri Choudhury Mohan	Minister of State in the Ministry of Information and Broadcasting (From 28.5.2009 to 22.9.2012)
95.	Jha, Shri Raghunath	Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 6.4.2008 to 22.5.2009*)
96.	Judev, Shri Dilip Singh	Minister of State in the Ministry of Environment and Forests (From 29.1.2003 to 17.11.2003, the date of his resignation)
97.	Kalmadi, Shri Suresh	Minister of State in the Ministry of Railways (From 15.9.1995 to 16.5.1996*)
98.	Kamson, Prof. Meijinlung	Minister of State in the Ministry of Home Affairs (From 15.9.1995 to 16.5.1996*)
99.	Kanti Singh, Smt.	Minister of State in the Ministry of Human Resource Development (From 1.6.1996 to 29.6.1996)
		Minister of State (Independent Charge) of the Ministry of Coal (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 10.1.1998, the date of her resignation)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Human Resource Development (From 23.5.2004 to 29.1.2006)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Department of Heavy Industries in the Ministry of Heavy Industries and Public Enterprises (From 29.1.2006 to 6.4.2008)
		Minister of State in the Ministries of Tourism and Culture (From 6.4.2008 to 22.5.2009*)
100.	Kanungo, Shri Nityanand	Deputy Minister in the Ministry of Commerce and Industry (From 28.9.1954 to 10.8.1955)
		Minister of Consumer Industries (As one of the Ministers of State) (From 10.8.1955 to 13.6.1956) (Again from 13.6.1956 to 17.4.1957*)
		Commerce and Industry (From 17.4.1957 to 25.4.1957)
		Minister of Commerce (As one of the Ministers of State) (From 25.4.1957 to 10.4.1962*)
		Minister of State in the Ministry of Commerce and Industry (From 10.4.1962 to 19.7.1963)
		Minister of Industry (As one of the Ministers of State) (From 19.7.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of Cultural Affairs (As one of the Ministers of State) (From 9.6.1964 to 13.6.1964)
		Minister of Civil Aviation (As one of the Ministers of State) (From 13.6.1964 to 31.7.1965)

^{*}The date of dissolution of the Council of Ministers. ${}^{\tiny{\textcircled{\tiny{0}}}\tiny{\textcircled{\tiny{0}}}}\text{Minister}$ of Cabinet rank but not Member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
101.	Karmakar, Shri D.P.	Deputy Minister in the Ministry of Commerce (From 14.8.1950 to 13.5.1952*)
		Deputy Minister of Commerce and Industry (From 13.5.1952 to 12.8.1952)
		©©Minister of Commerce and Industry (From 12.8.1952 to 21.8.1952)
		©©Minister of Commerce (From 21.8.1952 to 13.6.1956)
		Minister of Trade (As Minister of State) (From 13.6.1956 to 17.4.1957*)
		Minister of Health (As Minister of State) (From 17.4.1957 to 9.4.1962*)
102.	Kataria, Shri Lalchand	Minister of State in the Ministry of Defence (From 28.10.2012 to 30.10.2012)
		Minister of State in the Ministry of Rural Development (From 30.10.2012 to 26.05.2014*)
103.	Kathiria, Dr. Vallabhbhai	Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 13.10.1999 to 29.1.2003)
		Minister of State in the Ministry of Human Resource Development (From 30.1.2003 to 9.1.2004)
		Minister of State in the Ministry of Health and Family Welfare (From 29.1.2003 to 30.1.2003*) (Again from 9.1.2004 to 22.5.2004*)
104.	Kaur, Smt. Preneet	Minister of State in the Ministry of External Affairs (From 28.5.2009 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers. ${}^{\tiny{\bigcirc}\bigcirc}$ Minister of Cabinet rank but not Member of the Cabinet.

Sl.No.	Name	Portfolios alongwith the period thereof
105.	Keskar, Dr. B.V.	Deputy Minister in the Ministry of External Affairs and Commonwealth Relations (From 7.12.1948 to 26.1.1950*)
		Deputy Minister of External Affairs (From 31.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 13.5.1952*)
		Also worked as Deputy Minister in the Ministries of Railways and Transport (From 10.3.1952 to 13.5.1952*)
		Minister of State in the Ministry of Information and Broadcasting (From 13.5.1952 to 17.4.1957*)
		Minister of Information and Broadcasting (As Minister of State) (From 17.4.1957 to 9.4.1962*)
106. K	Khadilkar, Shri R.K.	Minister of State in the charge of the Ministry of Supply (From 4.11.1969 to 18.3.1971)
		Also Minister of State in the Ministry of Finance (From 7.11.1969 to 27.6.1970)
		Minister of State in the Ministry of Labour, Employment and Rehabilitation (From 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Labour and Rehabilitation (From 2.5.1971 to 5.2.1973)
		Minister of State in the Ministry of Health and Family Planning (From 5.2.1973 to 9.11.1973)
		Minister of State in the Ministry of Supply and Rehabilitation (From 9.11.1973 to 1.12.1975)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
107.	Khan, Shri Aslam Sher	Minister of State in the Prime Minister's Office and Minister of State in the Ministry of Welfare (From 15.9.1995 to 16.5.1996*)
108.	Khan, Shri Mohd. Ayub	Minister of State in the Ministry of Agriculture (From 15.9.1995 to 16.5.1996*)
109.	Khan, Shri Shah Nawaz	Deputy Minister in the Ministry of Railways and Transport (From 20.9.1956 to 17.4.1957*)
		Deputy Minister in the Ministry of Railways (From 17.4.1957 to 10.4.1962*)
		Deputy Minister in the Ministry of Railways (From 16.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Food and Agriculture (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Labour, Employment and Rehabilitation (From 24.1.1966 to 13.3.1967*)
		Minister of State in the Ministry of Steel and Mines (From 2.5.1971 to 5.2.1973)
		Minister of State in the Ministry of Supply (From 5.2.1973 to 9.11.1973)
		Minister of State in the Ministry of Petroleum and Chemicals (From 9.11.1973 to 10.10.1974)
		Minister of State in the Ministry of Agriculture and Irrigation (From 10.10.1974 to 24.3.1977*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
110.	Khandela, Shri Mahadev S.	Minister of State in the Ministry of Road Transport and Highways (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Tribal Affairs (From 19.1.2011 to 27.10.2012)
111.	Khanna, Shri Vinod	Minister of State in the Ministry of Tourism and Culture (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of External Affairs (From 29.1.2003 to 22.5.2004*)
112.	Khaparde, Ms. Saroj	Minister of State in the Ministry of Health and Family Welfare (From 12.5.1986 to 4.7.1989)
		Minister of State in the Ministry of Textiles (From 4.7.1989 to 2.12.1989*)
113.	Killi, Dr. (Smt.) Kruparani	Minister of State in the Ministry of Communications and Information Technology (From 28.10.2012 to 26.05.2014*)
114.	Krishna Kumar, Shri S.	Deputy Minister in the Ministry of Health and Family Welfare (From 25.9.1985 to 22.10.1986)
		Deputy Minister in the Ministry of Textiles (From 22.10.1986 to 14.2.1988)
		Deputy Minister in the Ministry of Information and Broadcasting (From 14.2.1988 to 25.6.1988)
		Minister of State in the Ministry of Information and Broadcasting (From 25.6.1988 to 2.12.1989*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Petroleum and Natural Gas (From 21.6.1991 to 18.1.1993)
		Also Minister of State in the Ministry of Defence (From 26.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Non- Conventional Energy Sources (From 18.1.1993 to 13.9.1995)
		Also Minister of State in the Ministry of Agriculture (From 19.2.1993 to 13.9.1995)
115.	Krishnan, Shri G.Y.	Minister of State in the Ministry of Civil Aviation and Tourism—Department of Civil Aviation (From 15.9.1995 to 16.5.1996*)
116.	Krishnappa, Shri M.V.	Deputy Minister in the Ministry of Food and Agriculture (From 12.8.1952 to 17.4.1957*) (Again from 17.4.1957 to 25.4.1957)
		Deputy Minister in the Ministry of Food (From 25.4.1957 to 5.7.1957)
		Deputy Minister in the Ministry of Agriculture (From 5.7.1957 to 14.3.1962)
		Minister of State in the Ministry of Agriculture and Irrigation (From 4.8.1979 to 14.1.1980*)
117.	Kulaste, Shri Faggan Singh	Minister of State in the Ministry of Parliamentary Affairs (From 13.10.1999 to 22.11.1999)
		Minister of State in the Ministry of Tribal Affairs (From 22.11.1999 to 22.5.2004*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
118.	Kundu, Shri Samarendra	Minister of State in the Ministry of External Affairs (From 14.8.1977 to 28.7.1979*)
119.	Kurien, Prof. P.J.	Minister of State in the Ministry of Industry (From 21.6.1991 to 2.7.1992)
		Minister of State in the Ministry of Industry—Department of Small Scale, Agro and Rural Industries (From 2.7.1992 to 17.1.1993)
		Also Minister of State in the Ministry of Commerce (From 10.7.1992 to 17.1.1993)
		Minister of State in the Ministry of Non- Conventional Energy Sources (From 15.9.1995 to 16.5.1996*)
120.	Lakshmi, Smt. Panabaka	Minister of State in the Ministry of Health and Family Welfare (From 23.5.2004 to 29.3.2009)
		Minister of State (Independent charge) of the Ministry of Health and Family Welfare (From 29.3.2009 to 22.5.2009)
		Minister of State in the Ministry of Textiles (From 28.5.2009 to 30.10.2012) (Again from 5.3.2013 to 26.05.2014*)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 30.10.2012 to 26.05.2014*)
121.	Laskar, Shri Nihar Ranjan	Minister of State in the Ministries of: Health and Family Welfare (From 14.1.1980 to 15.1.1982)
		Home Affairs (From 15.1.1982 to 7.2.1984)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Commerce and Supply (From 7.2.1984 to 7.9.1984)
		Industry (From 7.9.1984 to 31.10.1984*) (Again from 12.11.1984 to 31.12.1984*)
122.	Laxman, Shri Bangaru	Minister of State in the Ministry of Planning (From 13.10.1999 to 22.11.1999)
		Minister of State in the Ministry of Statistics and Programme Implementation (From 13.10.1999 to 22.11.1999)
		Also Minister of State in the Ministry of Railways (From 22.11.1999 to 31.8.2000)
123.	Lenka, Shri K.C.	Minister of State in the Ministry of Agriculture (From 21.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Railways (From 18.1.1993 to 2.4.1994)
124.	Mahajan, Shri Vikram	Minister of State in the Ministry of Energy (From 8.6.1980 to 29.1.1983)
125.	Mahajan, Shrimati Sumitra	Minister of State in the Ministry of Human Resource Development (From 13.10.1999 to 1.7.2002)
		Minister of State in the Ministry of Communications and Information Technology (From 1.7.2002 to 24.5.2003)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 24.5.2003 to 22.5.2004*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
126.	Mahant, Shri Charan Das	Minister of State in the Ministry of Food Processing Industries (From 12.7.2011 to 26.05.2014*)
		Minister of State in the Ministry of Agriculture (From 12.7.2011 to 26.05.2014*)
127.	Maharia, Shri Subhash	Minister of State in the Ministry of Rural Development (From 13.10.1999 to 29.1.2003)
		Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 29.1.2003 to 22.5.2004*)
128.	Mahida, Shri Narendra Singh	Minister of State in the Ministry of Defence (From 26.6.1970 to 18.3.1971*)
	Mahishi, Dr. (Smt.) Sarojini	Deputy Minister attached to the Prime Minister (From 18.3.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Tourism and Civil Aviation (From 14.2.1969 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Tourism and Civil Aviation (From 2.5.1971 to 10.10.1974)
		Minister of State in the Ministry of Law, Justice and Company Affairs (From 10.10.1974 to 3.1.1976)
130.	Makwana, Shri Yogendra	Minister of State in the Ministry of Home Affairs (From 14.1.1980 to 15.1.1982)
		Minister of State in the Ministry of Communications (From 15.1.1982 to 29.1.1983)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Agriculture (From 29.1.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Health and Family Welfare— Department of Health (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Agriculture—Department of Agriculture and Co-operation (From 25.9.1985 to 14.2.1988)
		Minister of State in the Ministry of Steel and Mines (From 14.2.1988 to 2.10.1988)
131.	Malaviya, Shri Radhakrishna	Deputy Minister in the Ministry of Labour (From 14.2.1988 to 4.7.1989)
		Deputy Minister in the Ministry of Parliamentary Affairs (From 25.6.1988 to 4.7.1989)
		Minister of State in the Ministry of Parliamentary Affairs and Labour (From 4.7.1989 to 2.12.1989*)
132.	Malik, Shri Satya Pal	Minister of State in the Ministry of Parliamentary Affairs and Tourism (From 23.4.1990 to 10.11.1990*)
133.	Mallikarjun, Shri K.	Deputy Minister in the Ministry of Railways (From 8.6.1980 to 29.1.1983)
		Deputy Minister in the Ministry of Education and Social Welfare (From 21.11.1981 to 15.1.1982)
		Deputy Minister in the Ministry of Information and Broadcasting and Department of Parliamentary Affairs (From 15.1.1982 to 7.2.1984)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Works and Housing, Sports and Department of Parliamentary Affairs (From 15.1.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Railways (From 21.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Defence (From 18.1.1993 to 19.1.1995)
		Held charge of the Ministry of Parliamentary Affairs as a Minister of State (From 17.4.1994 to 16.5.1996*)
		Minister of State in the Ministry of Railways (From 21.8.1995 to 19.9.1995)
		Minister of State in the Ministry of Defence—Department of Defence (From 19.9.1995 to 16.5.1996*)
		Minister of State in the Ministry of Defence (Department of Defence Research and Development) (From 30.9.1995 to 16.5.1996*)
134.	Malti, Kumari Abha	Minister of State in the Ministry of Industry (From 14.8.1977 to 28.7.1979*)
135.	Mandal, Shri Dhanik Lal	Minister of State in the Ministry of Home Affairs (From 14.8.1977 to 15.7.1979) (Again from 30.7.1979 to 14.1.1980*)
136.	Marandi, Shri Babulal	Minister of State in the Ministry of Environment and Forests (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 7.11.2000)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
137.	Matang Singh, Shri	Minister of State in the Ministry of Parliamentary Affairs (From 10.2.1995 to 16.5.1996*)
138.	Maurya, Shri B.P.	Minister of State in the Ministry of Agriculture (From 11.1.1974 to 10.10.1974)
		Minister of State in the Ministry of Industry and Civil Supplies (From 10.10.1974 to 9.8.1976)
		Minister of State in the Ministry of Industry (From 9.8.1976 to 24.3.1977*)
139.	Meena, Shri Namo Narain	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Environment and Forests (From 23.5.2004 to 22.5.2009*)
		Minister of State in the Ministry of Finance (From 28.5.2009 to 26.05.2014*)
140.	Meena, Smt. Jas Kaur	Minister of State in the Ministry of Human Resource Development (From 29.1.2003 to 22.5.2004*)
141.	Meghwal, Shri Kailash	Minister of State in the Ministry of Social Justice and Empowerment (From 24.5.2003 to 22.5.2004*)
142.	Mehta, Shri Chimanbhai	Minister of State in the Ministry of Human Resource Development (From 23.4.1990 to 10.11.1990*)
143.	Mehta, Shri Om	Minister of State in the Department of Parliamentary Affairs (From 30.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Minister of State in the Department of Parliamentary Affairs and in the Ministry of Shipping and Transport (From 2.5.1971 to 5.2.1973)

 $^{{}^{*}\}text{The date of dissolution of the Council of Ministers.}$

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Department of Parliamentary Affairs and in the Ministry of Works and Housing (From 5.2.1973 to 10.10.1974)
		Minister of State in the Ministry of Home Affairs—Department of Personnel and Administrative Reforms and in the Department of Parliamentary Affairs (From 10.10.1974 to 24.3.1977*)
144.	Mehta, Smt. Jayawanti	Minister of State in the Ministry of Power (From 13.10.1999 to 22.5.2004*)
145.	Menon, Prof. M.G.K.	Minister of State in the Ministry of Science and Technology (From 18.12.1989 to 10.11.1990*)
		Also assisted the Prime Minister in the Departments of Atomic Energy, Electronics, Ocean Development and Space (From 25.12.1989 to 10.11.1990*)
		Also Minister of State in the Ministry of Human Resource Development (Department of Education) (From 20.1.1990 to 23.4.1990)
146.	Menon, Shrimati Lakshmi, N.	Deputy Minister of External Affairs (From 17.4.1957 to 10.4.1962*)
		Minister of State in the Ministry of External Affairs (From 16.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
147.	Mishra, Shri Bibudhendra	Deputy Minister in the Ministry of Law (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Industry (From 15.6.1964 to 20.6.1964)
		Deputy Minister in the Ministry of Industry and Supply (From 20.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of Industry (From 24.1.1966 to 13.3.1967*)
148.	Mishra, Shri Gargi Shankar	Minister of State in the Ministry of Energy—Department of Coal (From 15.1.1982 to 2.9.1982)
		Minister of State in the Ministry of Steel, Mines and Coal (From 2.9.1982 to 6.9.1982)
		Minister of State in the Ministry of Energy (From 6.9.1982 to 29.1.1983)
		Minister of State in the Ministry of Energy—Department of Petroleum (From 29.1.1983 to 31.10.1984) (Again from 4.11.1984 to 31.12.1984*)
149.	Mishra, Shri Harinath	Minister of State (Independent Charge) of the Ministry of Rural Development (From 29.1.1983 to 2.8.1984)
		Minister of State in the Ministry of Irrigation (From 2.8.1984 to 31.10.1984*)
		Minister of State in the Ministry of Planning (From 4.11.1984 to 31.12.1984*)
150.	Mohan, Dr. Chinta	Minister of State in the Ministry of Chemicals and Fertilizers (From 21.6.1991 to 17.1.1993)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
151.	Mookherjee, Shri Satyabrata	Minister of State in the Ministry of Chemicals and Fertilizers (From 30.9.2000 to 1.7.2002)
		Minister of State in the Ministry of Social Justice and Empowerment (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Planning in the Ministry of Statistics and Programme Implementation and in the Departments of Atomic Energy and Space (From 29.1.2003 to 22.5.2004*)
		Minister of State in the Ministry of Commerce and Industry (From 5.6.2003 to 22.5.2004)
152.	Moorthy, Shri A.K.	Minister of State in the Ministry of Railways (From 1.7.2002 to 15.1.2004, the date of his resignation)
153.	Morarka, Shri Kamal	Minister of State in the Prime Minister's Office (From 21.11.1990 to 21.6.1991*)
154.	Muhammad, Dr. V.A. Seyid	Minister of State in the Ministry of Law, Justice and Company Affairs (From 25.12.1975 to 24.3.1977*)
155.	Muni Lall, Shri	Minister of State in the Ministry of Labour (From 13.10.1999 to 1.7.2002 <i>i.e.</i> the date of his resignation)
156.	Murthy, Shri B.S.	Deputy Minister in the Ministry of Community Development and Cooperation (From 6.3.1959 to 10.4.1962*)
		Deputy Minister in the Ministry of Community Development, Panchayati Raj and Cooperation (From 16.4.1962 to 16.2.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Community Development and Cooperation (From 16.2.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Health and Family Planning (From 24.1.1966 to 13.3.1967*)
		Deputy Minister in the Ministry of Health and Family Planning (From 18.3.1967 to 14.11.1967)
		Deputy Minister in the Ministry of Health, Family Planning and Urban Development (From 14.11.1967 to 18.2.1969)
		Minister of State in the Ministry of Health and Family Planning and in the Ministry of Works, Housing and in Urban Development (From 18.2.1969 to 13.3.1971)
157.	Murthy, Shri M.V. Chandrasekhara	Minister of State in the Ministry of Finance (From 18.1.1993 to 16.5.1996*)
158.	Nagmani, Shri	Minister of State in the Ministry of Social Justice and Empowerment (From 24.5.2003 to 22.5.2004*)
159.	Naik, Shri Porika Balram	Minister of State in the Ministry of Social Justice and Empowerment (From 28.10.2012 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
160.	Naik, Shri Shripad Yasso	Minister of State in the Ministry of Agriculture (From 30.9.2000 to 2.11.2001)
		Minister of State in the Ministry of Shipping (From 2.11.2001 to 14.5.2002, the date of his resignation)
		Minister of State in the Ministry of Civil Aviation (From 1.7.2002 to 24.5.2003)
		Minister of State in the Ministry of Road Transport and Highways (From 24.5.2003 to 8.9.2003)
		Minister of State in the Ministry of Finance (From 8.9.2003 to 22.5.2004*)
161.	Namgyal, Shri P.	Deputy Minister in the Ministry of Surface Transport (From 15.2.1988 to 25.6.1988)
		Deputy Minister in the Ministries of Surface Transport and Parliamentary Affairs (From 25.6.1988 to 4.7.1989)
		Minister of State in the Ministries of Parliamentary Affairs and Department of Chemicals and Petro-Chemicals (From 4.7.1989 to 2.12.1989*)
162.	Napoleon, Shri D.	Minister of State in the Ministry of Social Justice and Empowerment (From 28.5.2009 to 20.3.2013)
163.	Naqvi, Shri Mukhtar Abbas	Minister of State in the Ministry of Information and Broadcasting (From 20.3.1998 to 13.10.1999*)
		Held charge of the Ministry of Parliamentary Affairs as a Minister of State (From 16.2.1999 to 13.10.1999*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
164.	Narah, Smt. Ranee	Minister of State in the Ministry of Tribal Affairs (From 28.10.2012 to 26.05.2014*)
165.	Narayanan, Shri K.R.	Minister of State in the Ministry of Planning (From 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of External Affairs (From 25.9.1985 to 22.10.1986)
		Minister of State in the Ministry of Science and Technology and in the Department of Atomic Energy, Electronics, Ocean Development and Space (From 22.10.1986 to 2.12.1989*)
166.	Narayanswamy, Shri V.	Minister of State in the Ministry of Parliamentary Affairs and Minister of State in the Ministry of Planning (From 6.4.2008 to 22.5.2009*)
		Minister of State in the Ministry of Planning (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Parliamentary Affairs (From 28.5.2009 to 12.7.2011)
		Minister of State in the Ministry of Personnel, Public Grievances and Pensions (From 15.11.2010 to 26.05.2014*)
		Minister of State in the Prime Minister's Office (From 19.1.2011 to 26.05.2014*)
167.	Narendra, Shri A.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Rural Development (From 23.5.2004 to 24.8.2006, the date of his resignation)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
168.	Natchiappan, Dr. E.M. Sudarsana	Minister of State in the Ministry of Commerce and Industry (From 17.6.2013 to 26.05.2014*)
169.	Nayar, Dr. Sushila	Minister of Health (As on of the Ministers of State) (From 10.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of Health and Family Planning (As one of the Ministers of State) (From 24.1.1966 to 13.3.1967*)
170.	Netam, Shri Arvind	Deputy Minister in the Ministry of Education and Social Welfare and in the Department of Culture (From 5.2.1973 to 24.3.1977*)
		Minister of State in the Ministry of Agriculture (From 18.1.1993 to 20.2.1996)
171.	Nitiraj Singh Choudhary, Shri	Minister of State in the Ministry of Petroleum, Chemicals and Mines and Metals (From 26.6.1970 to 18.3.1971*)
		Minister of State in the Ministry of Petroleum, Chemicals and Non-ferrous Metals (From 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Law and Justice (From 2.5.1971 to 5.2.1973)
		Minister of State in the Ministry of Law and Justice and Company Affairs (From 5.2.1973 to 17.10.1974)
172.	Oraon, Shri Kartik	Minister of State in the Ministry of Tourism and Civil Aviation (From 14.1.1980 to 8.6.1980)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Communications (From 8.6.1980 to 8.12.1981)
173.	Oraon, Shri Rameshwar	Minister of State in the Ministry of Tribal Affairs (From 6.4.2008 to 22.5.2009*)
174.	Oraon, Smt. Sumati	Deputy Minister in the Ministry of Welfare (From 14.2.1988 to 4.7.1989)
		Minister of State in the Ministry of Environment and Forests (From 4.7.1989 to 2.12.1989*)
175.	Oza, Shri Ghanshyam	Minister of State in the Ministry of Industrial Development (From 2.5.1971 to 17.3.1972)
176.	Pachouri, Shri Suresh	Minister of State in the Ministry of Defence (Department of Defence Production and Supplies) (From 15.9.1995 to 16.5.1996*)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Personnel, Public Grievances and Pensions and Minister of State in the Ministry of Parliamentary Affairs (From 23.5.2004 to 6.4.2008, the date of his resignation)
177.	Pahadia, Shri Jagannath	Deputy Minister in the Ministry of Finance (From 13.11.1967 to 17.10.1969)
		Deputy Minister in the Ministry of Food, Agriculture, Community Development and Cooperation (From 26.6.1970 to 18.3.1971*)
		Deputy Minister in the Department of Supply (From 18.3.1971 to 2.5.1971)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Agriculture (From 2.5.1971 to 22.7.1972)
		Deputy Minister in the Ministry of Communications (From 22.7.1972 to 23.12.1976)
		Deputy Minister in the Ministry of Labour (From 23.12.1976 to 24.3.1977*)
		Deputy Minister in the Ministry of Finance (From 14.1.1980 to 6.6.1980)
178.	Pal, Shri Debi Prosad	Minister of State in the Ministry of Finance (From 15.9.1995 to 16.5.1996*)
179.	Pala, Shri Vincent	Minister of State in the Ministry of Water Resources (From 28.5.2009 to 27.10.2012)
		Minister of State in the Ministry of Minority Affairs (From 19.1.2011 to 27.10.2012)
180.	Palanimanickam, Shri S.S.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Commerce and Industry (From 23.5.2004 to 25.5.2004)
		Minister of State in the Ministry of Finance (From 25.5.2004 to 22.5.2009*)
		Minister of State in the Ministry of Finance (From 28.5.2009 to 20.3.2013)
181.	Panigrahi, Shri Chintamani	Minister of State in the Ministry of Home Affairs (From 22.10.1986 to 25.6.1988)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Defence—Department of Defence Production and Defence Supplies (From 25.6.1988 to 4.7.1989)
182.	Panja, Shri Ajit Kumar	Minister of State in the Ministry of Planning (From 25.9.1985 to 22.10.1986)
		Minister of State in the Ministry of Planning and in the Ministry of Food and Civil Supplies (From 19.2.1986 to 22.10.1986)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 22.10.1986 to 14.2.1988)
		Minister of State in the Ministry of Finance—Department of Revenue (From 14.2.1988 to 2.12.1989*)
		Minister of State (Independent Charge) of the Ministry of Information and Broadcasting (From 21.6.1991 to 18.1.1993)
		Minister of State (Independent Charge) of the Ministry of Coal (From 18.1.1993 to 13.9.1995)
		Minister of State in the Ministry of External Affairs (From 13.10.1999 to 16.3.2001, the date of his resignation)
183.	Paswan, Dr. (Prof.) Sanjay	Minister of State in the Ministry of Communications and Information Technology (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Social Justice and Empowerment (From 29.1.2003 to 24.5.2003)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Human Resource Development (From 24.5.2003 to 22.5.2004*)
184.	Patel, Shri Prahlad	Minister of State in the Ministry of Coal (From 24.5.2003 to 22.5.2004*)
185.	Patel, Shri Rampoojan	Minister of State in the Ministry of Food and Civil Supplies (From 23.4.1990 to 10.11.1990*)
186.	Patel, Shri Uttambhai H.	Minister of State in the Ministry of Rural Development (From 21.6.1991 to 2.7.1992)
		Minister of State in the Ministry of Rural Development-Department of Rural Development (From 2.7.1992 to 11.6.1995)
		Minister of State in the Ministry of Rural Areas and Employment (Department of Rural Development) (From 11.6.1995 to 16.5.1996*)
187.	Patel, Smt. Urmilaben Chiman Bhai	Minister of State in the Ministry of Power (From 10.2.1995 to 16.5.1996*)
188.	Pathak, Shri Harin	Minister of State in the Ministry of Defence (From 13.10.1999 to 14.11.2000, the date of his resignation)
		Minister of State in the Ministry of Defence—Department of Defence Production and Supplies (From 15.10.2001 to 29.1.2003)
		Minister of State in the Ministry of Home Affairs and in the Ministry of Personnel, Public Grievances and Pensions (From 29.1.2003 to 22.5.2004*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
189.	Patil, Shri Annasaheb	Minister of State in the Ministry of Rural Development (From 1.9.2001 to 22.5.2004*)
190.	Patil, Shri Basangouda	Minister of State in the Ministry of Textiles (From 1.7.2002 to 8.9.2003)
		Minister of State in the Ministry of Railways (From 8.9.2003 to 22.5.2004*)
191.	Patil, Shri Basavraj	Minister of State in the Ministry of Steel and Mines (From 21.11.1990 to 20.2.1991)
192.	Patil, Shri Prateek Prakashbapu	Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 28.5.2009 to 14.6.2009)
		Minister of State in the Ministry of Youth Affairs and Sports (From 14.6.2009 to 19.1.2011)
		Minister of State in the Ministry of Coal (From 19.1.2011 to 26.05.2014*)
193.	Patil, Shri S.D.	Minister of State in the Ministry of Home Affairs (From 14.8.1977 to 28.7.1979*)
		Also Minister of State in the Ministry of Law, Justice and Company Affairs (From 20.2.1979 to 28.7.1979*)
194.	Patil, Smt. Suryakanta	Minister of State without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Rural Development and in the Ministry of Parliamentary Affairs (From 23.5.2004 to 22.5.2009*)
195.	Pattabhi Rama Rao, Shri S.B.	Minister of State in the Ministry of Finance (From 11.9.1982 to 7.2.1984)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Industry (From 7.2.1984 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
196.	Pattabhi Raman, Shri C.R.	Deputy Minister in the Ministry of Planning (From 8.5.1962 to 27.5.1964*)
		Deputy Minister in the Ministry of Labour, Employment and Planning (From 21.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Information and Broadcasting (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of Law (From 24.1.1966 to 13.3.1967*)
197.	Poojari, Shri Janardhana	Deputy Minister in the Ministry of Finance—Department of Banking and Insurance (From 15.1.1982 to 31.10.1984*) (Again from 31.10.1984 to 31.12.1984*)
		Minister of State in the Ministry of Finance (From 31.12.1984 to 14.2.1988)
		Minister of State in the Ministry of Agriculture—Department of Rural Development (From 14.2.1988 to 2.12.1989*)
198.	Potdukhe, Shri Shantaram	Minister of State in the Ministry of Finance (From 21.6.1991 to 17.1.1993)
199.	Prabhu, Shri Rajendra	Minister of State in the Ministry of Agriculture—Department of Fertilisers (From 22.10.1986 to 2.12.1989)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
200.	Pradhan, Dr. (Shri) Debendra	Minister of State in the Ministry of Surface Transport (From 19.3.1998 to 13.10.1999*) (Again from 13.10.1999 to 27.5.2000)
		Minister of State in the Ministry of Agriculture (From 27.5.2000 to 29.1.2003, the date of his resignation)
201.	Pradhan, Shri Ashok Kumar	Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 1.9.2001 to 1.7.2002)
		Minister of State in the Ministry of Labour (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Human Resource Development (From 29.1.2003 to 24.5.2003)
		Minister of State in the Ministry of Communications and Information Technology (From 24.5.2003 to 22.5.2004*)
202.	Prasad, Shri V. Sreenivasa	Minister of State in the Ministry of Consumer Affairs and Public Distribution (From 13.10.1999 to 17.7.2000)
		Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 17.7.2000 to 6.3.2004, the date of his resignation)
203.	Prasada, Shri Jitin	Minister of State in the Ministry of Steel (From 6.4.2008 to 22.5.2009*)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 28.5.2009 to 19.1.2011)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Road Transport and Highways (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Human Resource Development (From 28.10.2012 to 26.5.2014*)
204.	Prathap, Shri A. Sai	Minister of State in the Ministry of Steel (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 19.1.2011 to 12.7.2011)
205.	Punnuswamy, Shri E.	Minister of State in the Ministry of Petroleum and Natural Gas (From 13.10.1999 to 7.2.2001, the date of his resignation)
206.	Purandeswari, Smt. D.	Minister of State in the Ministry of Human Resource Development (From 29.1.2006 to 22.5.2009*)
		Minister of State in the Ministry of Human Resource Development (From 28.5.2009 to 28.10.2012)
		Minister of State in the Ministry of Commerce and Industry (From 28.10.2012 to 11.3.2014)
207.	Purkayastha, Shri Kabindra	Minister of State in the Ministry of Communications (From 20.3.1998 to 13.10.1999*)
208.	R.K. Kumar, Shri	Minister of State in the Ministry of Finance (From 20.3.1998 to 22.5.1998, the date of his resignation)
		Held charge of the Ministry of Parliamentary Affairs as a Minister of State (From 19.3.1998 to 22.5.1998)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
209.	Rachaiah, Shri R.	Minister of State in the Ministry of Industry (From 4.8.1979 to 14.1.1980*)
210.	Radhakrishnan, Shri P.	Minister of State in the Ministry of Youth Affairs and Sports (From 30.9.2000 to 29.1.2003)
		Minister of State in the Ministry of Urban Development and Poverty Alleviation (From 29.1.2003 to 8.9.2003)
		Minister of State in the Ministry of Road Transport and Highways (From 8.9.2003 to 22.5.2004*)
211.	Rahi, Shri Ram Lal	Deputy Minister in the Ministry of Home Affairs (From 21.6.1991 to 15.9.1995)
		Minister of State in the Ministry of Home Affairs (From 15.9.1995 to 16.5.1996*)
212.	Rahim, Shri A.A.	Minister of State in the Ministry of Law, Justice and Company Affairs (From 15.1.1982 to 2.9.1982)
		Minister of State in the Ministry of External Affairs (From 2.9.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
213.	Rajagopal, Shri Olencheri	Minister of State in the Ministry of Law, Justice and Company Affairs (From 13.10.1999 to 24.7.2000)
		Minister of State in the Ministry of Parliamentary Affairs (From 22.11.1999 to 1.7.2002)
		Held additional charge of Minister of State in the Ministry of Railways (From 31.8.2000 to 1.7.2002)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Parliamentary Affairs with additional charge of the Ministry of Urban Development and Poverty Alleviation (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Defence and in the Ministry of Parliamentary Affairs (From 29.1.2003 to 22.5.2004*)
214.	Rajashekharan, Shri M.V.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Planning (From 23.5.2004 to 6.4.2008, the date of his resignation)
215.	Ram Sewak, Shri Chaudhary	Deputy Minister in the Ministry of Steel, Mines and Metals (From 13.11.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Foreign Trade and Supply (From 14.2.1969 to 4.11.1969)
		Deputy Minister in the Ministry of Foreign Trade (From 4.11.1969 to 18.3.1971*)
		Minister of State in the Ministry of Health and Family Planning (From 1.12.1975 to 24.3.1977*)
216.	Ram Singh, Col.	Minister of State in the Ministry of Rural Development (Department of Wasteland Development) (From 2.7.1992 to 11.6.1995)
		Minister of State in the Ministry of Rural Areas and Employment (Department of Wasteland Development) (From 11.6.1995 to 30.3.1996)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
217.	Ramachandran, Shri Gingee N.	Minister of State in the Ministry of Textiles (From 13.10.1999 to 30.9.2000)
		Minister of State in the Ministry of Finance (From 30.9.2000 to 29.1.2003)
		Minister of State in the Ministry of Finance and Company Affairs (From 29.1.2003 to 24.5.2003, the date of his resignation)
		Minister of State in the Ministry of Textiles (From 8.9.2003 to 30.12.2003, the date of his resignation)
218.	Ramachandran, Shri Mullapally	Minister of State in the Ministry of Agriculture (From 21.6.1991 to 17.1.1993)
		Minister of State in the Ministry of Home Affairs (From 28.5.2009 to 26.05.2014*)
219.	Raman, Dr.	Minister of State in the Ministry of Commerce and Industry (From 13.10.1999 to 29.1.2003, the date of his resignation)
220.	Ramanna, Dr. Raja	Minister of State in the Ministry of Defence (From 20.1.1990 to 10.11.1990*)
221.	Rana, Shri M.B.	Minister of State in the Ministry of Shipping and Transport (From 5.2.1973 to 11.1.1974)
		Minister of State in the Ministry of Industrial Development (From 11.1.1974 to 31.7.1974, the date of his death)
222.	Rangayya Naidu, Shri P.V.	Deputy Minister in the Ministry of Communications (From 21.6.1991 to 18.1.1993)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Power (From 18.1.1993 to 10.2.1995)
		Minister of State in the Ministry of Water Resources (From 10.2.1995 to 16.5.1996*)
223.	Rao, Dr. (Shri) Dasari Narayan	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Coal and Mines (From 23.5.2004 to 27.11.2004)
		Minister of State in the Ministry of Coal (From 27.11.2004 to 6.4.2008, the date of his resignation)
		Minister of State in the Ministry of Mines (From 27.11.2004 to 7.2.2006)
224.	Rao, Dr. K.L.	Minister of Irrigation and Power (As one of the Ministers of State) (From 19.7.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Minister of State in the Ministry of Irrigation and Power (From 9.6.1964 to 19.7.1964) (Again from 19.7.1964 to 11.1.1966*) (Again from 24.1.1966 to 13.3.1967*) (Again from 13.3.1967 to 18.3.1971*) (Again from 18.3.1971 to 9.11.1973)
225.	Rao, Shri P. Ankineedu Prasad	Minister of State in the Ministry of Tourism and Civil Aviation (From 4.8.1979 to 27.11.1979)
226.	Rao, Shri Chennamaneni Vidyasagar	Minister of State in the Ministry of Home Affairs (From 13.10.1999 to 29.1.2003)
		Minister of State in the Ministry of Commerce and Industry (From 29.1.2003 to 22.5.2004*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
227.	Rao, Shri Indrajit Singh	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of External Affairs (From 23.5.2004 to 29.1.2006)
		Minister of State in the Ministry of Defence (From 29.1.2006 to 22.5.2009*)
228.	Rao, Shri Jagannath	Deputy Minister in the Ministry of Works, Housing and Supply (From 8.5.1962 to 15.11.1962)
		Deputy Minister in the Ministry of Works and Housing (From 15.11.1962 to 26.11.1962)
		Deputy Minister in the Ministry of Economic and Defence Coordination (From 26.11.1962 to 1.9.1963)
		Deputy Minister in the Department of Supply (From 1.9.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Law (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Also assumed the office of Deputy Minister in the Department of Social Security (From 6.7.1964 to 11.1.1966*)
		Minister of State in the Ministry of Labour, Employment and Rehabilitation (From 24.1.1966 to 14.2.1966)
		Minister of State in the Departments of Parliamentary Affairs and Communications (From 14.2.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State, Works, Housing and Supply (From 13.3.1967 to 14.2.1969)
		Minister of State in the Ministry of Petroleum and Chemicals, Mines and Metals (From 14.2.1969 to 27.6.1970)
		Minister of State in the Ministry of Law and in the Department of Social Welfare (From 27.6.1970 to 18.3.1971*)
229.	Rao, Shri S.B.P.B.K. Satyanarayana	Minister of State in the Ministry of Agriculture (From 13.10.1999 to 29.9.2000)
230.	Rath, Shri R.C.	Minister of State in the Ministry of Chemicals and Fertilisers (From 11.9.1982 to 31.10.1984)
231.	Rathwa, Shri Narayanbhai	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Railways (From 23.5.2004 to 22.5.2009*)
232.	Rawat, Shri Bachi Singh	Minister of State in the Ministry of Defence (From 13.10.1999 to 22.11.1999)
		Minister of State in the Ministry of Science and Technology—Department of Science and Technology (From 22.11.1999 to 22.5.2004*)
233.	Reddy, Shri K.J. Surya Prakash	Minister of State in the Ministry of Railways (From 28.10.2012 to 26.05.2014*)
234.	Reddy, Shri K.V. Raghunatha	Minister of State in the Ministry of Industrial Development and Company Affairs (From 13.3.1967 to 14.2.1969)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Industrial Development, Internal Trade and Company Affairs (From 14.2.1969 to 27.6.1970)
		Minister of State in the Department of Company Affairs (From 27.6.1970 to 18.3.1971*) (Again from 18.3.1971 to 5.2.1973)
		Minister of State in the Department of Labour and Rehabilitation (From 5.2.1973 to 9.11.1973)
		Minister of State in the Ministry of Labour (From 9.11.1973 to 24.3.1977*)
235.	Reddy, Shri T. Subbarami	Minister of State without portfolio (From 29.1.2006 to 31.1.2006)
		Minister of State in the Ministry of Mines (From 31.1.2006 to 6.4.2008, the date of his resignation)
236.	Regupathy, Shri S.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Home Affairs (From 23.5.2004 to 15.5.2007)
		Minister of State in the Ministry of Environment and Forests (From 15.5.2007 to 22.5.2009*)
237.	Rohtagi, Smt. Sushila	Deputy Minister in the Ministry of Finance (From 2.5.1971 to 24.3.1977*)
		Minister of State in the Ministry of Human Resource Development— Department of Education and Culture (From 25.9.1985 to 12.5.1986)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Energy—Department of Power (From 12.5.1986 to 9.5.1988)
		Also Minister of State in the Ministry of Petroleum and Natural Gas (From 24.6.1986 to 22.10.1986)
238.	Roy, Shri Saugata	Minister of State in the Ministry of Petroleum, Chemicals and Fertilisers (From 4.8.1979 to 14.1.1980*)
		Minister of State in the Ministry of Urban Development (From 28.5.2009 to 22.9.2012)
239.	Sahi, Shri Laliteshwar Prasad	Minister of State in the Ministry of Human Resource Development— Department of Education and Culture (From 14.2.1988 to 4.7.1989)
		Minister of State in the Ministry of Human Resource Development— Department of Education (From 4.7.1989 to 2.12.1989)
240.	Sahi, Smt. Krishna	Minister of State in the Ministry of Human Resource Development (Department of Education and Culture) (From 12.5.1986 to 14.2.1988)
		Minister of State in the Ministry of Water Resources (From 14.2.1988 to 4.7.1989)
		Minister of State in the Ministry of Human Resource Development (Department of Culture) (From 4.7.1989 to 2.12.1989*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Industry—Department of Industrial Development (From 2.7.1992 to 15.9.1995)
		Also Minister of State in the Ministry of Heavy Industries (From 19.2.1993 to 15.9.1995)
		Minister of State in the Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 15.9.1995 to 19.9.1995)
		Minister of State in the Ministry of Civil Supplies, Consumer Affairs and Public Distribution—Department of Civil Supplies (From 19.9.1995 to 16.5.1996)
241.	Sahu, Shri Chandrasekhar	Minister of State in the Ministry of Labour amd Employment (From 29.1.2006 to 24.10.2006)
		Minister of State in the Ministry of Rural Development (From 24.10.2006 to 22.5.2009*)
242.	Sai, Shri Larang	Minister of State in the Ministries of Labour and Parliamentary Affairs (From 14.8.1977 to 28.7.1979*)
243.	Sai, Shri Narhari Prasad Sukhdeo	Minister of State in the Ministry of Communications (From 14.8.1977 to 28.7.1979*)
244.	Saikia, Shri Muhi Ram	Minister of State in the Ministry of Human Resource Development (Department of Education) (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 19.3.1998*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
245.	Saksena, Shri Mohan Lal	Minister of State in the Ministry of Relief and Rehabilitation (From 18.6.1948 to 26.1.1950)
		Minister of State in the Ministry of Rehabilitation (From 31.1.1950 to 6.5.1950) (Again from 6.5.1950 to 1.6.1950)
246.	Sangma, Ms. Agatha	Minister of State in the Ministry of Rural Development (From 28.5.2009 to 27.10.2012)
247.	Santhanam, Shri K.	Minister of State in the Ministry of Transport and Railways (From 1.10.1948 to 26.1.1950*) (Again from 31.1. 1950 to 6.5.1950*) (Again from 6.5.1950 to 29.2.1952, the date of his resignation)
248.	Sathyanarayana, Shri Sarvey	Minister of State in the Ministry of Road Transport and Highways (From 28.10.2012 to 26.05.2014*)
249.	Satpalji Maharaj, Shri	Minister of State in the Ministry of Railways (From 6.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State in the Ministry of Finance (From 9.6.1997 to 19.3.1998*)
250.	Satpathy, Smt. Nandini	Deputy Minister in the Ministry of Information and Broadcasting (From 29.1.1966 to 13.3.1967*) (Again from 18.3.1967 to 14.2.1969)
		Deputy Minister without portfolio (From 14.2.1969 to 26.6.1970)
		Minister of State without portfolio (From 26.6.1970 to 18.3.1971*)
		Minister of State in the Ministry of Information and Broadcasting (From 18.3.1971 to 14.6.1972)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
251.	Savanoor, Smt. Ratnamala Dhareshwar	Minister of State (without portfolio) (From 3.6.1997 to 9.6.1997)
		Minister of State in the Ministry of Planning and Programme Implementation (From 9.6.1997 to 19.3.1998*)
252.	Seelam, Shri Jesudasu	Minister of State in the Ministry of Finance (From 17.6.2013 to 26.05.2014*)
253.	Selvi, Smt. V. Radhika	Minister of State in the Ministry of Home Affairs (From 18.5.2007 to 22.5.2009*)
254.	Shah, Shri Jayanthilal V.	Minister of State in the Ministry of Agriculture—Department of Agriculture and Cooperation (From 21.11.1990 to 25.4.1991)
255.	Sharma, Shri Venod	Minister of State in the Ministry of Civil Supplies, Consumer Affairs and Public Distribution (From 15.9.1995 to 19.9.1995)
		Minister of State in the Ministry of Civil Supplies, Consumer Affairs and Public Distribution—Department of Consumer Affairs and Public Distribution System (From 19.9.1995 to 16.5.1996*)
256.	Shastri, Shri Hari Krishna	Minister of State in the Ministry of Agriculture—Department of Agriculture Research and Education (From 14.2.1988 to 2.12.1989*)
257.	Sheo Narain, Shri	Minister of State in the Ministry of Railways (From 14.8.1977 to 28.7.1979*)
258.	Sher Singh, Prof.	Minister of State in the Ministry of Education (From 13.3.1967 to 14.2.1969)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Information and Broadcasting and in the Department of Communications (From 14.2.1969 to 18.3.1971*)
		Minister of State in the Ministry of Communications (From 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Agriculture (From 2.5.1971 to 12.1.1974)
		Minister of State in the Ministry of Communications (From 12.1.1974 to 10.10.1974)
		Minister of State in the Ministry of Defence (From 14.8.1977 to 28.7.1979*)
		Held charge of the Departments of Atomic Energy, Electronics, Science and Technology and Space as a Minister of State (From 24.3.1979 to 28.7.1979*)
259.	Shervani, Shri Saleem Iqbal	Minister of State in the Ministry of Health and Family Welfare (From 1.6.1996 to 29.6.1996)
		Minister of State (Independent Charge) of the Ministry of Health and Family Welfare (From 29.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 9.6.1997)
		Minister of State in the Ministry of External Affairs (From 9.6.1997 to 19.3.1998*)
260.	Shinde, Shri Anna Sahib P.	Deputy Minister in the Ministry of Food and Agriculture, Community Development and Cooperation (From 24.1.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Food and Agriculture, Community Development and Cooperation (From 13.3.1967 to 18.3.1971*)
		Minister of State in the Ministry of Food and Agriculture (From 18.3.1971 to 2.5.1971)
		Minister of State in the Ministry of Agriculture (From 2.5.1971 to 10.10.1974)
		Minister of State in the Ministry of Agriculture and Irrigation (From 10.10.1974 to 24.3.1977*)
261.	Shrangare, Shri T.S.	Minister of State in the Ministry of Communications (From 31.7.1979 to 14.1.1980*)
262.	Shreedharan, Shri Arangil	Minister of State in the Ministry of Commerce (From 23.4.1990 to 10.11.1990*)
263.	Shukla, Shri Rajeev	Minister of State in the Ministry of Parliamentary Affairs (From 12.7.2011 to 26.05.2014*)
		Minister of State in the Ministry of Planning (From 28.10.2012 to 26.05.2014*)
264.	Sibtey Razi, Syed	Minister of State in the Ministry of Home Affairs (From 15.9.1995 to 16.5.1996*)
265.	Sidhwa, Shri R.K.	Minister of State for Home Affairs (From 11.10.1951 to 13.5.1952*)
266.	Sikdar, Shri Tapan	Minister of State in the Ministry of Communications (From 13.10.1999 to 11.1.2002)
		Minister of State in the Ministry of Communications and Information Technology (From 11.1.2002 to 1.7.2002)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Chemicals and Fertilizers (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Small Scale Industries and in the Department of Development of North Eastern Region (From 29.1.2003 to 22.5.2004*)
267.	Silvera, Shri C.	Minister of State in the Ministry of Health and Family Welfare (From 17.2.1994 to 15.9.1995)
		Minister of State in the Ministry of Industry (From 15.9.1995 to 16.5.1996*)
268.	Singh, Shri (Dr.) Akhilesh Prasad	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministries of Agriculture and Consumer Affairs, Food and Public Distribution (From 23.5.2004 to 22.5.2009*)
269.	Singh, Shri Bhanu Pratap	Minister of State in the Ministry of Agriculture and Irrigation (From 14.8.1977 to 15.7.1979)
		Minister of State in the Ministry of Rural Reconstruction (From 30.7.1979 to 14.1.1980*)
270.	Singh, Shri C.P.N.	Minister of State in the Ministry of Defence and other Departments held by the Prime Minister (From 3.3.1980 to 19.10.1980)
		Minister of State in the Department of Science and Technology (From 19.10.1980 to 2.2.1982)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of State in the Department of Electronics (From 30.10.1980 to 15.2.1982)
		Minister of State in the Department of Science and Technology, Electronics and Environment (From 15.1.1982 to 15.7.1982)
		Minister of State in the Department of Science and Technology, Electronics and Environment and Ocean Development (From 15.2.1982 to 2.9.1982)
		Minister of State in the Department of Science and Technology, Electronics, Ocean Development and the Department of Non-Conventional Energy Sources in the Ministry of Energy (From 2.9.1982 to 2.2.1983)
271.	Singh, Shri Chhattrapal	Minister of State in the Ministry of Chemicals and Fertilizers (From 29.1.2003 to 16.3.2004*)
272.	Singh, Shri Hari Kishore	Minister of State in the Ministry of External Affairs (From 23.4.1990 to 10.11.1990)
273.	Singh, Shri Jitendra	Minister of State in the Ministry of Home Affairs (From 12.7.2011 to 28.10.2012)
		Minister of State (Independent Charge) of the Ministry of Youth Affairs and Sports (From 28.10.2012 to 26.05.2014*)
		Minister of State in the Ministry of Defence (From 28.10.2012 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
274.	Singh, Shri Lalit Vijoy	Minister of State in the Ministry of Defence (From 21.11.1990 to 21.6.1991*)
275.	Singh, Shri R.P.N.	Minister of State in the Ministry of Road Transport and Highways (From 28.5.2009 to 19.1.2011)
		Minister of State in the Ministry of Petroleum and Natural Gas (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Corporate Affairs (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Home Affairs (From 28.10.2012 to 26.05.2014*)
276.	Singh, Shri Ram Bahad	dur Minister of State in the Ministry of Agriculture—Department of Rural Development (From 21.11.1990 to 21.6.1991*)
277.	Singh, Shri Surendra F	Pal Deputy Minister in the Ministry of External Affairs (From 18.3.1967 to 18.3.1971*) (Again from 18.3.1971 to 5.2.1973)
		Minister of State in the Ministry of External Affairs (From 5.2.1973 to 10.10.1974)
	Minister of State in the Ministry of Tourism and Civil Aviation (From 10.10.1974 to 23.12.1976)	
		Minister of State in the Ministry of Railways (From 23.12.1976 to 24.3.1977*)
278.	Singh, Smt. Usha	Deputy Minister in the Ministry of Welfare—Department of Women and Child Development (From 23.4.1990 to 5.11.1990)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Tourism (From 21.11.1990 to 10.4.1991)
279.	Sinha, Dr. Ram Kripal	Minister of State in the Ministry of Labour and Parliamentary Affairs (From 14.8.1977 to 28.7.1979*)
280.	Sinha, Smt. Kamla	Minister of State (without portfolio) (From 3.6.1997 to 9.6.1997)
		Minister of State in the Ministry of External Affairs (From 9.6.1997 to 19.3.1998*)
281.	Sinha, Smt. Ramdulari	Minister of State in the Ministry of Information and Broadcasting (From 8.6.1980 to 19.10.1980)
		Minister of State in the Ministry of Labour and Rehabilitation (From 19.10.1980 to 15.1.1982)
		Minister of State in the Ministries of Industry, Steel and Mines (From 15.1.1982 to 14.2.1983)
		Minister of State in the Ministry of Commerce (From 14.2.1983 to 7.2.1984)
		Minister of State in the Ministry of Home Affairs (From 7.2.1984 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*) (Again from 31.12.1984 to 25.9.1985)
		Minister of State in the Ministry of Steel and Mines—Department of Mines (From 25.9.1985 to 14.2.1988)
282.	Sisodia, Shri Sawai Singh	Minister of State in the Ministry of Finance (Department of Revenue and Expenditure) (From 19.10.1980 to 2.9.1982)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
283.	Sompal, Shri	Minister of State in the Ministry of Agriculture (From 19.3.1998 to 13.10.1999*)
		Minister of State in the Ministry of Water Resources (From 3.2.1999 to 13.10.1999*)
284.	Somu, Shri N.V.N.	Minister of State in the Ministry of Defence (From 6.7.1996 to 21.4.1997*) (Again from 21.4.1997 to 14.11.1997)
285.	Sukhbans Kaur, Smt.	Minister of State in the Ministry of Civil Aviation and Tourism (Department of Tourism) (From 2.7.1992 to 16.5.1996*)
286.	Suman, Shri Ramjilal	Minister of State in the Ministry of Labour and Welfare (From 21.11.1990 to 21.6.1991*)
287.	Suresh, Shri Kodikunnil	Minister of State in the Ministry of Labour and Employment (From 28.10.2012 to 26.05.2014*)
288.	Swami Chinmayanand, Shri	Minister of State in the Ministry of Home Affairs (From 24.5.2003 to 22.5.2004*)
289.	Swami, Shri I. D.	Minister of State in the Ministry of Home Affairs (From 13.10.1999 to 22.5.2004*)
290.	Swaminathan, Shri R.V.	Minister of State in the Ministry of Agriculture (From 14.1.1980 to 29.1.1983)
		Also Minister of State in the Ministry of Rural Reconstruction (From 24.11.1980 to 29.1.1983)
291.	Tariq Anwar, Shri	Minister of State in the Ministry of Agriculture (From 28.10.2012 to 26.05.2014*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Food Processing Industries (From 28.10.2012 to 26.05.2014*)
292.	Taslim Uddin, Shri	Minister of State in the Ministry of Home Affairs (From 1.6.1996 to 9.7.1996, the date of his resignation)
		Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 23.5.2004 to 25.5.2004)
		Minister of State in the Ministry of Agriculture and Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 25.5.2004 to 22.5.2009*)
293.	Tewari, Prof. Kamla Kant	Minister of State in the Ministry of Industry (Department of Public Enterprises) (From 12.5.1986 to 23.7.1987)
		Minister of State in the Ministry of External Affairs (From 25.6.1988 to 22.4.1989)
		Minister of State in the Ministry of Information and Broadcasting (From 22.4.1989 to 2.12.1989*)
294.	Thakur Chaoba Singh, Shri	Minister of State in the Ministry of Culture, Youth Affairs and Sports (From 13.10.1999 to 2.2.2000)
		Minister of State of the Department of Youth Affairs and Sports (From 2.2.2000 to 27.5.2000)
		Minister of State of the Department of Food Processing in the Ministry of Agriculture (From 27.5.2000 to 1.9.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
295.	Thakur, Shri Rameshwar	Minister of State in the Ministry of Finance (From 26.6.1991 to 18.1.1993)
		Minister of State in the Ministry of Rural Development (Department of Rural Development) (From 18.1.1993 to 17.4.1994)
		Minister of State in the Ministry of Rural Development (Department of Rural Development) and Minister of State in the Ministry of Parliamentary Affairs (From 17.4.1994 to 22.12.1994)
296.	Thangkabalu, Shri K.V.	Minister of State in the Ministry of Health and Family Welfare (From. 19.1.1993 to 16.5.1996*)
297.	Thara Devi, Smt. D.K. Siddartha	Minister of State in the Ministry of Health and Family Welfare (From 26.6.1991 to 18.1.1993)
298. Tharoor, Dr. Shashi	Tharoor, Dr. Shashi	Minister of State in the Ministry of External Affairs (From 28.5.2009 to 18.4.2010, the date of his resignation)
		Minister of State in the Ministry of Human Resource Development (From 28.10.2012 to 26.05.2014*)
299.	Thirunavukkarasar, Shri S.U.	Minister of State in the Ministry of Shipping (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Communications and Information Technology (From 29.1.2003 to 22.5.2004*)
300.	Thomas, Shri Alungal Mathai	Deputy Minister in the Ministry of Food and Agriculture (From 21.5.1957 to 10.4.1962*) (Again from 16.4.1962 to 21.11.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Food and Agriculture (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 9.6.1964 to 13.6.1964)
		Minister of State, Defence Production in the Ministry of Defence (From 13.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Minister of State in the Ministry of Defence (From 24.1.1966 to 31.1.1966*)
		Minister of State in the Ministry of Defence—Department of Defence Production (From 31.1.1966 to 13.3.1967*)
301.	Thomas, Shri P.C.	Minister of State in the Ministry of Law and Justice (From 24.5.2003 to 22.5.2004*)
302.	Thungan, Shri P.K.	Deputy Minister in the Ministry of Supply and Rehabilitation (From 8.6.1980 to 15.1.1982)
		Deputy Minister in the Ministries of Education, Culture and Social Welfare (From 15.1.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Minister of State in the Ministry of Commerce and Industry (From 21.6.1991 to 2.7.1992)
		Minister of State in the Ministry of Industry—Department of Heavy Industry and Public Enterprises (From 2.7.1992 to 18.1.1993)
		Minister of State in the Ministry of Urban Affairs and Employment (From 18.1.1993 to 13.9.1995)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Also Minister of State in the Ministry of Water Resources (From 19.1.1993 to 10.2.1995)
303.	Trivedi, Shri H.M.	Minister of State in the Ministry of Shipping and Transport (From 17.10.1974 to 24.3.1977*)
304.	Uppalpati, Shri Venkata Krishnan Raju	Minister of State in the Ministry of External Affairs (From 30.9.2000 to 22.7.2001)
		Minister of State in the Ministry of Defence (From 22.7.2001 to 1.7.2002)
		Minister of State in the Ministry of Consumer Affairs, Food and Public Distribution (From 1.7.2002 to 29.1.2003)
		Minister of State in the Ministry of Rural Development (From 29.1.2003 to 22.5 2004*)
305.	Varma, Shri Chandradeo Prasad	Minister of State in the Ministry of Rural Areas and Employment (From 1.6.1996 to 21.4.1997*) (Again from 21.4.1997 to 20.6.1997, the date of his resignation)
306.	Velu, Shri R.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Railways (From 23.5.2004 to 29.3.2009, the date of his resignation)
307.	Venkatapathy, Shri K.	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Law and Justice (From 23.5.2004 to 22.5.2009*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
308.	Venkatasubbaiah, Shri P.	Minister of State in the Ministry of Home Affairs (From 14.1.1980 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
		Also Minister of State in the Ministry of Parliamentary Affairs (From 16.1.1980 to 2.9.1982)
309.	Venugopal, Shri K.C.	Minister of State in the Ministry of Power (From 19.1.2011 to 28.10.2012)
		Minister of State in the Ministry of Civil Aviation (From 28.10.2012 to 26.05.2014*)
310.	Venugopalachari, Dr. S.	Minister of State in the Ministry of Power (From 1.6.1996 to 9.6.1997)
		Minister of State in the Ministry of Non- Conventional Energy Sources (From 29.6.1996 to 21.2.1997)
		Minister of State in the Ministry of Agriculture (From 9.6.1997 to 19.3.1998*)
311.	Verma, Km. Vimla	Minister of State in the Ministry of Human Resource Development (Department of Women and Child Development) (From 15.9.1995 to 16.5.1996*)
312.	Verma, Prof. (Smt.) Rita	Minister of State in the Ministry of Mines and Minerals (From 13.10.1999 to 27.5.2000)
		Minister of State in the Ministry of Health and Family Welfare (From 27.5.2000 to 30.9.2000)
		Minister of State in the Ministry of Rural Development (From 30.9.2000 to 1.9.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Human Resource Development (From 1.9.2001 to 29.1.2003, the date of her resignation)
313.	Verma, Shri Upendra Nath	Minister of State in the Ministry of Agriculture (Department of Rural Development) (From 23.4.1990 to 10.11.1990*)
314.	Yadav, Shri Arun	Minister of State in the Ministry of Youth Affairs and Sports (From 28.5.2009 to 14.6.2009)
		Minister of State in the Ministry of Heavy Industries and Public Enterprises (From 14.6.2009 to 19.1.2011)
		Minister of State in the Ministry of Agriculture (From 19.1.2011 to 12.7.2011)
		Minister of State in the Ministry of Food Processing Industries (From 19.1.2011 to 12.7.2011)
315.	Yadav, Shri Chandrajit	Minister of State in the Ministry of Steel and Mines (From 10.10.1974 to 24.3.1977*)
316.	Yadav, Shri Hukumdeo Narayan	Minister of Food Processing Industries and Textiles (From 21.11.1990 to 21.6.1991*)
		Minister of State in the Ministry of Agriculture (From 13.10.1999 to 27.5.2000)
		Minister of State in the Ministry of Surface Transport (From 27.5.2000 to 7.11.2000)
		Minister of State in the Ministries of Surface Transport and Shipping (From 7.11.2000 to 2.11.2001)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Minister of State in the Ministry of Agriculture (From 2.11.2001 to 22.5.2004)
317.	Yadav, Shri Jagdambi Prasad	Minister of State in the Ministry of Health and Family Welfare (From 14.8.1977 to 26.1.1979)
		Minister of State in the Ministry of Industry (From 26.1.1979 to 28.7.1979*)
318.	Yadav, Shri Jay Prakash Narayan	Minister without portfolio (From 22.5.2004 to 23.5.2004)
		Minister of State in the Ministry of Water Resources (From 23.5.2004 to 6.11.2005) (Again from 24.10.2006 to 22.5.2009*)
319.	Yadav, Shri Narsingh	Minister of State in the Ministry of Law, Justice and Company Affairs (From 14.8.1977 to 11.7.1978)
		Minister of State in the Ministry of Petroleum, Chemicals and Fertilizers (From 26.1.1979 to 15.7.1979)
		Minister of State in the Ministry of Communications (From 30.7.1979 to 14.1.1980*)
320.	Yadav, Shri Ramanand	Minister of State in the Ministry of Agriculture (Department of Rural Development) (From 12.5.1986 to 14.2.1988)
		Minister of State in the Ministry of Steel and Mines (Department of Mines) (From 14.2.1988 to 12.4.1988)
321.	Yadav, Shri Satpal Singh	Minister of State in the Ministry of Food and Consumer Affairs (From 19.3.1998 to 13.10.1999*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
322.	Yadav, Shri Shyamlal	Minister of State in the Ministry of Agriculture—Department of Agriculture and Cooperation (From 14.2.1988 to 2.12.1989*)
323.	Zulfiquarullah, Shri	Minister of State in the Ministry of Finance (From 14.8.1977 to 15.7.1979)
		Minister of State in the Ministry of Communications and also Muslim Waqfs (From 30.7.1979 to 4.12.1979)

^{*}The date of dissolution of the Council of Ministers.

PART VII

DEPUTY MINISTERS/PARLIAMENTARY SECRETARIES

Sl.No.	Name	Portfolios alongwith the period thereof
1.	Abid Ali, Shri	Deputy Minister in the Ministry of Labour (From 12.8.1952 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962*)
2.	Ajay Singh, Shri	Deputy Minister in the Ministry of Railways (From 23.4.1990 to 10.11.1990*)
3.	Alva, Smt. Violet	Deputy Minister in the Ministry of Home Affairs (From 23.4.1957 to 10.4.1962*)
4.	Arif, Shri Mohammad Usman	Deputy Minister in the Ministry of Works and Housing (From 8.6.1980 to 15.1.1982)
		Deputy Minister in the Ministry of Agriculture and Civil Supplies (From 15.1.1982 to 6.9.1982)
		Deputy Minister in the Ministry of Civil Supplies (From 6.9.1982 to 14.2.1983)
		Deputy Minister in the Ministry of Works and Housing (From 14.2.1983 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
5.	Barot, Shri Magan Bhai	Deputy Minister in the Ministry of Finance (From 8.6.1980 to 15.1.1982)
6.	Basappa, Shri Kondajji	Deputy Minister in the Ministry of Health and Family Planning (From 5.2.1973 to 10.10.1974)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
7.	Bhagwati, Shri B.C.	Deputy Minister in the Ministry of Transport and Communications (From 8.5.1962 to 1.9.1963)
		Deputy Minister in the Department of Posts and Telegraphs (From 1.9.1963 to 13.5.1964)
		Deputy Minister in the Department of Communications (From 13.5.1964 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Works, Housing and Urban Development (From 24.1.1966 to 13.3.1967*)
8.	Bhanu Pratap Singh, Shri	Deputy Minister in the Ministry of Industrial Development and Company Affairs (From 18.3.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Industrial Development, Internal Trade and Company Affairs (From 14.2.1969 to 27.6.1970)
		Deputy Minister in the Ministries of Petroleum and Chemicals, and Mines and Metals (From 27.6.1970 to 31.8.1970)
9.	Bhonsle, Shri J.K.	Deputy Minister in the Ministry of Rehabilitation (From 12.8.1952 to 16.4.1957*)
10.	Buragohain, Shri S.N.	Deputy Minister in the Ministry of Works, Mines and Power (From 14.8.1950 to 13.5.1952)
		Deputy Minister in the Ministry of Works, Housing and Supply (From 13.5.1952 to 4.10.1953)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
11.	Chanda, Shri Anil Kumar	Deputy Minister in the Ministry of External Affairs (From 12.8.1952 to 17.4.1957) (Again from 17.4.1957 to 1.5.1957)
		Deputy Minister in the Ministry of Works, Housing and Supply (From 1.5.1957 to 10.4.1962*)
12.	Chaturvedi, Shri Roshanlal	Deputy Minister in the Department of Parliamentary Affairs (From 18.3.1967 to 14.11.1967)
		Deputy Minister in the Ministry of Railways (From 14.11.1967 to 18.3.1971)
13.	Chowdhary, Shri Desai	Deputy Minister in the Ministry of Health and Family Welfare (From 21.11.1990 to 21.6.1991*)
		Also Deputy Minister in the Ministry of Industry (From 28.11.1990 to 21.6.1991*)
14.	Das, Shri Bipinpal	Deputy Minister in the Ministry of External Affairs (From 17.10.1974 to 24.3.1977*)
15.	Das, Dr. Mono Mohan	Deputy Minister in the Ministry of Education (From 1.5.1956 to 17.4.1957*)
		Deputy Minister in the Ministry of Education and Scientific Research (From 17.4.1957 to 14.4.1958)
		Deputy Minister in the Ministry of Scientific Research and Cultural Affairs (From 14.4.1958 to 10.4.1962*)
		Deputy Minister in the Ministry of Scientific Research and Cultural Affairs (From 16.4.1962 to 21.11.1963)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Education (From 21.11.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Rehabilitation (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
16.	Dhankar, Shri Jagdeep	Deputy Minister in the Ministry of Parliamentary Affairs (From 23.4.1990 to 5.11.1990)
17.	Ering, Shri D.	Deputy Minister in the Ministry of Food, Agriculture, Community Development and Cooperation (From 18.3.1967 to 21.6.1970)
18.	Hansda, Shri Subodh Chandra	Deputy Minister in the Ministry of Steel and Mines (From 6.2.1973 to 10.10.1974)
19.	Himatsinghji, Major General	Deputy Minister in the Ministry of Defence (From 14.8.1950 to 29.2.1952)
20.	Iqbal Singh, Sardar	Deputy Minister in the Ministry of Petroleum and Chemicals (From 24.1.1966 to 13.3.1967*)
		Deputy Minister in the Ministry of Works, Housing and Supply (From 18.3.1967 to 14.2.1969)
		Deputy Minister in the Department of Parliamentary Affairs and in the Ministry of Shipping and Transport (From 14.2.1969 to 8.7.1970)
		Deputy Minister in the Ministry of Shipping and Transport (From 8.7.1970 to 18.3.1971*)
21.	Ishaque, Shri A.K.M.	Deputy Minister in the Ministry of Health and Family Planning (From 10.10.1974 to 24.3.1977*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
22.	Jahanara Jaipal Singh, Smt.	Deputy Minister in the Ministry of Transport and Civil Aviation (From 15.2.1966 to 13.3.1967*)
		Deputy Minister in the Ministry of Tourism and Civil Aviation (From 18.3.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Education and Youth Services (From 14.2.1969 to 27.6.1970)
23.	Jai Prakash, Shri	Deputy Minister in the Ministry of Petroleum and Chemicals (From 21.11.1990 to 21.6.1991*)
		Also Deputy Minister in the Ministry of Communications (From 7.12.1990 to 21.6.1991*)
24.	Jamir, Shri S.C.	Deputy Minister in the Ministry of Railways (From 18.3.1967 to 14.11.1967)
		Deputy Minister in the Ministry of Labour, Employment and Rehabilitation (From 14.11.1967 to 27.6.1970)
		Deputy Minister in the Ministry of Food, Agriculture, Community Development and Cooperation (From 27.6.1970 to 18.3.1971*)
25.	Joshi, Miss Kumudben M.	Deputy Minister in the Ministry of Information and Broadcasting (From 19.10.1980 to 15.1.1982)
		Deputy Minister in the Ministry of Health and Family Welfare (From 15.1.1982 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)
26.	Kamla Kumari, Miss	Deputy Minister in the Ministry of Agriculture (From 19.10.1980 to 29.1.1983)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Rural Reconstruction (From 24.11.1980 to 29.1.1983)
27.	Kamala Kumari, Smt. K.	Deputy Minister in the Ministry of Welfare (From 21.6.1991 to 17.1.1993)
28.	Kedar Nath Singh, Shri	Deputy Minister in the Department of Parliamentary Affairs (From 2.5.1971 to 10.10.1974)
		Deputy Minister in the Ministry of Agriculture and Irrigation (From 10.10.1974 to 24.3.1977*)
29.	Khursheed Lal, Shri	Deputy Minister in the Ministry of Communications (From 1.10.1948 to 26.1.1950*) (Again from 31.1.1950 to 6.5.1950*) (Again from 6.5.1950 to 29.9.1951)
30.	Kisku, Shri A.K.	Deputy Minister in the Ministry of Education and Youth Services (From 30.6.1970 to 18.3.1971*)
		Deputy Minister in the Ministry of Health and Family Planning (From 18.3.1971 to 10.10.1974)
31.	Krishna, Shri M.R.	Deputy Minister in the Ministry of Defence (From 16.11.1967 to 27.6.1970)
		Deputy Minister in the Ministry of Industrial Development and Internal Trade (From 27.6.1970 to 18.3.1971*)
32.	Kureel, Shri Baijnath	Deputy Minister in the Ministry of Irrigation and Power (From 2.5.1971 to 4.2.1973)
33.	Majhi, Shri C.P.	Deputy Minister in the Ministry of Petroleum and Chemicals (From 17.10.1974 to 3.1.1976)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Chemicals and Fertilisers (From 3.1.1976 to 24.3.1977*)
34.	Majithia, Sardar Surjit Singh	Deputy Minister in the Ministry of Defence (From 12.8.1952 to 17.4.1957*) (Again from 17.4.1957 to 10.4.1962*)
35.	Malviya, Shri Ratanlal Kishorilal	Deputy Minister in the Ministry of Labour and Employment (From 4.12.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
36.	Mehdi, Shri Syed Ahmed	Deputy Minister in the Ministry of Mines and Metals (From 15.2.1966 to 13.3.1967*)
37.	Misra, Shri Shyam Dhar	Deputy Minister in the Ministry of Community Development, Panchayati Raj and Cooperation (From 8.5.1962 to 16.2.1963)
		Deputy Minister in the Ministry of Community Development and Cooperation (From 16.2.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Irrigation and Power (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Defence (From 24.1.1966 to 14.2.1966)
		Deputy Minister in the Ministry of Food, Agriculture, Community Development and Cooperation (From 14.2.1966 to 13.3.1967*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
38.	Mohanty, Shri Brijmohan	Deputy Minister in the Ministry of Civil Supplies (From 19.10.1980 to 15.1.1982)
		Deputy Minister in the Ministry of Works and Housing (From 15.1.1982 to 29.1.1983)
39.	Mohiuddin Ahmed, Shri	Deputy Minister of Civil Aviation in the Ministry of Transport and Communications (From 2.4.1958 to 10.4.1962*)
		Deputy Minister in the Ministry of Transport and Communications (From 16.4.1962 to 1.9.1963)
		Deputy Minister in the Ministry of Transport (From 1.9.1963 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 5.1.1966)
40.	Mohsin, Shri F.H.	Deputy Minister in the Ministry of Home Affairs (From 2.5.1971 to 24.3.1977*)
41.	Naskar, Shri Purnendu Sekhar	Deputy Minister in the Ministry of Rehabilitation (From 2.4.1958 to 10.4.1962*)
		Deputy Minister in the Ministry of Works, Housing and Supply (From 16.4.1962 to 15.11.1962)
		Deputy Minister in the Ministry of Works and Housing (From 15.11.1962 to 5.12.1962)
		Deputy Minister in the Ministry of Works, Housing and Rehabilitation (From 5.12.1962 to 16.4.1964)
		Deputy Minister in the Ministry of Rehabilitation (From 16.4.1964 to 27.5.1964*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Health (From 1.10.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
		Deputy Minister in the Ministry of Home Affairs (From 24.1.1966 to 13.3.1967*)
42.	Nayak, Shri Nakul	Parliamentary Secretary in the Prime Minister's Office (From 25.11.1990 to 21.6.1991*)
43.	Nyamagoundar, Shri S.B.	Deputy Minister in the Ministry of Coal (From 26.6.1991 to 17.1.1993)
44.	Panjahazari, Shri Raghbir Singh	Deputy Minister in the Department of Parliamentary Affairs (From 30.6.1970 to 18.3.1971*)
45.	Parthasarathy, Shri P.	Deputy Minister in the Department of Parliamentary Affairs (From 30.6.1970 to 18.3.1971*)
46.	Patel, Shri Ahmed	Parliamentary Secretary to the Prime Minister (From 1.1.1985 to 25.9.1985)
47.	Patel, Shri Prabhu Das	Deputy Minister in the Ministry of Agriculture and Irrigation (From 23.10.1974 to 14.3.1977)
48.	Patel, Shri Shantilal P.	Deputy Minister in the Ministry of Commerce (From 21.11.1990 to 23.4.1991)
49.	Patil, Shri Vijay N.	Deputy Minister in the Department of Science, Technology and Space (From 8.6.1980 to 19.10.1980)
		Deputy Minister in the Ministry of Communications (From 19.10.1980 to 31.10.1984*) (Again from 4.11.1984 to 31.12.1984*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
50.	Raju, Dr. D.S.	Deputy Minister in the Ministry of Health (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Defence (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
51.	Ramachandran, Dr. (Smt.) T.S. Soundaram	Deputy Minister in the Ministry of Education (From 16.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*) (Again from 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)
52.	Ramaswamy, Shri K.S.	Deputy Minister in the Ministry of Home Affairs (From 18.3.1967 to 18.3.1971*)
		Also Deputy Minister in the Department of Personnel in Cabinet Secretariat (From 23.8.1970 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Deputy Minister in the Ministry of Education and Social Welfare (From 2.5.1971 to 11.11.1972)
53.	Ramaswamy, Shri S.V.	Deputy Minister in the Ministry of Railways (From 2.4.1958 to 10.4.1962) (Again from 16.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Commerce (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
54.	Rao, Shri J.B. Muthyal	Deputy Minister in the Department of Parliamentary Affairs (From 18.3.1967 to 14.11.1967)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Department of Social Welfare (From 14.11.1967 to 14.2.1969)
		Also Deputy Minister in the Ministry of Petroleum and Chemicals (From 17.11.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Law and in the Department of Social Welfare (From 14.2.1969 to 17.10.1969)
55.	Rao, Dr. M. Sanjeevi	Deputy Minister in the Department of Electronics (From 15.1.1982 to 31.10.1984*)
		Also Deputy Minister in the Ministry of Food and Civil Supplies (From 26.7.1983 to 31.10.1984*)
		Deputy Minister in the Department of Electronics (From 4.11.1984 to 31.12.1984*)
56.	Rao, Shri M. Thirumala	Deputy Minister in the Ministry of Food and Agriculture (From 21.8.1950 to 13.5.1952*)
57.	Roy, Shri Bishwanath	Deputy Minister in the Ministry of Labour, Employment and Rehabilitation (From 30.6.1970 to 18.3.1971*)
58.	Sahu, Shri Rameshwar	Deputy Minister in the Ministry of Finance (From 1.10.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*)
59.	Saleem, Shri Mohammad Yunus	Deputy Minister in the Ministry of Law (From 16.11.1967 to 14.2.1969)
		Deputy Minister in the Ministry of Law in the Department of Social Welfare (From 14.2.1969 to 27.6.1970)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Railways (From 27.6.1970 to 18.3.1971*)
60.	Satish Chandra, Shri	Deputy Minister in the Ministry of Defence (From 27.11.1952 to 14.7.1955)
		Deputy Minister in the Department of Defence Production (From 14.7.1955 to 17.4.1957*)
		Deputy Minister in the Ministry of Commerce and Industry (From 17.4.1957 to 10.4.1962*)
61.	Sham Nath, Shri	Deputy Minister in the Ministry of Information and Broadcasting (From 8.5.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
		Deputy Minister in the Ministry of Railways (From 15.6.1964 to 11.1.1966*) (Again from 11.1.1966 to 24.1.1966*) (Again from 24.1.1966 to 13.3.1967*)
62.	Shastri, Shri Anil	Deputy Minister in the Ministry of Finance (From 23.4.1990 to 10.11.1990*)
63.	Sidheshwar Prasad, Shri	Deputy Minister in the Ministry of Irrigation and Power (From 13.11.1967 to 18.3.1971*) (Again from 18.3.1971 to 2.5.1971)
		Deputy Minister in the Ministry of Industrial Development (From 2.5.1971 to 5.2.1973)
		Deputy Minister in the Ministry of Heavy Industry (From 5.2.1973 to 9.11.1973)
		Deputy Minister in the Ministry of Irrigation and Power (From 9.11.1973 to 10.10.1974)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
		Deputy Minister in the Ministry of Energy (From 10.10.1974 to 24.3.1977*)
64.	Sinha, Shri Dharam Bir	Deputy Minister in the Ministry of Information and Broadcasting (From 2.5.1971 to 24.3.1977*)
65.	Sinha, Shrimati Tarkeshwari	Deputy Minister in the Ministry of Economic Affairs (From 2.4.1958 to 10.5.1958)
		Deputy Minister in the Ministry of Finance (From 10.5.1958 to 10.4.1962*) (Again from 16.4.1962 to 27.5.1964*) (Again from 27.5.1964 to 9.6.1964*)
66.	Sukhdev Prasad, Shri	Deputy Minister in the Ministry of Steel and Mines (From 5.2.1973 to 24.3.1977*)
67.	Venkata Reddy, Shri P.	Deputy Minister in the Ministry of Industry (From 8.6.1980 to 19.10.1980)
		Deputy Minister in the Ministry of Labour (From 19.10.1980 to 15.1.1982)
68.	Verma, Shri Balgovind	Deputy Minister in the Ministry of Labour and Rehabilitation (From 2.5.1971 to 5.2.1973)
		Deputy Minister in the Ministry of Irrigation and Power (From 5.2.1973 to 9.11.1973)
		Deputy Minister in the Ministry of Labour (From 9.11.1973 to 23.12.1976)
		Deputy Minister in the Ministry of Communications (From 23.12.1976 to 24.3.1977*)

^{*}The date of dissolution of the Council of Ministers.

Sl.No.	Name	Portfolios alongwith the period thereof
69.	Yadav, Prof. D.P.	Deputy Minister in the Ministry of Education and Social Welfare (From 2.5.1971 to 28.5.1971)
		Deputy Minister in the Ministry of Education and Social Welfare and in the Department of Culture (From 28.5.1971 to 24.3.1977*)

^{*}The date of dissolution of the Council of Ministers.

ANNEXURE

LIST OF COUNCIL OF MINISTERS

(As on 28.08.2015)

Shri Narendra Modi **Prime Minister** and also in-charge of:

Ministry of Personnel, Public Grievances and Pensions;

Department of Atomic Energy; Department of Space; and

All important Policy issues and all other portfolios

not allocated to any Minister.

CABINET MINISTERS

1.	Shri Raj Nath Singh	Minister of Home Affairs.
2.	Smt. Sushma Swaraj	Minister of External Affairs; and Minister of Overseas Indian Affairs.
3.	Shri Arun Jaitley	Minister of Finance; Minister of Corporate Affairs; and Minister of Information and Broadcasting.
4.	Shri M. Venkaiah Naidu	Minister of Urban Development; Minister of Housing and Urban Poverty Alleviation; and Minister of Parliamentary Affairs.
5.	Shri Nitin Jairam Gadkari	Minister of Road Transport and Highways; and Minister of Shipping.
6.	Shri Manohar Parrikar	Minister of Defence.
7.	Shri Suresh Prabhu	Minister of Railways.
8.	Shri D.V. Sadananda Gowda	Minister of Law and Justice.
9.	Sushri Uma Bharati	Minister of Water Resources, River Development and Ganga Rejuvenation.
10.	Dr. Najma A. Heptulla	Minister of Minority Affairs.
11.	Shri Ramvilas Paswan	Minister of Consumer Affairs, Food and Public Distribution.

238		Council of Ministers (1947—2015)
12.	Shri Kalraj Mishra	Minister of Micro, Small and Medium Enterprises.
13.	Smt. Maneka Sanjay Gandhi	Minister of Women and Child Development.
14.	Shri Ananthkumar	Minister of Chemicals and Fertilizers.
15.	Shri Ravi Shankar Prasad	Minister of Communications and Information Technology.
16.	Shri Jagat Prakash Nadda	Minister of Health and Family Welfare.
17.	Shri Ashok Gajapathi Raju Pusapati	Minister of Civil Aviation.
18.	Shri Anant Geete	Minister of Heavy Industries and Public Enterprises.
19.	Smt. Harsimrat Kaur Badal	Minister of Food Processing Industries.
20.	Shri Narendra Singh Tomar	Minister of Mines; and Minister of Steel.
21.	Shri Chaudhary Birender Singh	Minister of Rural Development; Minister of Panchayati Raj; and Minister of Drinking Water and Sanitation
22.	Shri Jual Oram	Minister of Tribal Affairs.
23.	Shri Radha Mohan Singh	Minister of Agriculture and Farmers Welfare.
24.	Shri Thaawar Chand Gehlot	Minister of Social Justice and Empowerment.
25.	Smt. Smriti Zubin Irani	Minister of Human Resource Development.
26.	Dr. Harsh Vardhan	Minister of Science and Technology; and Minister of Earth Sciences.

MINISTERS OF STATE (INDEPENDENT CHARGE)

1. General (Retd.) V.K. Singh Minister of State (Independent Charge) of the Ministry of Statistics and Programme Implementation; Minister of State in the Ministry of External Affairs; and Minister of State in the Ministry of Overseas Indian Affairs.

	,	
2.	Rao Inderjit Singh	Minister of State (Independent Charge) of the Ministry of Planning; and Minister of State in the Ministry of Defence.
3.	Shri Santosh Kumar Gangwar	Minister of State (Independent Charge) of the Ministry of Textiles.
4.	Shri Bandaru Dattatreya	Minister of State (Independent Charge) of the Ministry of Labour and Employment.
5.	Shri Rajiv Pratap Rudy	Minister of State (Independent Charge) of the Ministry of Skill Development and Entrepreneurship; and Minister of State in the Ministry of Parliamentary Affairs.
6.	Shri Shripad Yesso Naik	Minister of State (Independent Charge) of the Ministry of Ayurveda, Yoga and Naturopathy, Unani, Siddha and Homoeopathy (AYUSH); and Minister of State in the Ministry of Health and Family Welfare.
7.	Shri Dharmendra Pradhan	Minister of State (Independent Charge) of the Ministry of Petroleum and Natural Gas.
8.	Shri Sarbananda Sonowal	Minister of State (Independent Charge) of the Ministry of Youth Affairs and Sports.
9.	Shri Prakash Javadekar	Minister of State (Independent Charge) of the Ministry of Environment, Forests and Climate Change.
10.	Shri Piyush Goyal	Minister of State (Independent Charge) of the Ministry of Power; Minister of State (Independent Charge) of the Ministry of Coal; and Minister of State (Independent Charge) of the Ministry of New and Renewable Energy.
11.	Dr. Jitendra Singh	Minister of State (Independent Charge) of the Ministry of Development of North Eastern Region; Minister of State in the Prime Minister's Office;

Minister of State in the Ministry of Personnel, Public Grievances and

Pensions;

Minister of State in the Department of

Atomic Energy; and

Minister of State in the Department of

Space.

12. Smt. Nirmala Sitharaman Minister of State (Independent Charge)

of the Ministry of Commerce and

Industry.

13. Dr. Mahesh Sharma Minister of State (Independent Charge)

of the Ministry of Culture;

Minister of State (Independent Charge)

of the Ministry of Tourism; and

Minister of State in the Ministry of Civil

Aviation.

MINISTERS OF STATE

1. Shri Mukhtar Abbas Naqvi Minister of State in the Ministry of

Minority Affairs; and

Minister of State in the Ministry of

Parliamentary Affairs.

2. Shri Ram Kripal Yadav Minister of State in the Ministry of

Drinking Water and Sanitation.

3. Shri Haribhai Parthibhai

Chaudhary

Minister of State in the Ministry of Home

Affairs.

4. Shri Sanwar Lal Jat Minister of State in the Ministry of Water

Resources, River Development and

Ganga Rejuvenation.

5. Shri Mohanbhai Kalyanjibhai Kundariya Minister of State in the Ministry of Agriculture and Farmers Welfare.

Shri Giriraj Singh Minister of State in the Ministry of Micro, 6.

Small and Medium Enterprises.

7. Shri Hansraj Gangaram

Ahir

Minister of State in the Ministry of

Chemicals and Fertilizers.

Shri G.M. Siddeshwara 8.

Minister of State in the Ministry of Heavy

Industries and Public Enterprises.

9. Shri Manoj Sinha Minister of State in the Ministry of

Railways.

10.	Shri Nihalchand	Minister of State in the Ministry of Panchayati Raj.
11.	Shri Upendra Kushwaha	Minister of State in the Ministry of Human Resource Development.
12.	Shri Radhakrishnan P.	Minister of State in the Ministry of Road Transport and Highways; and Minister of State in the Ministry of Shipping.
13.	Shri Kiren Rijiju	Minister of State in the Ministry of Home Affairs.
14.	Shri Krishan Pal	Minister of State in the Ministry of Social Justice and Empowerment.
15.	Dr. Sanjeev Kumar Balyan	Minister of State in the Ministry of Agriculture and Farmers Welfare.
16.	Shri Mansukhbhai Dhanjibhai Vasava	Minister of State in the Ministry of Tribal Affairs.
17.	Shri Vishnu Deo Sai	Minister of State in the Ministry of Mines; and Minister of State in the Ministry of Steel.
18.	Shri Sudarshan Bhagat	Minister of State in the Ministry of Rural Development.
19.	Prof. (Dr.) Ram Shankar Katheria	Minister of State in the Ministry of Human Resource Development.
20.	Shri Y.S. Chowdary	Minister of State in the Ministry of Science and Technology; and Minister of State in the Ministry of Earth Sciences.
21.	Shri Jayant Sinha	Minister of State in the Ministry of Finance.
22.	Col. Rajyavardhan Singh Rathore	Minister of State in the Ministry of Information and Broadcasting.
23.	Shri Babul Supriyo	Minister of State in the Ministry of Urban Development; and Minister of State in the Ministry of Housing and Urban Poverty Alleviation.
24.	Sadhvi Niranjan Jyoti	Minister of State in the Ministry of Food Processing Industries.
25.	Shri Vijay Sampla	Minister of State in the Ministry of Social Justice and Empowerment.

Note: Incorporates changes in the Council of Ministers as on 05.03.2015.

NAME INDEX

Name	Page
A	
Abdullah, Dr. Farooq	24
Abdullah, Omar	149
Abid Ali	223
Adhikari, Sisir	149
Adsul, Anandrao Vithoba	149
Advani, L.K	21
Aggarwal, Satish	149
Ahir, H.G.	240
Ahluwalia, S.S.	150
Ahmad, Shaqeel	150
Ahmad, Abrar	150
Ahmed, E.	149
Ahmed, F.A.	24
Ahmed, Sultan	151
Ahmed, Kamaluddin	150
Aiyer, Mani Shankar	24
Ajay Singh	223
Ajit Singh	
Alageshan, O.V.	151
Alagh, Yogendra	127
Alagiri, M.K.	25
Alam, Rafique	127
Alva, Margaret	151
Alva, Violet	223
Ambareesh, M.H.	152
Ambedkar, B.R.	25
Amrit Kaur, Rajkumari	
Ananth Kumar	25, 238
Anjiah, T	
Ansari, Z.R.	
Antony, A.K.	27
Antulay, A.R.	
Apang, Omar	
Arif, Mohammad Usman	223
Arjun Singh	27

Name	Page
Arun Singh	152
Arunachalam, M	28
Ashwini Kumar	63
Austin, Henry	152
Ayyangar, N. Gopalaswami	28
Azad, Bhagwat Jha	128
Azad, Ghulam Nabi	29
Azad, Maulana Abul Kalam	28
В	
Baalu, T.R.	30
Badal Harsimrat K	238
Badal, P.S.	30
Badal, Sukhbir Singh	152
Bahuguna, H.N.	30
Bais, Rames	128
Baishya, Birendra Prasad	31
Baitha, D.L.	152
Bagrodia, Santosh	152
Bajpayee, Rajendra Kumari	129
Balasubramanian, S.R.	152
Baldev Singh, Sardar	31
Baleshwar Ram	153
Balyan, S.K.	241
Benerjee, Mamata	31
Bandyopadhayay, Sudip	153
Bansal, P.K.	31
Bansi Lal	32
Barnala, Surjit Singh	32
Barooah, Dev Kant	32
Barot, Magan Bhai	223
Bartakaki, Renuka Devi	153
Barua, Bedabrata	153
Basappa, Kondajji	223
Basavrajeshwari	153
Beg, Arif	153
Behra, Bhajma	
Bhabha, C.H.	32
Bhagat, B.R.	
Bhagat H K I	34

Name Page
Bhagat, Sudarshan
Bhagawati B.C
Bhajan Lal 34 Bhakt Darshan 154
Bhanu Pratap Singh
Bhardwaj, H.R
Bhatia, R.L
Bhoi, Krupasindhu
Bhonsle J.K. 224
Bhuria, Kanti Lal
Bir Sen
Biswas, C.C
Bommai, S.R
Brahm Prakash
Buragohain, S.N
Buta Singh
С
Chagla, M.C
Chakravarty, Bijoya
Chanana, Charanjit
Chand Ram
Chanda, Anil Kumar
Chandrakar, Chandulal
Chandra Shekhar
Chandra Shekhar M
Chandrasekhar, S
Charan Singh
Chattopadhyay, D.P
Chaturvedi, Bhuvnesh
Chaturvedi, Roshanlal
Chaudhary, H.P
Chaudhary, Tusharbhai
Chaudhuri, A.B.A. Ghani Khan
Chaudhuri, Sachindra
Chaudhury, Dalbir Singh
Chaudhury, M.J
Chaudhury, Rashida Haque
Chauhan, Shriram
Chavan, D.R

Name	Page
Chavan, Prithviraj	129
Chavan, S.B.	38
Chavan, Y.B.	21
Chetty, R.K. Shanmukham	38
Chidambaram, P	
Chikhalia, Bhavanaben Devrajbhai	158
Chiranjeevi, K	
Choudhary, A.H. Khan	
Choudhury, M.H.	40
Choudhury, N.K.	158
Chowdhary, Desai	225
Chowdhary, Y.S.	241
Chowdhury, A.R.	
Chowdhury, Renuka	
Chowdhury, S	
Chunder, Pratap Chandra	
D	
Dalbir Singh	150
Dandavate, Madhu	
Dar, Mohammad Magbool	
Das, Akhilesh	
Das, Bhakta Charan	
Das, Bipin Pal	
Dasmunsi, Deepa	
Das, M.M.	
Dasmunshi, Priyaranjan	
Dassappa, H.C	
Datar, B.N.	
Dattatreya, Bandaru	,
Deb, Santosh Mohan	
Deo, V. Kishore Chandra S.	
Deora, Milind	
Deora, Murali	
Desai, Hitendra	
Desai, Khandubhai Kasanji	
Desai, Morarji R.	
Deshmukh, C.D.	
Deshmukh, P	
Deshmukh Vilasrao	42

Name	Page
Deve Gowda, H.D.	3
Devi, Lal	22
Dey, Surendra Kumar	124
Dhakne, Baban Rao	159
Dhananjaya Kumar, V	159
Dhankar Jagdeep	226
Dhanushkodi Athithan, R	160
Dhar, D.P.	42
Dharam Vir	160
Dharia Mohan	42
Dhawan, Harmohan	131
Dhawan, R.K.	131
Dhillon, G.S.	
Dhindsa, S.S.	43
Digvijay, Singh	160
Dikshit, Shiela	
Dikshit, Uma Shankar	
Dinesh Singh	
Diwakar, R.R.	
Dubey, Bindeshwary	
Dutt, Brahm	
Dutt, Sunil	
E	
Elangovan, E.V.K.S.	161
Engti, B.S.	
Ering, D.	
-	
Ering, Ninong	
,	131
F	
Faleiro, Eduardo	162
Farook, M.O.H.	162
Fatmi, A.A.	163
Fazlur Rahman	44
Fernandes, George	45
Fernandes, Oscar	45
Fotedar, M.L.	
G	
Gadgil, N.V.	46
Gadgil, V.N.	

Name	Page
Gadhvi, B.K.	163
Gadkari, Nitin J	237
Gafoor, Abdul	46
Gaikwad Patil, J	163
Gamang, G. Giridhar	132
Gandhi, D.K.M.	163
Gandhi, Indira	5
Gandhi, Maneka	133, 238
Gandhi, Rajiv	3
Gandhi, Selvan S	163
Ganesh, K.R.	163
Gangwar, S.K.	164, 239
Gautam, S.P.	133
Gavit, Manikrao	164
Geete, A.G	47, 238
Gehlot, Ashok	133
Gehlot, T.C.	238
George, A.C.	164
Ghatowar, Paban Singh	134
Ghose, Sanker	165
Ghosh, Parimal	165
Gill, M.S	47
Giri, V.V	47
Gobardhan Bhagey	165
Goel, Vijay	165
Gogoi, Tarun	134
Gokhale, H.R.	47
Gopal, K	166
Goswami, Dinesh	47
Gowda, D.V.S.	237
Goyal, K.K.	166
Goyal, Piyush	239
Goyal, Vedprakash	47
Guha, Arun Chandra	125
Guha, Phulrenu	166
Gujral, I.K.	7
Gulshan, Dhanna Singh	166
Gupta, Indrajit	
Gupta, Chaman Lal	166
Gupta, Prem Chand	48
Gurupadaswamv. M.S.	48

Name	Page
н	
Hajarnavis, R.M.	166
Handique, B.K.	48
Hansda, Soubodh Chandra	226
Hanumanthaiya, K	49
Harsh, Vardhan	238
Hathi, Jai Sukh Lal	49
Hedge, Ramakrishna	50
Heptulla, Nazma A.	237
Himatsinhji	226
Hussain, Sarwar	167
Hussain, Syed Shahnawaz	50
1	
•	
Ibrahim, C.M.	
Ibrahim, Hafiz Mohammad	
Iqbal Singh, Sardar	
Irani, Smriti Z	
Ishaque, A.K.M.	226
J	
Jacob, M.M	167
Jaffer Sharief, C.K.	51
Jagadeesan, Subbulakshmi	168
Jagathrakshakan, S	168
Jagbir Singh	168
Jagjivan Ram	22
Jagmohan	51
Jahanara, Jaipal Singh	227
Jai Prakash	227
Jaichandra Singh, R.K.	168
Jain, Ajit Prasad	52
Jain, Pradeep	168
Jairam Das Daulat Ram	52
Jaiswal, Shriprakash	52
Jaitley, Arun	53, 237
Jakhar, Balram	53
Jalappa, R.L.	53
Jamir, S.C.	227
Janakrishnamurthy, K	54

Name	Page
Janarthanan Kadambur, R	169
Jaswant Singh	54
Jat, S.L	240
Jatiya, S.N.	54
Javadekar, Prakash	239
Jena, Srikant	135
Jethmalani, Ram	54
Jha, Ragunath	169
Joshi, Kumudben M	227
Joshi, C.P.	55
Joshi, Murli Manohar	55
Joshi, Manohar	56
Jual Oram	56
Judev, Dilip Singh	169
Jyoti, Niranjan	241
К	
Kabir, Humayun	56
Kacker, S.N.	57
Kakmadi, Suresh	169
Kalvi, Kalyan Singh	57
Kamal Nath	57
Kamala Kumari, K	228
Kamat, Gurudas	135
Kamla Kumari	227
Kamson, Meijinlung	169
Kannappan, M	136
Kanti Singh	169
Kanungo, Nityanand	170
Karan Singh	57
Karmarkar, D.P.	171
Karunakaran, K	57
Kataria, L.C.	171
Katheria R.S.	241
Kathiria, Vallabhbhai	171
Katju, K.N.	58
Katoch, C.K.	58
Kaul, Shiela	58
Kaur, Preneet	171
Kasuhal Jagannath	58

Name Page	e
Kaushik, Purushottam5	8
Kedar Nath Singh	8
Keskar, B.V	2
Kesri, Sitaram5	9
Khadikar, R.K	'2
Khalap, Ramakant D	6
Khan, Arif Mohammad5	
Khan, Aslam Sher	′3
Khan, Khurshid Alam	6
Khan, K. Rahman	9
Khan, Mohd. Ayub	′3
Khan, Shah Nawaz	
Khandela, M.S	
Khanduri, B.C. 6	
Khanna, H.R	
Khanna, Mehar Chand	
Khanna, Vinod	
Khaparde, Saroj	
Kharge, Mallikarjun6	
Khurana, M.L	
Khursheed, Salman	
Khurseed Lal	
Kidwai, Mohsina	
Kidwai,Rafi Ahmed	
Killi, Kruparani	
Kisku, A.K	
Kotadia Manubhai	
Krishan, M.R	
Krishna, S.M	
Krishna Kumar, S	
Krishnamachari, T.T	
Krishnan, G.Y	
Krishnappa, M.V	
Kulaste, Faggan Singh	
Kumaramangalam, P.R	
Kumaramangalam, S. Mohan	
Kundariya, M.K	
Kundu, Samarendra	
Kureel Baijnath	
Kurien, P.J	

Name	Page
Kushwaha, Upendra	241
Kyndiah, P.R.	65
L	
	474
Lakshmi, P.	
Laskar, N.R.	
Laxman, Bangaru	
Lenka, K.C.	1//
M	
Mahavir Prasad	66
Mahajan, Promod	65
Mahajan, Sumitra	177
Mahajan, Vikram	177
Mahant, Charan Das	178
Maharia, Subhash	178
Mahida, Narendra Singh	178
Mahishi, Sarojini	178
Maiti, Abha	180
Majhi, C.P.	228
Majithia, Surjit Singh	229
Makan, Ajay	66
Makwana, Yogendra	178
Malviya, K.D.	67
Malaviya, Ratanlal Kishori Lal	229
Malaviya, Radhakrishna	179
Malaviya, Satya Prakash	67
Malik, Satya Pal	179
Mallikarjun, K	179
Mandal, Dhanik Lal	180
Manmohan Singh	8
Maran, Dayanidhi	68
Maran, Murasoli	68
Marandi, Babulal	180
Masood, Rasheed	137
Matang Singh	181
Mathai, John	68
Maurya, B.P.	181
Meena, Jas Kaur	181
Meena, Namo Narain	181

Name Pa _i	ge
Meghwal, Kailash	81
Mehdi, Syed Ahmed	29
Mehta, Ashoka	68
Mehta, Chiman Bhai	81
Mehta, Jayawanti 1	82
Mehta, Om 1	81
Mehtab, Hare Krishna	69
Meira Kumar	69
Menon, Lakshmi N	82
Menon, M.G.K	82
Menon, P. Govinda	69
Menon, V.K. Krishna	69
Mir Qasim, Syed	70
Mirdha, Nathuram	70
Mirdha, Ram Niwas	70
Mishra, B	82
Mishra, Chaturanan	71
Mishra, G.S	83
Mishra, Harinath 1	83
Mishra, Jagannath	71
Mishra, Janeshwar	72
Mishra, Kalraj	238
Mishra, L.N.	72
Mishra, S.N.	73
Misra, S.D	29
Modi, Narendra 2	237
Mohan, Chinta 18	83
Mohanty, Brijmohan	230
Mohite, Subodh	73
Mohiuddin Ahmed	230
Mohsim, F.H	30
Moily, M. Veerappa	73
Mookerjee, S.P.	74
Mookherjee, Satya Brata 18	84
Moorthy, A.K	
Morarka, Kamal 1	84
Muhammad, V.A. Seyid	
Mukherjee, Pranab	
Munda, Kariya	
Muni Lall	84

Name	Page
Muniyappa, K.H.	137
Munshi, K.M	75
Murthy, B.S.	184
Murthy, M. Rajasekara	137
Murthy, M.V. Chandrasekhara	185
Muthaiah, R	75
Muthu, Sathiavani	75
Muttemwar, Vilas Baburao	137
N	
Nadda, J.P.	238
Nagmani	185
Naidu, Venkaiah M	75, 237
Naidu, Yerran K.	
Naik, P.B.	
Naik, Ram	
Naik, Shripad Yesso	186, 239
Namgyal, P	186
Nanda, Gulzari Lal	
Napoleon, D.	186
Naqvi, Mukhtar Abbas	
Narah, Ranee	187
Narasimha Rao, P.V.	11
Narayanan, K.R.	187
Narayan Swamy, V	187
Narendra, A.	187
Naskar, P.S.	230
Natchiappan, E.M.S.	188
Natarajan, Jayanthi	138
Natwar Singh K	76
Nayak, Nakul	231
Nayar, Sushila	188
Nehru, Arun K.	77
Nehru, Jawaharlal	14
Neogy, K.C.	77
Netam, Arvind	188
Nihalchand.	241
Nishad, J.N	138
Nitiraj Singh, Choudhary	188
Nitish Kumar	77

Name	Page
Nurul Hasan, S.	135
Nyamagounder, S.B	231
0	
Ola, Sis Ram	78
Oram, Jual	238
Oraon, Kartik	188
Oraon, Rameshwar	189
Oraon, Sumati	189
Oza, Ghanshyam	189
P	
Pachouri, Suresh	189
Pahadia, Jagannath	189
Pai, T.A	78
Pajnor, A. Bala	79
Pal, Debi Prosad	190
Pal, Krishan	241
Pala, Vincent	190
Palanimanickam, S.S.	190
Pallamraju, M.M.	79
Pandey, Kedar	79
Pandey, Raj Mangal	79
Panigrahi, Chintamani	190
Panja, Ajit Kumar	191
Panjahazari, R.S.	231
Pant, Govind Ballabh	79
Pant, Krishna Chandra	79
Parrikar, Manohar	237
Parthasarthy, P	231
Paswan, Ram Vilas	80, 237
Paswan, Sanjay	191
Pataskar, H.V.	126
Patel, Ahmed	231
Patel, A.K.	139
Patel, Dinsha J.	81
Patel, H.M.	81
Patel, Prabhu Das	231
Patel, Praful	82
Patel, Prahlad Singh	192

Name Page
Patel, Rampoojan
Patel, Sardar Vallabh Bhai
Patel, Shantilal P
Patel, Urmilaben Chimanbhai
Patel, Uttambhai H
Pathak, GS
Pathak, Harin
Patil, Annasaheb
Patil, Babagounda
Patil, Balasaheb Vikhe82
Patil, Basangouda
Patil, Basavraj
Patil, Prateek P
Patil, S.D
Patil, S.K
Patil, Shivraj V83
Patil, Suryakanta
Patil, Veerendra
Patil, Vijay N
Patnaik, Biju84
Patnaik, Janaki Ballabh84
Patnaik, Naveen
Pattabhi Rama Rao, S.B
Pattabhi Raman, C.R
Patwa, S.L84
Pawar, Sharad85
Pilot, Rajesh
Pilot, Sachin
Poojari, Janardhana
Poonacha, C.M85
Potdhuke, S
Prabhu, Rajendra
Prabhu, Suresh Prabhakar
Pradhan, Ashok Kumar
Pradhan, Debendra
Pradhan, Dharmendra
Prasad, Ravi Shankar
Prasad, V. Sreenivasa
Prasada, Jatin

Name Page
Punnuswamy, E
Purkayastha, Kabindra
Purandeswari, D
Pusapati, A.G.R
Q
Qureshi, Mohammad Shafi
R
Rachaiah, R
Radhakrishnan, P
Raghuramaiah, Kotha
Rahi, Ram Lal
Rahim, A.A
Rai, Kalpnath
Raja, A
Rajagopal, O
Raj Bahadur
Raj Narain
Rajagopalachari, C
Rajashikharan, M.V
Raje, Vasundhara
Rajendra Prasad90
Rajnath Singh
Raju, D.S
R.K. Kumar
Ram Kinkar90
Ram Sewak, Chaudhury
Ram Singh
Ramachandran, Gingee N
Ramachandran, Mullapally
Ramachandran, P
Ramachandran, T.S. Soundaram
Ramaiah, B.B
Ramamurthy, K
Raman
Ramanna, Raja
Ramaswamy, K.S
Ramaswamy, S.V
Ramdoss, Anbumani91

Name Pag	ge
Ramesh, Jairam	91
Ramoowalia, Balwant Singh	91
Rana, Kashi Ram	92
Rana, M.B	99
Rangayya Naidu, P.V 11	99
Rao, C.V	00
Rao, Dasari Narayan	00
Rao, Indrajit Singh201,23	39
Rao, J.B. Muthyal	32
Rao, J. Vengla	92
Rao, Jagannath20	.01
Rao, K.L	00
Rao, K. Chandra Shekhar	92
Rao, K.S.S.	92
Rao, M. Sanjeevi	33
Rao, M. Thirumala	33
Rao, P.A.P	00
Rao, S.B.P.B.K.S	
Rao, V.K.R.V.	92
Rath, R.C	02
Rathore, R.S	41
Rathwa, Naranbhai	02
Rawat, Bachi Singh	02
Rawat, Harish	92
Ray, Dilip Kumar	42
Ray, Rabi	
Ray, S.S.	93
Reddi, B. Gopala	93
Reddy, J.S	
Reddy, K. Brahmananda	
Reddy, K.C.	
Reddy, K.J.S.P	
Reddy, K.V. Raghunatha	
Reddy, K. Vijaya Bhaskara	
Reddy, M. Chenna	
Reddy, N. Sanjiva	
Reddy, T.S	
Regupathy, S	
Rijiju, Kiren	
Rohtagi, Sushila	

Name	Page
Routray, Nilamani	96
Roy, Bishwanath	233
Roy, Mukul	96
Roy, Saugata	204
Rudi, Rajiv Pratap	142, 239
S	
Sahay, Subodh Kant	96
Sahi, Krishna	
Sahi, L.P.	204
Sahu, Chandrasekhar	
Sahu, Rameshwar	
Sai, Larang	
Sai, Narhari Prasad Sukhdeo	
Sai, V.D.	
Saikia, Muhi Ram	
Sai Pratap, A.	
Saksena, Mohan Lal	
Saleem, Mohammad Yunus	
Salve, N.K.P.	
Sampla, Vijay	
Sangma, A.	
Sangma, P.A.	
Sanjivayya, D.	
Sanathanam, K.	
Saran, Daulat Ram	
Sartaj Singh	
Sathe, Vasant	
Satyanarayana, Sarvey	
Satish, Chandra	
Satpalji, Maharaj	
Satpathy, Nandini	
Savanoor, R.D.	
Sayeed, Mufti Mohammed	
Sayeed, P.M.	
Scindia, Jyotiraditya	
Scindia, Madhav Rao	
Seelam, J.	
Selja	
Selvi, V. Radhika	

Name Page
Sen, Asoke K
Sen, Triguna
Sethi, Arjun Charan
Sethi, P.C.B
Shah, Jayantilal V
Shah, K.K
Shah, M.C
Shah, Manubhai
Shakeel-ur-Rehman
Sham Nath
Shankaranand, B
Shanmugam, N.T
Shanta Kumar
Shanti Bhushan
Sharma, Anand
Sharma, A.P
Sharma, Satish
Sharma, Mahesh
Sharma, Nawal Kishore
Sharma, S.D
Sharma, Venod
Shastri, Anil
Shastri, Bhola Paswan
Shastri, Hari Krishna
Shastri, Lal Bahadur
Sheo Narain
Sher Singh
Shervani, Saleem Iqbal
Shinde, A.P
Shinde, S.K
Shiv Shankar, P
Shourie, Arun
Shrangare, T.S
Shreedharan, Arangil
Shrimali, K.L
Shukla, Rajeev
Shukla, V.C
Sibal, Kapil
Sibtey Razi, Syed

Name Pag	ge
Sidheswar Prasad	 34
Siddeshwara, G.M	40
Sidhwa, R.K	09
Sikandar Bakht	09
Sikdar, Tapan	09
Silvera, C	10
Singh, Akhilesh	10
Singh, Bhanu Pratap21	10
Singh, Bhishma Narain	10
Singh, Bir Bahadur	10
Singh, C.S	44
Singh, Chattrapal	11
Singh, C.P.N	10
Singh, Giriraj24	40
Singh, Hari Kishore	11
Singh, Jitendra	39
Singh, Lalit Vijoy21	12
Singh, R.P.N	12
Singh, Radha Mohan23	
Singh, Raghuvansh Prasad	09
Singh, Ram Bahadur21	12
Singh, Ram Subhag	09
Singh, Rao Birendra	38
Singh, Sanjay14	44
Singh, Surendra Pal21	12
Singh, T.N	26
Singh, Usha	12
Singh, Virbhadra1	11
Singh, V.K	38
Singh, Vishwanath Pratap1	16
Singh Deo, Shri K.P.	43
Sinha, Dharam Bir	35
Sinha, Jayant	41
Sinha, Kamla	13
Sinha, Manoj	40
Sinha, Ram Dulari	13
Sinha, Ram Kripal	13
Sinha, S.N	11
Sinha, Shatrughan	12

Name Page	e
Sinha, Tarkeshwari	35
Sinha, Yashwant	2
Sisodia, S.S	3
Sitharaman, N	10
Solanki, Bharatsinh	15
Solanki, Madhavsinh	2
Sompal	4
Somu, N.V.N	4
Soni, Ambika	3
Sonowal, S	9
Soren, Shibu	3
Soz, Saifuddin	3
Sri Prakasa	3
Stephen, C.M	4
Subbarayan, P	4
Subramaniam, C	4
Sukh Ram	15
Sukhbans Kaur	4
Sukhdev Prasad	5
Suman, Ramjilal	4
Supriyo, Babul	11
Suraj Bhan	5
Suresh, K	4
Sushma Swaraj	37
Swami, Chinmayanand	4
Swami, I.D	4
Swaminathan, R.V	4
Swamy, Subramaniam	5
Swaran Singh	5
Syed, Mahmud	6
Т	
Tariq Anwar	4
Taslim Uddin, Md	
Tewari, Prof. K.K	
Tewari, Manish	
Thakur, Chaoba Singh	
Thakur, C.P	
Thakur, Rameshwar	
Thambi Durai M	

Name P	age
Thangkabalu, K.V.	216
Thara Devi, D.K. Siddartha	216
Tharoor, Shashi	
Thirunavukkarsar, S.U.	216
Thomas, Alungal Mathai	216
Thomas, K.V.	146
Thomas, P.C.	217
Thungan, P.K.	217
Tirath, Krishna	146
Tiwari, Narayan Dutt	117
Tomar, N.S.	238
Tripathi, Kamlapati	117
Tripathy, B.K.	146
Trivedi, Dinesh	
Trivedi, H.M.	218
Tyagi, Mahavir	118
Tytler, Jagdish	
U	
Uma Bharati, Sushri	237
Unnikrishnan, K.P.	
Upendra, P.	
Uppalapati V. Krishnam Raju	
	210
V	
Vaghela, Shankersinh	119
Vajpayee, Atal Bihari	. 17
Varma, Beni Prasad	119
Varma, Chandradeo Prasad	218
Varma, Ravindra	119
Vasan, G.K.	120
Vasava, M.D.	241
Vayalar, Ravi	120
Veerendra Kumar, M.P.	147
Velu, R	218
Venkatapathy, K	218
Venkata Reddy, P	
Venkataraman, R	120
Venkatasubbaiah, P	219
Venkataswamy, G	121

Name	Page
Venkateswarlu, U	147
Venkatraman, T.G	121
Venugopal, K.C	219
Venugopalachari, S	219
Verma, Balgovind	235
Verma, Brij Lal	121
Verma, Rita	219
Verma, S.S	221
Verma, Upendra Nath	220
Verma, Vikram	
Verma, Vimla	219
Vora, Motilal	
Vyas, Girija	
W	
Wasnik, Mukul	122
Yadav, Arun	
Yadav, Balram Singh	
Yadav, Chandrajit	
Yadav, D.P.	
Yadav, Devendra Prasad	
Yadav, Hukumdeo Narayan	
Yadav, J.P.	
Yadav, Laloo Prasad	
Yadav, Mulayam Singh	
Yadav, Narsingh	
Yadav, Jay Prakash Narayan	
Yadav, Ram Kripal	
Yadav, Ram Lakhan Singh	
Yadav, Ramanand	
Yadav, Satpal Singh	
Yadav, Sharad	
Yadav, Shyamlal	
Z	
Zail Singh, Giani	
7ulfiquarullah	

SUBJECT INDEX

Name	Page
A	
Administrative Reforms	4, 38, 83, 106, 141, 144, 182
Agreement (Indo-Pak)	35
Agriculture	3, 17, 18, 19, 22, 23, 24, 25, 30, 32, 34, 35, 36, 43, 48, 50, 52, 53, 59, 61, 69, 70, 71, 75, 76, 77, 82, 84, 85, 90, 93, 109, 110, 114, 115, 116, 124, 146, 147, 153, 155, 157, 173, 175, 177, 178, 179, 181, 186, 188, 189, 190, 194, 195, 199, 202, 207, 208, 209, 210, 212, 214, 215, 216, 217, 219, 220, 221, 222, 223, 226, 227, 228, 229, 231, 233, 238, 240, 241
Agricultural Research/Agriculture Research and Education	207
Agro Industries	28, 66, 75, 126, 133, 135, 141, 158
Animal Husbandry	71, 142
	,
Atomic Energy	1, 3, 5, 6, 7, 8, 10, 11, 12, 15, 16, 17, 18, 19, 48, 80, 83, 141, 155, 182, 184, 187, 208, 240
Aviation	96
Ayurveda, Yoga and Naturopathy, Unani, Siddha and Homoeopathy (AYUSH)	239
В	
Banking	74, 81, 194
Broadcasting	1, 5, 7, 9, 19, 21, 23, 34, 41, 50, 51, 53, 58, 59, 65, 89, 94, 96, 97, 98, 99, 102, 105, 108, 111, 112, 113, 115, 119, 122, 128, 132, 140, 144, 146, 152, 161, 168, 169, 172, 174, 179, 186, 191, 194, 206, 208, 213, 215, 227, 234, 235, 237, 241
С	
Chemicals	3, 13, 17, 25, 28, 30, 31, 32, 43, 47, 48, 56, 60, 67, 68, 72, 78, 79, 81, 83, 84, 85, 86, 87, 88, 99, 100, 102, 106, 123, 128, 135, 139, 151, 153, 156, 157, 158, 162, 163, 168, 173, 183, 184, 186, 188, 202, 204, 210, 211, 221, 224, 226, 227, 228, 229, 233, 238, 240

Name	Page
Civil Aviation	4, 25, 29, 50, 51, 57, 58, 59, 82, 83, 84, 85, 86, 88, 89, 100, 105, 111, 112, 120, 122, 123, 126, 128, 131, 133, 134, 136, 138, 142, 146, 155, 159, 162, 163, 166, 170, 175, 178, 186, 188, 200, 212, 214, 219, 227, 230, 238, 240
Civil Expenditure	94, 126
Civil Supplies	8, 13, 14, 17, 26, 27, 29, 34, 36, 42, 70, 71, 74, 78, 92, 105, 106, 108, 110, 111, 122, 128, 144, 145, 150, 152, 153, 164, 165, 166, 167, 181, 191, 192, 205, 207, 223, 230, 233
Climate Change	239
Coal	8, 9, 17, 20, 21, 31, 35, 37, 41, 50, 51, 52, 53, 75, 76, 81, 84, 98, 99, 113, 117, 119, 138, 140, 142, 143, 147, 152, 156, 169, 183, 191, 192, 193, 200, 231, 239
Commerce	2, 3, 13, 14, 15, 16, 17, 26, 27, 32, 36, 37, 38, 39, 40, 42, 44, 50, 53, 57, 60, 62, 63, 68, 69, 74, 77, 83, 86, 91, 95, 97, 105, 106, 107, 113, 115, 117, 122, 131, 136, 141, 142, 149, 150, 152, 153, 161, 166, 168, 170, 171, 176, 177, 184, 188, 190, 196, 199, 200, 209, 213, 217, 231, 232, 234, 240
Commonwealth Relations	14, 172
Communications	3, 4, 7, 15,17,19, 22, 27, 30, 36, 45, 51, 56, 61, 66, 68, 71, 72, 73, 81, 82, 88, 89, 90, 93, 95, 99, 100, 104, 105, 107, 109, 110, 112, 114, 115, 119, 121, 131, 132, 135, 139, 142, 144, 145, 150, 159, 174, 177, 178, 189, 190, 191, 195, 196, 199, 201, 205, 208, 209, 216, 221, 222, 224, 227, 228, 230, 231, 235, 238
Community Development	48, 69, 114, 124, 125, 184, 185, 189, 208, 209, 226, 227, 229
Company Affairs	3, 13, 24, 35, 39, 47, 48, 53, 54, 55, 57, 58, 59, 60, 63, 64, 84, 95, 104, 106, 117, 136, 149, 153, 158, 178, 184, 188, 193, 197, 199, 202, 203, 221, 224
Consumer Industries/Consumer Affairs	2, 8, 13, 14, 17, 27, 32, 35, 36, 71, 85, 104, 115, 122, 123, 146, 150, 157, 170, 178, 195, 205, 207, 210, 215, 218, 221, 237
Co-operation	42, 48, 69, 70, 77, 110, 114, 124, 125, 153, 165, 166, 179, 184, 185, 189, 207, 208, 209, 222, 226, 227, 229
Corporate Affairs	41, 48, 61, 73, 140, 212, 237
Co-ordination	33

Name	Page
Cottage Industries	109
Cultural Affairs	56, 167, 170, 225
Culture	4, 10, 25, 26, 40, 52, 57, 58, 64, 65, 93, 94, 113, 133, 135, 144, 153, 156, 157, 158, 166, 170, 174, 188, 203, 204, 215, 217, 240
D	
Dairying	71, 142
Defence	1, 4, 6, 11, 12, 13, 15, 17, 20, 21, 23, 27, 28, 31, 32, 33, 38, 45, 48, 50, 54, 58, 65, 69, 70, 73, 75, 79, 80, 83, 84, 85, 86, 87, 101, 108, 115, 116, 120, 123, 131, 143, 144, 145, 152, 157, 166, 168, 171, 175, 178, 180, 189, 191, 192, 198, 199, 201, 202, 208, 210, 211, 212, 214, 217, 218, 226, 228, 229, 232, 234, 237, 239
Defence Co-ordination	49, 63, 201
Defence Expenditure	125
Defence Organisation	118
Defence Production	69, 70, 73, 83, 87, 101, 108, 131, 145, 152, 189, 191, 192, 217, 234
Defence Research & Development/	152
Organisation	152
Defence Supplies Deputy Prime Minister	50, 145, 152, 189, 191, 192 1, 2, 21, 22, 23
Development of North Eastern Region	25, 49, 65, 107, 116, 134, 210, 239
Disinvestment	53, 107
Drinking Water and Sanitation	91, 136, 145, 238, 240
	71, 130, 113, 230, 210
E	
Earth Sciences	32, 42, 63, 95, 109, 120, 130, 238, 241
Ecology	160
Economic Affairs	49, 63, 93, 162, 235
Economic Co-ordination	201
Education	24, 28, 29, 37, 38, 40, 56, 57, 58, 64, 80, 92, 93, 100, 103, 107, 128, 135, 153, 154, 156, 157, 166, 167, 179, 182, 188, 203, 204, 205, 207, 217, 225, 226, 227, 228, 232, 236
Electronics	1, 3, 4, 6, 7, 12, 17, 18, 54, 64, 80, 83, 162, 182, 187, 208, 211, 233
Employment	3, 8, 9, 10, 11, 13, 14, 18, 22, 23, 28, 46, 49, 58, 60, 71, 72, 76, 78, 87, 92, 98, 109, 120, 121, 128, 131, 137, 147, 148, 150, 157, 172, 173, 192, 194, 198, 201, 205, 214, 217, 218, 227, 229, 233, 239

Name	Page
Energy	37, 44, 46, 51, 57, 59, 72, 77, 80, 91, 99, 106, 140, 144, 156, 159, 177, 183, 204, 211, 235
Environment	3, 4, 9, 160, 211, 239
Environment & Forests	17, 20, 26, 27, 30, 34, 57, 73, 86, 90, 91, 96, 113, 129, 133, 138, 139, 154, 169, 180, 181, 189, 203
Estates	43
Expenditure	85, 118, 163, 214
External Affairs	1, 3, 4, 5, 6, 7, 9, 11, 12, 14, 15, 16, 17, 18, 22, 33, 34, 37, 38, 43, 44, 54, 61, 62, 70, 73, 74, 75, 76, 77, 104, 106, 108, 109, 112, 113, 116, 117, 126, 136, 141, 149, 153, 154, 160, 161, 162, 171, 172, 174, 176, 182, 187, 191, 197, 201, 208, 211, 212, 213, 215, 216, 218, 225, 237, 238
F	
Family Planning	37, 57, 102, 112, 155, 165, 172, 185, 188, 198, 223, 226, 228
Family Welfare	1, 2, 3, 8, 12, 13, 19, 27, 30, 46, 61, 62, 71, 89, 91, 93, 96, 98, 103, 104, 112, 115, 116, 118, 122, 127, 130, 131, 134, 137, 139, 143, 153, 158, 163, 171, 174, 176, 179, 208, 210, 216, 219, 221, 225, 227, 238
Farmers Welfare	238, 240, 241
Fertilizers	3, 13, 17, 25, 28, 30, 31, 32, 43, 48, 60, 67, 72, 76, 78, 79, 81, 84, 85, 86, 99, 102, 106, 123, 128, 135, 139, 156, 162, 183, 184, 194, 202, 204, 210, 211, 221, 229, 238, 240
Finance	1, 2, 4, 5, 8, 9, 10, 15, 16, 17, 22, 31, 33, 38, 39, 40, 42, 47, 54, 62, 63, 68, 70, 72, 74, 75, 79, 81, 82, 85, 101, 108, 112, 115, 117, 118, 120, 125, 126, 131, 143, 147, 149, 150, 158, 160, 162, 163, 169, 172, 181, 185, 186, 189, 190, 191, 193, 194, 196, 199, 203, 206, 207, 213, 216, 222, 223, 233, 234, 235, 237, 241
Food	8, 13, 17, 23, 24, 25, 29, 32, 34, 35, 48, 52, 61, 69, 70, 71, 75, 82, 85, 90, 92, 104, 106, 109, 110, 111, 114, 116, 122, 124, 128, 134, 140, 144, 145, 146, 152, 157, 167, 173, 175, 178, 191, 192, 195, 208, 209, 210, 215, 216, 217, 218, 221, 226, 233, 237
Food Processing Industries	3, 17, 18, 20, 50, 66, 85, 90, 93, 94, 96, 97, 123, 132, 134, 142, 143, 144, 147, 166, 178, 209, 215, 220, 238, 241

Name	Page
Forests	3, 4, 9, 239
Foreign Trade	33, 73, 164, 198
Fuel	67, 115, 116, 151, 166
н	
Health	1, 2, 3, 8, 12, 13, 19, 25, 27, 30, 37, 46, 57, 61, 62, 69, 71, 89, 91, 93, 96, 98, 102, 103, 104, 112, 115, 116, 118, 122, 127, 129, 130, 131, 134, 137, 139, 143, 153, 155, 158, 163, 165, 171, 172, 174, 176, 179, 185, 188, 198, 208, 210, 216, 219, 221, 223, 225, 226, 227, 228, 231, 232, 238
Heavy Engineering	33, 64, 80, 85, 86, 101, 114, 116, 126
Heavy Industries	2, 41, 42, 56, 73, 78, 79, 82, 86, 101, 103, 114, 156, 164, 169, 170, 171, 193, 196, 205, 215, 217, 220, 234, 238, 240
Home Affairs	1, 2, 3, 5, 11, 12, 14, 16, 21, 22, 23, 29, 36, 38, 39, 40, 43, 47, 49, 52, 55, 58, 59, 66, 70, 71, 72, 76, 77, 79, 80, 82, 83, 90, 95, 96, 97, 99, 102, 105, 107, 108, 120, 123, 129, 135, 139, 150, 158, 159, 164, 167, 169, 176, 178, 180, 182, 190, 192, 193, 197, 199, 200, 203, 207, 209, 211, 212, 213, 214, 215, 219, 223, 230, 231, 232, 237, 240, 241
Homoeopathy	134
Housing	7, 34, 36, 37, 41, 42, 46, 88, 90, 94, 95, 102, 105, 109, 110, 115, 125, 126, 155, 156, 165, 180, 182, 185, 201, 202, 223, 224, 226, 230
Housing & Urban Poverty Alleviation	10, 65, 67, 122, 237
Human Resource Development	12, 13, 17, 27, 31, 35, 50, 55, 64, 79, 100, 106, 109, 118, 122, 150, 151, 153, 154, 159, 160, 163, 165, 169, 171, 177, 181, 182, 192, 195, 196, 203, 204, 205, 216, 219, 220, 238, 241
1	
Indian System of Medicines	134
Industrial Development	24, 26, 28, 40, 44, 74, 103, 114, 189, 199, 202, 203, 205, 224, 228, 234
Industrial Policy and Promotion	63
Industrial Research	80
Industry	1, 2, 3, 6, 14, 15, 16, 25, 26, 28, 40, 44, 45, 53, 57, 59, 62, 63, 68, 69, 74, 78, 79, 84, 86, 87, 91, 92, 95, 97, 98, 103, 105, 107, 109, 111, 116, 117, 120, 121, 126, 135, 140, 142, 149, 152, 155, 161, 164, 168, 170, 171, 176, 177, 180, 181, 183, 184, 188, 190, 194, 196, 197, 199, 200, 205, 210, 213, 215, 217, 221, 225, 234, 235, 240

Name	Page
Information	1, 5, 7, 9, 19, 21, 23, 34, 41, 50, 51, 53, 58, 59, 65, 89, 94, 96, 97, 98, 99, 102, 105, 108, 111, 112, 113, 115, 119, 122, 128, 132, 140, 144, 146, 152, 161, 168, 169, 172, 174, 179, 186, 191, 194, 206, 208, 213, 215, 227, 234, 235, 237, 241
Information Technology	19, 66, 68, 90, 93, 107, 109, 135, 139, 142, 150, 159, 174, 177, 191, 195, 209, 216, 238
Insurance	194
Internal Security	77, 139
Internal Trade	24, 44, 203, 224, 228
International Trade	103
Iron	101, 126
Irrigation	. 10, 23, 24, 26, 30, 32, 36, 37, 40, 49, 51, 70, 79, 80, 82, 102, 104, 111, 151, 173, 183, 200, 209, 210, 228, 229, 231, 234, 235
J	
Jammu & Kashmir/Kashmir Affairs	3, 7, 13, 18, 19, 45
Justice	13, 34, 35, 39, 44, 47, 53, 54, 55, 57, 58, 60, 61, 63, 64, 73, 95, 101, 104, 106, 109, 115, 117, 136, 140, 153, 158, 178, 184, 188, 193, 197, 217, 218, 221, 237
L	
Labour	1, 8, 9, 10, 11, 12, 17, 19, 22, 23, 28, 42, 44, 46, 47, 49, 50, 54, 60, 61, 72, 78, 80, 83, 84, 87, 91, 92, 97, 98, 117, 119, 120, 121, 123, 127, 128, 132, 134, 146, 147, 157, 160, 164, 165, 172, 173, 179, 184, 190, 194, 195, 201, 203, 205, 213, 214, 222, 227, 229, 233, 235, 239
Law	13, 25, 34, 35, 39, 44, 47, 49, 53, 54, 55, 57, 58, 60, 61, 63, 64, 69, 73, 82, 87, 95, 100, 101, 104, 106, 109, 115, 117, 126, 136, 140, 153, 157, 158, 166, 167, 178, 182, 184, 188, 193, 194, 197, 201, 202, 217, 218, 221, 233, 237
Legal Affairs	126
M	_
Materials Planning	87
Metals	94, 100, 101, 125, 126, 157, 188, 198, 202, 224, 229
Micro, Small & Medium Enterprises	42, 81, 111, 120, 137, 238, 240

Name	Page
Minerals	19, 64, 84, 219
Mines	3, 6, 8, 17, 19, 20, 21, 31, 41, 43, 46, 47, 48, 49, 64, 66, 67, 74, 75, 76, 78, 80, 81, 84, 85, 94, 95, 97, 99, 100, 101, 113, 114, 115, 116, 117, 119, 125, 126, 128, 132, 140, 148, 151, 155, 157, 163, 166, 173, 179, 183, 188, 193, 198, 200, 202, 203, 213, 219, 220, 221, 224, 226, 229, 235, 238, 241
Minority Affairs	27, 35, 60, 61, 125, 162, 190
N	
Natural Gas	3, 8, 18, 24, 30, 41, 72, 73, 76, 81, 91, 94, 104, 117, 127, 131, 143, 145, 161, 164, 175, 176, 177, 195, 196, 204, 212, 239
Natural Resources	29, 67, 113
New & Renewable Energy	24, 168, 239
Non-Conventional Energy Sources	3, 8, 12, 18, 20, 30, 64, 75, 136, 138, 140, 145, 175, 176, 211, 219
Non-Ferrous Metals	158, 188
Non-Resident Indian Affairs	147
0	
Ocean Development	1, 3, 4, 7, 12, 17, 18, 19, 55, 64, 83, 108, 162, 182, 187, 211
Oil	67
Overseas Indian Affairs	9, 45, 120, 147, 237, 238
P	
Panchayati Raj	24, 41, 42, 55, 124, 184, 229, 238, 241
Parliamentary Affairs	7, 29, 32, 34, 36, 40, 48, 57, 59, 60, 63, 64, 65, 66, 67, 76, 81, 88, 89, 93, 97, 103, 107, 108, 110, 111, 112, 115, 119, 120, 122, 130, 134, 135, 138, 140, 142, 147, 148, 150, 151, 152, 157, 158, 161, 162, 164, 165, 166, 167, 175, 179, 180, 181, 182, 186, 187, 189, 193, 196, 197, 198, 201, 205, 209, 213, 216, 219, 225, 226, 228, 231, 232, 237, 239, 240
Parliamentary Secretary	45, 152
Pensions	1, 3, 7, 8, 10, 12, 17, 18, 19, 21, 38, 39, 83, 106, 129, 130, 141, 144, 151, 152, 162, 169, 187, 189, 192, 240
Personnel	1, 3, 4, 7, 8, 10, 12, 17, 18, 19, 21, 38, 39, 70, 83, 106, 129, 130, 141, 144, 151, 152, 161, 162, 169, 182, 187, 189, 192, 232, 240

Name	Page
Petro Chemicals	168, 186
Petroleum	3, 8, 18, 24, 26, 30, 31, 32, 41, 47, 48, 56, 67, 68, 72, 73, 76, 79, 81, 83, 87, 88, 91, 94, 100, 102, 104, 106, 117, 127, 131, 143, 145, 151, 153, 156, 157, 158, 161, 163, 164, 173, 175, 176, 177, 183, 188, 195, 196, 202, 204, 212, 221, 224, 226, 227, 228, 233, 239
Planning	1, 3, 5, 6, 7, 8, 10, 11, 18, 19, 33, 35, 38, 42, 44, 63, 68, 72, 73, 76, 87, 102, 106, 107, 108, 112, 114, 117, 127, 132, 141, 145, 148, 157, 162, 165, 177, 183, 184, 187, 191, 194, 198, 207, 209, 239
Planning Commission	112, 117
Posts	100, 224
Poverty Alleviation	26, 43, 51, 54, 60, 65, 130, 131, 137, 150, 197, 198, 241
Power	3, 8, 9, 10, 18, 19, 20, 24, 40, 46, 47, 49, 51, 59, 64, 73, 77, 80, 82, 86, 91, 97, 99, 104, 105, 127, 140, 143, 144, 145, 151, 182, 192, 200, 204, 219, 224, 228, 229, 234, 235, 239
Prime Minister	1, 2, 3, 5, 11, 14, 15, 16, 17, 18, 19, 45, 152, 178, 211, 231
Prime Minister's Office	129, 130, 155, 161, 165, 173, 182, 184, 187, 231, 239
Production	46, 95
Programme Implementation	1, 3, 4, 8, 18, 20, 35, 37, 76, 106, 112, 127, 132, 133, 145, 148, 162, 165, 207
Public Distribution	8, 13, 14, 17, 27, 35, 36, 71, 85, 104, 122, 123, 146, 150, 157, 178, 195, 205, 207, 210, 215, 218, 237
Public Enterprises	41, 42, 56, 73, 82, 86, 164, 169, 170, 171, 193, 196, 215, 217, 220, 238, 240
Public Grievances	1, 3, 7, 8, 10, 12, 17, 18, 19, 21, 38, 39, 83, 106, 129, 130, 141, 144, 151, 152, 162, 169, 187, 189, 192, 240
R	
Railways	10, 11, 13, 15, 18, 19, 20, 22, 28, 31, 32, 36, 37, 40, 45, 49, 51, 56, 60, 66, 69, 72, 73, 76, 77, 78, 79, 81, 82, 85, 86, 96, 99, 100, 102, 110, 115, 116, 117, 118, 122, 137, 145, 149, 151, 158, 159, 161, 165, 169, 172, 173, 177, 179, 180, 184, 193, 197, 202, 206, 207, 212, 218, 223, 225, 227, 232, 234, 237, 240

Name	Page
Rehabilitation	22, 23, 36, 50, 52, 61, 70, 72, 77, 83, 84, 88, 90, 98, 109, 118, 121, 125, 128, 132, 157, 160, 172, 173, 201, 203, 206, 213, 217, 224, 226, 227, 230, 233, 235
Relief	77, 206
Revenue	74, 81, 85, 94, 118, 125, 126, 191, 214
River Development and Ganga Rejuvenation	237, 240
River Valley Scheme	10
Road Transport and Highways	10, 30, 46, 56, 57, 60, 136, 156, 174, 186, 196, 197, 206, 212, 237, 241
Rural Areas/Affairs	18, 71, 76, 137, 192, 198, 218
Rural Development	14, 26, 36, 42, 55, 61, 75, 84, 89, 91, 92, 104, 109, 110, 121, 139, 149, 153, 155, 168, 171, 178, 183, 187, 192, 193, 194, 198, 205, 206, 212, 216, 218, 219, 220, 238, 241
Rural Employment	137
Rural Industries	133, 135, 141, 158, 176
Rural Reconstruction	110, 153, 210, 214, 228
S	
Science	1, 3, 6, 12, 17, 18, 32, 42, 55, 63, 64, 78, 80, 83, 94, 108, 109, 114, 120, 127, 130, 155, 164, 182, 187, 202, 208, 210, 211, 231, 238, 241
Scientific Research	14, 29, 56, 67, 80, 107, 113, 225
Shipping	2, 23, 26, 30, 36, 43, 47, 53, 59, 72, 74, 83, 84, 89, 92, 95, 96, 105, 112, 114, 117, 120, 137, 154, 155, 156, 159, 163, 181, 186, 199, 216, 218, 220, 226, 237, 246
Skill Development and Entrepreneurship	239
Small Scale Industries	28, 66, 116, 135, 141, 176, 210
Social Justice and Empowerment	54, 60, 65, 69, 122, 133, 164, 168, 181, 184, 185, 186, 191, 238, 241
Social Security	100, 109, 129, 167, 201
Social Welfare	38, 40, 49, 58, 68, 69, 75, 87, 88, 93, 129, 135, 153, 156, 166, 179, 188, 202, 217, 232, 233, 236
Space	1, 3, 4, 6, 7, 8, 10, 12, 17, 18, 19, 80, 83, 141, 155, 182, 184, 187, 208, 231, 240
Sports	4, 24, 25, 31, 36, 134, 180, 239
States	23, 28, 29, 34, 58, 97, 100
Statistics & Programme Implementation	20, 45, 47, 52, 106, 120, 133, 135, 165,

Name	Page
Steel	3, 6, 18, 31, 33, 41, 46, 47, 64, 66, 67, 74, 75, 76, 78, 80, 81, 84, 85, 86, 94, 95, 97, 99, 101, 111, 114, 115, 116, 117, 119, 126, 128, 142, 146, 155, 159, 173, 179, 183, 193, 195, 196, 198, 213, 220, 221, 226, 235, 238, 241
Supply	3, 14, 16, 33, 36, 40, 46, 49, 69, 70, 74, 87, 88, 90, 94, 95, 97, 98, 109, 115, 116, 121, 125, 126, 128, 132, 144, 158, 167, 172, 173, 177, 183, 189, 198, 201, 202, 217, 224, 225, 226, 230
Surface Transport	1, 18, 20, 54, 62, 75, 77, 90, 117, 119, 121, 137, 139, 147, 186, 195, 220
Т	
Technical Development	49, 87, 167
Technology	1, 3, 6, 12, 17, 18, 32, 42, 55, 63, 64, 78, 80, 83, 94, 108, 109, 114, 120, 127, 130, 155, 164, 182, 187, 202, 208, 210, 211, 231, 238, 241
Telegraphs	100, 224
Textiles	3, 18, 38, 50, 53, 57, 68, 71, 92, 105, 119, 121, 123, 127, 134, 137, 142, 144, 150, 161, 174, 176, 193, 199, 220, 239
Tourism	4, 22, 25, 26, 29, 34, 50, 52, 57, 58, 60, 62, 65, 77, 83, 84, 86, 88, 89, 96, 97, 99, 100, 105, 111, 113, 118, 130, 132, 133, 134, 135, 136, 146, 151, 152, 155, 158, 159, 162, 170, 174, 175, 178, 179, 188, 200, 212, 213, 214, 227, 240
Trade	171
Training	38, 83, 141, 144, 161
Transport	2, 4, 15, 22, 23, 26, 28, 32, 36, 43, 56, 59, 62, 68, 72, 74, 82, 83, 84, 85, 88, 89, 92, 95, 96, 100, 105, 114, 117, 139, 146, 151, 154, 155, 156, 172, 173, 181, 199, 206, 218, 224, 226, 227, 230
Tribal Affairs	35, 41, 56, 65, 156, 174, 175, 187, 189, 238, 241
U	
Urban Affairs	3, 8, 13, 14, 28, 54, 58, 109, 121, 130, 131, 147, 148, 150, 217
Urban Development	7, 9, 26, 28, 29, 46, 51, 55, 57, 58, 60, 62, 66, 68, 94, 98, 102, 112, 126, 130, 131, 155, 158, 159, 165, 185, 197, 198, 204, 224, 237, 241
Urban Employment	43, 65, 150

Name	Page
W	
Waqfs Muslim	222
Wasteland Development	198
Water Resources	1, 4, 7, 10, 18, 27, 30, 32, 40, 41, 44, 61, 62, 66, 69, 71, 72, 78, 93, 101, 104, 108, 113, 116, 154, 167, 190, 200, 204, 214, 218, 221, 237, 240
Welfare	1, 59, 75, 80, 91, 92, 129, 132, 133, 173, 189, 213, 214, 228
Women	129, 153, 238
Women & Child Development	31, 118, 130, 146, 151, 212, 219, 238
Works	7, 34, 36, 37, 41, 42, 43, 46, 88, 90, 94, 95, 102, 105, 109, 110, 115, 125, 126, 155, 156, 165, 180, 182, 185, 201, 202, 223, 224, 225, 226
Υ	
Youth Affairs	4, 24, 25, 31, 239
Youth Affairs & Sports	9, 43, 44, 45, 47, 50, 67, 118, 122, 151, 159, 160, 166, 168, 193, 197, 211, 215, 220
Youth Services	92, 154, 227, 228.

